H24/3662

#### 

#### Sec. Research Graduate School Committee

Under the Copyright Act 1968, this thesis must be used only under the normal conditions of scholarly fair dealing for the purposes of research, criticism or review. In particular no results or conclusions should be extracted from it, nor should it be copied or closely paraphrased in whole or in part without the written consent of the author. Proper written acknowledgement should be made for any assistance obtained from this thesis.

# Pharmacological characterisation of relaxin and the relaxin receptor

Courtney Peta Judkins B.Sc. (Hons) (Monash University)

This thesis is presented to the Faculty of Medicine, Monash University for the degree of

## **Doctor of Philosophy**

Department of Pharmacology Monash University Victoria Australia

February 2004

Table of Contents	i
Abstract	ix
Declaration	xi
Acknowledgments	xii
Publications	xiii
Abbreviations	xvii

## Chapter One: General Introduction

1

i

1.1	History of relaxin				
1.2	Characterisation of relaxin				
	1.2.1	Structure	and molecular characterisation of relaxin	3	
	1.2.2	Relaxir: a	and the insulin gene superfamily	5	
	1.2.3	Sources of	of relaxin	9	
		1.2.3.1	Extraction of native relaxin	9	
		1.2.3.2	Synthetics production of relaxin peptides	9	
		1.2.3.3	Production of relaxin by recombinant methods	10	
	1.2.4	Structure	-activity relationships of relaxin	11	
		1.2.4.1	A chain	11	
		1.2.4.2	B chain	12	
1.3	Bioassays for relaxin				
	1.3.1	<i>In vivo</i> fi	inctional assays	13	
		1.3.1.1	Guinea pig pubis symphysis palpation assay	13	
		1.3.1.2	Mouse interpubic ligament elongation assay	14	
	1.3.2	In vitro f	unctional assays	15	
		1.3.2.1	Uterine contractility/relaxation assay	15	
		1.3.2.2	Rat isolated atrial bioassay	15	
		1.3.2.3	Cellular functional assays	16	
1.4	The physiological actions of relaxin				
	1.4.1	Female r	eproductive tissues	17	
	1.4.2	Male rep	roductive tissues	19	
	1.4.3	Cardiova	iscular system	20	
	1.4.4	Brain		21	

	1.4.5	Connective tissue and cell growth			
	1.4.6	Relaxin gene knockout mouse	24		
1.5	Relaxin receptor binding assays				
	1.5.1	Specific relaxin radioligands	27		
		1.5.1.1 Radioiodinated relaxin	27		
		1.5.1.2 Phosphorylated relaxin	28		
	1.5.2	In vivo localisation of relaxin receptors	28		
	1.5.3	In vitro localisation of relaxin receptors			
	1.5.4	Pharmacokinetic binding assays			
1.6	Identification of the relaxin receptor				
	1.6.1	Leucine-rich repeat-containing G-protein coupled receptor			
		family	31		
	1.6.2	LGR7, the putative relaxin receptor			
1.7	Aims o	f the present study			
-					

## Chapter Two: General Methods

36

100.00

2.1	Cell culture					
	2.1.1	Maintena	nce of cell lines	37		
		2.1.1.1	Non-adherent cell lines	37		
		2.1.1.2	Adherent cell lines	38		
	2.1.2	Freezing	Freezing and thawing of cells			
		2.1.2.1	Freezing of cells	39		
		2.1.2.2	Thawing of cells	39		
	2.1.3	Cell counting				
	2.1.4	Transient transfection of adherent cells				
	2.1.5	Production of a cell line stably expressing a foreign protein				
2.2	Production of recombinant protein					
•	2.2.1	Incorpora	ation of Pml I restriction site into the pSinHis			
		plasmid		41		
		2.2.1.1	Transformation of pSinHis plasmid into DH5a cells	45		
		2.2.1.2	Colony selection	45		

ii

		2.2.1.3	Amplification of altered pSinHis plasmid DNA	46
	2.2.2	Productio	n of the relaxin DNA fragment	47
		2.2.2.1	Purification of relaxin PCR DNA fragment	47
		2.2.2.2	Ligation of relaxin DNA fragments into PCR-script	48
	2.2.3	Ligation o	of relaxin DNA fragment into the pSinHis plasmid	49
		2.2.3.1	Sequencing of pSinHis plasmid containing the	
			relaxin construct	50
	2.2.4	RNA tran	scription of DNA templates	51
	2.2.5	Transfect	ion of pSinHis RNA transcripts into BHK cells	52
	2.2.6	Protein extraction		
		2.2.6.1	Prepare column for purification	54
		2.2.6.2	Protein purification	54
	2.2.7	Analysis	of recombinant protein	55
2.3	Radiolig	gand bindii	ng studies	56
	2.3.1	Preparation of [ <sup>33</sup> P]-H2RLX		
		2.3.1.1	Labelling of recombinant H2RLX with $[\gamma^{-33}P]$ -ATP	56
		2.3.1.2	Separation of unreacted $[\gamma^{-33}P]$ -ATP from labelled	
			relaxin	56
		2.3.1.3	Separation of labelled and unlabelled relaxin	58
	2.3.2	Quentitat	ive autoradiography	59
		2.3.2.1	Tissue preparation	59
		2.3.2.2	[ <sup>33</sup> P]-H2RLX quantitative autoradiography binding	
			protocol	60
		2.3.2.3	Phosphorimaging	62
•		2.3.2.4	Preparation of $[\gamma^{-33}P]$ -ATP reference standards	62
		2.3.2.5	High resolution autoradiography using photographic	;
			emulsion	62
	2.3.3	High-thro	oughput cellular binding assay	63
		2.3.3.1	Cell preparation	63
		2.3.3.2	[ <sup>33</sup> P]-H2RLX cell binding assay	64
2.4	Cellular	functiona	l assays	65
	2.4.1	cAMP as	say	65
	2.4.2	AlphaScr	een <sup>™</sup> cAMP assay	65

たび三部とした

· : ....

.

. . . 4

2.5	Determination of RNA and DNA concentration	66
2.6	Protein determination	67
2.7	Protein analysis	67
2.8	Data analysis	69
2.9	Drugs and reagents	70

Chap	oter Thre	ee: Struct	ure-activity relationships of relaxin analogues		
		at the	human relaxin receptor	72	
3.1	Introdu	uction		73	
3.2	Methods				
	3.2.1	Develop	ment of the high through-put 96 well binding assay	76	
		3.2.1.1	Harvesting filter pre-treatment	76	
		3.2.1.2	Cell number and setup	79	
	3.2.2	Cell cult	ture	81	
		3.2.2.1	THP-1 cell preparation	81	
	3.2.3	High thr	ough-put 96 well binding assay	81	
		3.2.3.1	[ <sup>33</sup> P]-H2RLX (B33) preparation	82	
		3.2.3.2	Saturation binding protocol	82	
		3.2.3.3	Competition binding protocol	82	
	3.2.4	Analysis	6	83	
	3.2.5	Drugs at	nd reagents	83	
3.3	Result	s		84	
	3.3.1	Characte	eristics of [ <sup>33</sup> P]-H2RLX binding to the human relaxi	n	
		receptor		84	
	3.3.2	Competi	ition for binding at the human relaxin receptor	84	
		3.3.2.1	Native human relaxin analogues	84	
		3.3.2.2	Species homologues of relaxin	84	
		3.3.2.3	Effect of shortening the B chain on affinity of		
			HIRLX	85	
		3.3.2.4	Relaxin-like affinity of INSL3 analogues	85	
3.4	Discus	sion		92	

.

## iv

.

Ģ

-----

6

And in succession in the second s

•

Chap	ter Four	: Produc	tion of reco	mbinant relaxin	96
4.1	Introdu	ction			97
4.2	Method	ls and Res	ults		100
	4.2.1	Ligation	of the RLX	insert into pSinHis plasmid	102
		4.2.1.1	Incorporat	tion of Pml I restriction site into pSinHis	
			plasmid		102
		4.2.1.2	Preparatio	n of the RLX DNA insert	102
		4.2.1.3	Ligation o	f RLX insert into pSinHis plasmid	103
		4.2.1.4	Activity o	f Pml I and Sph I enzymes	109
		4.2.1.5	Hybridisa	tion screening for the RLX insert	109
			4.2.1.5.1	Preparation of the RLX DNA probe	109
			4.2.1.5.2	Hybridisation screen	110
		4.2.1.6	Regenerat	ion of the RLX insert	114
		4.2.1.7	Blunt-end	ligation reaction	114
		4.2.1.8	Generatio	n of RLX insert from PCR script	114
			4.2.1.8.1	Sequencing of PCR script plasmid	
				containing the RLX insert	115
		4.2.1.9	Freeze 'n	Squeeze purification and ligation of RLX	
			insert into	the pSinHis plasmid	117
	4.2.2	RNA tra	nscription o	f DNA templates	118
	4.2.3	RLX pro	otein produc	tion and extraction	121
	4.2.4	Analysis	s of recombi	nant protein	121
4.3	Discus	sion			125
Char	oter Five:	: .Charact	terisation of	f the putative relaxin receptor, LGR7	128

5.1	Introdu	uction		129
5.2	Methods			
	5.2.1	Cell cult	ure	131
		5.2.1.1	Transient transfection of CHO K1 cells	131
	5.2.2	96 well t	oinding assay	132
		5.2.2.1	[ <sup>33</sup> P]-H2RLX (B33) preparation	132

v

.

		5.2.2.2	Saturation binding protocol	132
		5.2.2.3	Competition binding protocol	133
	5.2.3	cAMP ac	cumulation assay	133
	5.2.4	Analysis		134
	5.2.5	Drugs an	d reagents	134
5.3	Results			135
	5.3.1	Characte	ristics of [ <sup>33</sup> P]-H2RLX binding to the LGR7 receptor	135
	5.3.2	Competit	tion for [ <sup>33</sup> P]-H2RLX binding at the LGR7 receptor	135
	5.3.3	Stimulati	on of cAMP accumulation by the LGR7 receptor	139
5.4	Discuss	ion		141

<b>Chapter Six:</b>	Characterisation of LGR7 and LGR8 receptors	144
---------------------	---	-----

6.1	Introdu	ction		145		
6.2	Method	S		147		
	6.2.1	Producti	on of a cell line stably expressing the LGR7 and the			
		LGR8 re	ceptors	147		
	6.2.2	Cell cult	Cell culture			
	6.2.3	High thr	ough-put 96 well binding assay	148		
		6.2.3.1	[ <sup>33</sup> P]-H2RLX (B33) preparation	148		
		6.2.3.2	Saturation binding protocol	148		
		6.2.3.3	Competition binding protocol	149		
	6.2.4	cAMP accumulation assay				
	6.2.5	Analysis				
	6.2.6	Drugs and reagents				
6.3	Results			151		
	6.3.1	Characte	eristics of [ <sup>33</sup> P]-H2RLX binding	151		
		6.3.1.1	[ <sup>33</sup> P]-H2RLX binding characteristics of the LGR7			
			receptor	151		
		6.3.1.2	[ <sup>33</sup> P]-H2RLX binding characteristics of the LGR8			
			receptor	151		
	6.3.2	Competi	tion binding characteristics	154		
		6.3.2.1	Competition for binding at the LGR7 receptor	154		
				vi		

.

.

		6.3.2.2	Competition for binding at the LGR8 receptor	154
	6.3.3	cAMP a	ccumulation assay	159
		6.3.3.1	Stimulation of cAMP production by activation of th	e
			LGR7 receptor stably expressed in HEK293 cells	159
		6.3.3.2	Stimulation of cAMP production by activation of th	e
			LGR8 receptor stably expressed in HEK293 cells	162
6.4	Discuss	ion		164
		1.005		1.60
Chap	oter Sevei	n: LGR7	expression in the human uterus	168
7.1	introdu	ction		169
7.2	Method	ls		171
	7.2.1	Human t	issue collection and preparation	171
		7.2.1.1	Preparation of slide mounted uterus sections	171
		7.2.1.2	RNA extraction from human uterus tissue samples	171
	7.2.2	Expressi	on and localisation of the relaxin receptor using	
		[ <sup>33</sup> P]-H2	RLX	172
		7.2.2.1	[ <sup>33</sup> P]-H2RLX (B33) preparation	172
		7.2.2.2	Relaxin receptor expression throughout the phases	
			of the menstrual cycle	172
		7.2.2.3	Localisation of the relaxin receptor using high	
			resolution autoradiography	172
	7.2.3	Characte	erisation of [ <sup>33</sup> P]-H2RLX binding in the proliferative	
		and secr	etory phases	173
	7.2.4	Analysis	of relaxin, LGR7 and related gene expression	174
	7.2.5	Statistic	al analysis	177
7.3	Results			178
	7.3.1	Changes	in binding of [ <sup>33</sup> P]-H2RLX with the phases of the	
		menstru	al cycle	178
	7.3.2	Localisa	tion of the relaxin receptor in uterine sections using	
		emulsion	n autoradiography	178

emulsion autoradiography1787.3.3Characterisation of [<sup>33</sup>P]-H2RLX binding in the proliferative<br/>and secretory phases179

vii

7.3.4	Analysis of relaxin, LGR7 and related gene expression	185							
Discussion									
Acknow	wledgements	194							
er Eigh	t: General Discussion	195							
Genera	l Discussion	196							
	7.3.4 Discuss Acknow er Eigh Genera	<ul> <li>7.3.4 Analysis of relaxin, LGR7 and related gene expression</li> <li>Discussion</li> <li>Acknowledgements</li> <li>er Eight: General Discussion</li> <li>General Discussion</li> </ul>							

## Chapter Nine: References

#### Abstract

The work presented in this thesis entitled "Pharmacological characterisation of relaxin and the relaxin receptor" investigated the pharmacological interaction of various relaxin analogues with the human relaxin receptor expressed (i) endogenously in THP-1 cells, (ii) recombinantly in CHO K1 and HEK293 cells, and (iii) physiologically in uterine tissue. Investigation of the structural interaction between HRLX and the relaxin receptor involved the development of a method for the recombinant production of relaxin peptides, which in conjunction with the use of bioassays utilising human receptors, would allow determination of features important for relaxin activity.

Chapter One provides a general overview of relaxin, including the history of relaxin discovery and isolation, the structure-activity relationships and physiological characterisation of relaxin in various species, and the identification of the relaxin receptor. Chapter Two details the methods used in this thesis to investigate the interaction between relaxin and the human relaxin receptor.

The interaction between the human relaxin receptor endogenously expressed in THPi cells and various relaxin and INSL3 analogues is described in Chapter Three. Relaxin-like affinity was determined using a high through-put binding assay and the selective relaxin radioligand, [<sup>33</sup>P]-H2RLX. Affinity for the relaxin receptor was influenced by changes in the structural-integrity of the relaxin peptide and the presence of the relaxin binding motif along the alpha helix of the B chain. The results demonstrated distinct differences in the interaction between the relaxin analogues and the human versus the rat relaxin receptor. To investigate novel areas within human relaxin that influence the affinity for the relaxin receptor, a mammalian recombinant peptide expression system was developed, described in Chapter Four. Although it appeared that the recombinant relaxin peptide was successfully synthesised, isolation of the protein failed, halting further peptide production.

Chapter Five confirmed the identity of the putative relaxin receptor, LGR7. Binding and cAMP assays analysing the interaction between various relaxin analogues and

ìx

the human LGR7 receptor transiently expressed in CHO K1 cells, correlated with previous studies at the relaxin receptor expressed in THP-1 cells. Chapter Six investigated the interaction between relaxin analogues and the LGR7 and LGR8 receptors stably expressed in HEK293 cells. While all relaxin analogues could interact with LGR7, they selectively interacted with LGR8. The pharmacological interaction between relaxin and the two receptors also differed, with evidence suggesting binding to a single population of sites at LGR7 and binding to two sites on LGR8.

Chapter Seven provides evidence for a specific role for relaxin in human reproduction. Relaxin receptor binding studies and examination of gene expression in human uterine tissue showed a distinct cyclic expression of the receptor, peaking during the early-mid secretory phase. This information coupled with known circulating levels of relaxin peptide suggests a role for relaxin in implantation.

The contribution of the results presented in this thesis to the existing literature is discussed in Chapter Eight. Overall, this thesis has extended the knowledge of the interaction between relaxin and the human relaxin receptor, shown distinct differences in the ability of relaxin analogues to interact at LGR7 and LGR8 and provided further evidence supporting a function of relaxin in human reproduction.

#### Declaration

I declare that the contents of this thesis have not previously been accepted for the award of any other degree or diploma in any other university or institution. Furthermore, this thesis contains no material previously published by another person except where due reference is made.

I certify that the writing of this thesis with the results, interpretation, opinions and suggestions presented are entirely my own work.

This thesis is less than 100,000 words, excluding tables, figure legends and references.



**Courtney Peta Judkins** 

February 2004

#### Acknowledgments

I would like to take this opportunity to thank those who gave their support, encouragement and most importantly, their friendship during the course of this thesis.

Firstly the boss, Roger, for never locking his door. Your time and patience was greatly appreciated. To Geoff, my home away from home supervisor, thank you for always smiling and your endless encouragement. To my past and present lab members, I have had a ball sharing the last four years with you. Thank you for your support, friendship and laughter. Thanks also to all the members of the Monash Pharmacology department, especially Emma for her support and friendship, and Dan for his endless assistance, even when it was inconvenient for him.

To my family and friends for always being there, and to my parents for the neverending support and encouragement you have provided for me, I would not have accomplished as much without you.

And to my gorgeous husband Adam, you are my rock, this is yours as much as it is mine. Here's to us!

#### Publications

During the course of this research, the following publications have been published or submitted for publication. The publications which contain work not related to or not included in this thesis are denoted by (\*).

#### Papers

- \*Mathieu, M.N., Wade, J.D., Catimel, B., Bond, C.P., Nice, E.C., Summers, R.J., Otvos Jr, L. & Tregear G.W. (2001). Synthesis, conformational studies and biological activity of N<sup>α</sup>-mono-biotinylated rat relaxin. Journal of Peptide Research, 57, 374-382.
- Bathgate, R.A.D., Samuel, C.S., Burazin, T.C.D., Layfield, S., Claasz, A.A., Reytomas, I.G.T., Dawson, N.F., Zhao, C., Bond, C.P., Summers, R.J., Parry, L.J., Wade, J.D. & Tregear, G.W. (2002). Human relaxin gene 3 (H3) and the equivalent mouse relaxin (M3) gene. *Journal of Biological Chemistry*, 277, 1148-1157.
- Claasz, A.A., Bond, C.P., Bathgate, R.A., Otvos Jr, L., Dawson, N.F., Summers, R.J., Tregear, G.W. & Wade, J.D. (2002). Relaxin-like bioactivity of ovine insulin 3 (INSL3) analogues. *European Journal of Biochemistry*, 269, 6287-6293.
- \*Tan, Y.Y., Dawson, N.F., Kompa, A.R., Bond, C.P., Claasz, A.A, Wade, J.D., Tregear, G.W. & Summers, R.J. (2002). Structural requirements for the interaction of sheep insulin-like factor 3 with relaxin receptors in rat atria. *European Journal of Pharmacology*, 457, 153-160.
- \*Samuel, C.S., Zhao, C., Bathgate, R.A., Bond, C.P., Burton, M.D., Parry, L.J., Summers, R.J., Tang, M.L., Amento, E.P. and Tregear, G.W. (2003). Relaxin deficiency in mice is associated with an age-related progression of pulmonary fibrosis. *FASEB Journal*, 17:121-123

- \*Samuel, C.S., Zhao, C., Bond, C.P., Hewitson, T.D., Amento, E.P. and Summers, R.J. (2004). Relaxin-1 deficient mice develop an age-related progression of renal fibrosis. *Kidney International*, accepted for publication.
- Bond, C.P., Parry, L.J., Samuel, C.S., Gehring, H.M., Lederman, F.L., Rogers,
  P.A.W. and Summers, R.J. (2004). Increased expression of the RLX receptor
  (LGR7) in human endometrium during the secretory phase of the menstrual cycle. *Journal of Clinical Endocrinology & Metabolism*, submitted for publication.

#### Papers/Book Chapters in Conference proceedings

- \* Mathieu, M.N., Bond, C.P., Wade, J.D., Summers, R.J. & Tregear, G.W. (2001).
  N<sup>α</sup>-mono-biotinylated rat relaxin: synthesis and activity. In: Relaxin 2000: Proceedings of the Third International Conference on Relaxin & Related Peptides. eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. p 239-240. Netherlands: Kluwer Academic Publishers.
- Tan, Y.Y., Dawson, N.F., Kompa, A.R., Bond, C.P., Claasz, A.A., Wade, J.D., Tregear, G.W. & Summers, R.J. (2001). Structure-activity relationships of relaxin and relaxin-like factor/insulin 3. In: *Relaxin 2000: Proceedings of the Third International Conference on Relaxin & Related Peptides.* eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. p 255-260. Netherlands: Kluwer Academic Publishers.
- Tan, Y.Y., Bond, C.P., Wade, J.D. Tregear, G.W. & Summers, R.J. (2001). Bioassays for relaxin and relaxin-related peptides. In: *Relaxin 2000: Proceedings of the Third International Conference on Relaxin & Related Peptides.* eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. p 271-272. Netherlands: Kluwer Academic Publishers.

- Bond, C.P., Tan, Y.Y., Tregear, G.W. and Summers, R.J. (2001). Determination of structural features for interaction with the relaxin receptor. Proceedings of the Australasian Society of Clinical & Experimental Pharmacologists & Toxicologists, 9, P1-23.
- Bathgate, R.A.D., Scott, D., Samuel, C.S., Burazin, T.C.D., Layfield, S., Claasz,
  A.A., Dawson, N.F., Ma, S., Bond, C.P., Summers, R.J., Wade, J.D. &
  Tregear, G.W. (2002). Relaxin 3, a novel member of the insulin/relaxin superfamily of peptide hormones. *Biology of Reproduction*, 66 (Supp 1), 249.
- \*Samuel, C.S., Zhao, C., Bathgate, R.A.D., Bond, C.P., Burton, M.D., Parry, L.J., Summers, R.J., Tang, M.L., and Tregear, G.W. (2002). Relaxin deficient mice develop an age-related progression of pulmonary fibrosis. *Proceedings* of the Matrix Biology Society of Australia & New Zealand, #29.
- Bond, C.P., Samuel, C.S., Parry, L.J., Zhao, C., Gehring, H., Rogers, P., Tregear, G.W. and Summers, R.J. (2002). Regulation of the relaxin receptor in the human uterus. Proceedings of the Australian Health & Medical Research Congress, #221.
- \*Samuel, C.S., Zhao, C., Bathgate, R.A.D., Bond, C.P., Burton, M.D., Parry, L.J., Summers, R.J., Tang, M.L., and Tregear, G.W. (2002). Relaxin: A potential therapy for pulmonary fibrosis. *Proceedings of the Australian Health & Medical Research Congress*, #2423.
- Bond, C.P., Parry, L.J., Samuel, C.S., Gehring, H., Rogers, P., and Summers, R.J. (2003). Expression of the relaxin receptor (LGR7) and related genes during the phases of the menstrual cycle. *Proceedings of the Endocrine Society of Australia*, #176.

- Halls, M.L., Bond, C.P., Bathgate, R.A. and Summers, R.J. (2003). Characterisation of sites on the LGR7 (relaxin) and LGR8 (insulin3) receptors required for binding and signal transduction. Proceedings of the Molecular Pharmacology of G-Protein Coupled Receptors, P6.
- Bond, C.P., Bathgate, R. and Summers, R.J. (2003). Comparison of the interaction of relaxin with LGR7 & LGR8 receptors. Proceedings of the Australasian Society of Clinical & verimental Pharmacologists & Toxicologists, 10, 125.

## Abbreviations

ATP	adenosine 5'-triphosphate
BDM	2,3-butanedione monoxime
BHK cells	baby hamster kidney cells
bp(s)	base pair(s)
BSA	bovine serum albumin
cAMP	cyclic AMP
CHO-K1 cells	chinese hamster ovary cells
DMSO	dimethyl sulfoxide
DNA	deoxyribonucleic acid
DTT	dithiothreitol
EDTA	ethylenediaminetetraacetic acid
FBS	foetal bovine serum
HIRLX	human gene 1 relaxin
H2RLX	human gene 2 relaxin
H3RLX	human gene 3 relaxin
HBS	hepes buffer saline
HEK293T cells	human embryonic kidney cells
IBMX	3-isobutyl-1-methylxanthine
INSL3	insulin 3
LGR7	leucine rich-repeat g-protein coupled receptor 7
LGR8	leucine rich-repeat g-protein coupled receptor 8
OVLT	organum vasculosum of the lamina terminalis
PBS	phosphate buffered saline
PCR	polymerase chain reaction
PEG	polyethylene glycol
PEI	polyethylenimine
PMSF	phenylmethylsulphonyfluoride
PPS marker	precision protein standard marker
PRLX	porcine relaxin
RatRLX	rat gene 1 relaxin
Rat3RLX	rat gene 3 relaxin

xvii

RLX	relaxin
RMRLX	rhesus monkey relaxin
RNA	ribonucleic acid
SDS	sodium dodecyl sulfate
TBS	tris buffered saline
TBS-T	TBS tween 20
TFA	trifluroacetic acid
THP-1 cells	human monocyte cell line
vmDTg	ventromedial dorsal tegmental nucleus

# **Chapter One**

## **General Introduction**

#### 1.1 History of relaxin

In the early 1920's, F.L. Hisaw investigated the physiological changes that occur to the pubic symphysis during pregnancy in guinea pigs, including the relaxation of the pubis and the increase of connective tissue at the symphysis which caused separation of the bones. He reported that when virgin female guinea pigs, close to oestrus, were injected with serum taken from pregnant rabbits, there was relaxation of the pubic ligaments similar to that naturally occurring in pregnancy (Hisaw, 1926). The substance responsible for the relaxation response could be obtained from rabbit placenta and from the corpora lutea of sows, and was found to work in conjunction with oestrin (oestrogen). In 1930, a crude extract of this substance was extracted from the corpus lutea of sows, identified as a hormone, and given the name "Relaxin" (Fevold *et al.*, 1930).

Over the subsequent several decades, interest in relaxin lessened due to limited resources for the purification and testing of biological extracts. During this time however, there were several important findings made with impure porcine relaxin extracts. Relaxin was shown to inhibit spontaneous contractions of oestrogen-primed guinea pig myometrium (Krantz *et al.*, 1950) and to promote elongation of the interpubic ligament (Hall, 1947), identifying relaxin as a key hormone of pregnancy.

In the early 1970's, the successful purification of porcine relaxin (PRLX) from frozen sow ovaries was reported (Sherwood *et al.*, 1974), with the extract displaying strong relaxin-like biological activity in the mouse interpubic ligament bioassay (see Section 1.3.1.2). This led to the isolation and purification of relaxin from the ovaries, corpora lutea or placentae of various pregnant animals, including rats, sharks, whales, horses, dogs and rabbits (Sherwood, 1994), and the cloning of human relaxin in the 1980's (Hudson *et al.*, 1983).

#### 1.2

#### Characterisation of Relaxin

#### 1.2.1 Structure and molecular characterisation of relaxin

Early modelling of the relaxin structure, predicted from purified PRLX, indicated that the peptide was comprised of two polypeptide chains, an A chain and a B chain of 22 and 32 amino acids respectively (Schwabe et al., 1976; Schwabe et al., 1977a) with a molecular weight of around 6000 Daltons. The two chains were linked via two disulphide bridges, with an intra-chain disulphide bridge in the A-chain. When relaxin was first discovered, it shared very similar structural properties to insulin (Fevold et al., 1930). Although PRLX contained only 11 residues that were identical to porcine insulin, the preservation of the hydrophobic residues and the position of the disulphide bridges was consistent with the structure of porcine insulin (Bedarkar et al., 1977; Schwabe et al., 1977b). This strengthened the suggestion that relaxin had structural homology with insulin and insulin-like growth factors (IGFs), and led to the successful modelling of relaxin into an insulin-like tertiary structure (Bedarkar et al., 1977; Isaacs et al., 1978) (Figure 1.1). The hypothetical structure was confirmed by an x-ray crystal structure of human relaxin (Eigenbrot et al., 1991). While the x-ray crystal structure of relaxin highlighted that like insulin relaxin crystallises into a dimer, the orientation of the dimers and binding interfaces were completely different therefore suggesting that similarities in structure do not necessarily lead to similarities in receptor binding mechanisms.

With a similar structure to insulin, it was predicted that relaxin would have a precursor protein resembling pro-insulin (Snell *et al.*, 1975). Prepro-relaxin was shown by nucleotide sequence analysis of the cDNA to consist of a signal peptide, B-chain, connecting peptide and A-chain (Reddy *et al.*, 1992). The biosynthetic pathway of relaxin involves prepro-relaxin undergoing enzymatic cleavage by prohormone convertase (Layden *et al.*, 1996) of the signal peptide as it is translocated across the endoplasmic reticulum to produce pro-relaxin. Pro-relaxin is further cleaved at the A-chain/connecting peptide and B-chain/connecting peptide junctions to form relaxin (Reddy *et al.*, 1992).

## Figure 1.1

## Schematic structure of human insulin and human relaxin peptides.

As with all members of the insulin gene superfamily, the insulin and relaxin peptides are comprised of two separate peptide chains, and A and a B chain.







#### 1.2.2 Relaxin and the insulin gene superfamily

Although the active form of relaxin in all species contains the A & B chain configuration, there is a large range of peptide sequence diversity between species homologues of relaxin. However the cysteine residues forming the disulphide bonds linking the two chain structure are highly conserved (Figure 1.2). Since the original cloning of the human relaxin gene in the 1980s (Hudson et al., 1983; Hudson et al., 1984), it was known that the human genome expresses two relaxin genes, human gene 1 (HIRLX) and human gene 2 (H2RLX) relaxin. The H2RLX peptide is predominantly expressed in the ovaries (Hudson et al., 1984) but can also be locally expressed in a variety of other tissues, including mammary glands (Bongers-Binder et al., 1991; Eddie et al., 1989) and the prostate gland (Sokol et al., 1989). The distribution of tissues expressing H2RLX mRNA however, is greater still than those that express the relaxin protein, and includes the cardiovascular and central nervous systems (Section 1.4). The expression and relevance of HIRLX is unclear although it is biologically active (Tan et al., 1998; Wade et al., 1996). However it was recently discovered that the human genome contained a third relaxin (Bathgate et al., 2002), human gene 3 relaxin (H3RLX) and the expression and activity of this peptide is still under investigation. Until 2002, it was thought that most species only expressed one relaxin gene, equivalent to H2RLX, with only primates such as chimpanzee, gorilla, orang-utan and humans expressing two relaxin genes, equivalent to H1RLX and H2RLX. With the isolation of H3RLX came the discovery of an equivalent gene in mice and rats, termed M3RLX (Bathgate et al., 2002) and Rat3RLX (Burazin et al., 2002) respectively due to their homology with H3RLX.

Relaxin is a member of the insulin gene superfamily due to the structural similarities to insulin and insulin-like growth factors (Figure 1.3). The human insulin gene scherfamily is made up of a growing number of hormones that regulate cell growth and metabolism, including insulin, insulin-like growth factor I (IGF-I), insulin-like growth factor II (IGF-II), Leydig cell insulin-like protein (Ley-IL or INSL3), early placenta insulin-like peptide (EPIL or INSL4), insulin-like peptide 5 (INSL5), testis-

Chapter 1: General Introduction

にするが理想者は利用用は「は、1000年間は1000年間には、1000年間に

6

derived relaxin-like factor (INSL6) and the relaxin peptides (H1, H2 & H3RLX). The insulin-like peptides have structural features such as highly conserved cysteine residues that form disulphide bonds between the two peptide chains (Figure 1.3), and are essential for cell growth and differentiation, development and energy regulation.

INSL3, also known as relaxin-like factor (RLF), is the member of the insulin-gene superfamily most closely related to relaxin. The close relationship between INSL3 and relaxin is of interest since ruminants, such as sheep and cows, do not produce an active relaxin due to the numerous stop codons in the sequence corresponding to the C-peptide region (Roche *et al.*, 1993) yet respond to relaxin treatment (Porter *et al.*, 1981) indicating the presence of a receptor that can recognize relaxin. It has been suggested that ruminant INSL3 is a replacement for relaxin as INSL3 is produced in the ovary of ruminants, similar to relaxin in other mammalian species, although there has been no confirmation of this suggestion (Bathgate *et al.*, 2001).

#### Figure 1.2

#### The amino acid sequences of species homologues of relaxin.

A comparison of the amino acid sequence from a variety of species homologues of the relaxin peptide. The cysteine (C) residues that form the disulphide bonds (blue) are aligned and highlighted (pink). The arginine (R) and isoleucine (I) residues important for receptor binding (green) are situated in the B-chain.

Chapter 1: General Latroduction

<u>A Chain</u>																																					
				ł				5				10	0 1						5 20																		
HIRLX				R	₽	Y	v	А	L	F	E	к			L	I	G		т	κ	R	s	L	х	ĸ	¥	5										
H2RLX				0	L	Y	S	A	L	Ā	N	ĸ		ć.	R	v	G		Т	ĸ	R	8	Ľ	A	R	F	1									•	
H3RLX				D	v	L	A	G	L	S	\$	8		<u>.</u>	ĸ	W	G	:	8	ĸ	8	E	I	8	8	L	ł										
Porcine						R	м	T	L	s	E	ĸ		ġ	Q	v	G		r	R	ĸ	D	I	A	R	L	Į										
Rat 1				Z	8	Ģ	A	L	L	8	E	Q		÷	H	I	G		T	R	R	3	I	A	ĸ	L											
Rat 3				Ð	V	L	A	G	L	8	3	S			e	W	G	l	8	ĸ	g	Q	1	8	9	I,											
Mouse 1				E	\$	G	G	L	М	9	Q	Q			H	۷	G		8	R	R	3	I	A	x	L	Y	·									
Mouse 3				D	V	L	A	Ģ	L	8	S	S			E.	W	Ģ		S	ĸ	\$	Q	I	8	8	L											
Guines pig				2	L	D	M	Т	V	8	E	ĸ			Q	۷	Ģ	I	T	R	R	F	I	A	N	9											
Horse								Z	L	8	Ħ	ĸ			X	W	G		T	R	ĸ	E	L	A	R	Q											
Rabbit				Z	F	\$	Ē	8	L	₽	B	Ē			K	¥	Ģ		P	R	X	Y	L	L	M	¥	ŀ										
Rhesus monkey				Z	L	X	M	Т	L	8	N	ĸ			Ħ	I	G		T	ĸ	ĸ	8	L	A	ĸ	F	ľ										
Gorilla 1				R	₽	¥	V	A	L	F	E	ĸ			L	I	Ģ	I	T	K	R	S	L	λ	ĸ	Y	:.										
Gorilla 2				2	L	Y	S	A	L	A	N	ĸ			K	v	G		T	Ķ	R	8	L	A	R	F	÷.										
																											1										
<u>B Chain</u>																																					
				ł				5				;	10					15					20			1		25					30				
HIRLX				•	ĸ	w	к	D	ס	v	T	ĸ	L I		a		R	L	v		x	0			т	5	G (	M	<b>9</b>	т	W	8	R	R	8	ь	
H2RLX				D	s	W	M	E	2	v	ī	ĸ	L		G		E	L	v		λ	ō		A	ī		Ġ.	M	8	T	W	8	ĸ	R	s	Ľ	
H3RLX				-	R	Ä	A	₽	Y	G	v	R	L		G		E	F	I		A	v		F	T	i.	Ģ	G	8	R	W	-			-	-	
																_				_			-			_		•									
Porcine					Q	\$	т	N	D	F	r	к	A		G		É	L	v	E	L	W	Ā	E.	I	8	G	8	V.	8	W	Ģ	R	T	A	L	
Rati	R	V	8	E	E	W	M	D	Q	V	I	Q	v	i.	G	ŀ	G	Y	λ		λ	W		Ľ	۷		G	X	\$	۷	Ģ	R	L	A	<b>L</b>		
Rat3					R	₽	A	₽	¥	G	V	ĸ	L		G		Ż	F	I		Å	۷		F	Ť		G	G	<b>S</b> :	R	W						
Mousel	R	۷	S	Е	Ē	W	M	Þ	G	F	I	R	М	`. '	G		R	Y	Α	ł	E	L		ĸ	Ī	2	G.	A	<b>S</b> '	V	Ģ	R	L	A	L		
Mouse3					R	₽	A	P	T	G	۷	ĸ	L		G		E	F	I		A	v		F	Т	ł	G	G	<b>S</b> :	R	W						
Guines pig					G	F	L	D	ĸ	V	I	ĸ	۷	:	G		D	L	v		I	ĸ		D	I		G	K	I	L	L	G	D	М	T	Т	G
Horse					Q	ĸ	₽	Þ	D	V	I	к	A	. I	G		Z	L	λ	ŧ	L	R		E	I		G	8	L	S	W	ĸ					
Rabbit	R	V	T	Y	E	W	M	M	E	N	v	ĸ	I		R	N	D	F	V		T	A		E	V		Ģ	H	v	H	L	E	R				
Rhesus Monkey				A	ĸ	W	M	D	Ð	v	I	K	A		G		E	Ľ	V	÷	A	Q		A	I		Ģ	ĸ	9	T	L	G	ĸ	R	S	L	
Gorilla 1				A	ĸ	W	ĸ	D	D	V	I	K	L		G	C	E	L	V		A	Q		λ	I		G	M	8	T	₩	\$	ĸ	R	8	L	
Gorilla 2				D	3	W	ĸ	D	D	V	I	ĸ	L		G		E	L	V		¥	Q	E	X	I		G.	M	9	T	W	\$	ĸ	R	8	L	

7

## Figure 1.3

### The amino acid sequences of the human insulin-gene super family.

All members of the human insulin-gene superfamily contain an identical positioning of the cysteine (C) residues (pink) that form the connecting disulphide bond (blue). The  $\sim$  indicates the sequence is still continuing.

<u>A Chain</u>																																		
		1				5					10	10				15					20													
HIRLX		R	₽	¥	v	A	L	F	Ē	ĸ			L	I	Ģ		т	ĸ	R	s	L	A	ĸ	Y	2									
H2RLX		Q	L	Y	8	A	L	A	N	ĸ			н	۷	G		т	ĸ	R	9	L	λ	R	F	Į.									
H3RLX		D	V	L	A	G	L	8	8	8			ĸ	W	G		8	ĸ	8	E	I	8	9	L	Ľ									
Insulia						G	I	v	E	õ			T	8	I		8	L	Y	Q	L	E	N	Y		N								
IGF1	~	A	₽	Q	т	G	I	v	Ď	E	•.		F	R	\$		D	L	R	R	L	E	м	Y	ľ	A	•							
IGF2	~	R	R	8	R	G	I	v	Ē	E			F	R	8		Ď	L	A	L	L	E	T	Y		А	•							
INSL3		A	A	λ	T	N	₽	A	Ŕ	¥	2		L	S	Ģ		T	Q	Q	D	L	L	Т	L		~								
INSLA	R	\$	Ģ	R	H	R	F	D	₽	F			E	V	I		Ø	Ď	Ģ	T	\$	v	ĸ	L		~								
INSL5	~	M	s	R	Q	D	L	Q	Ť	L			Т	D	G		8	M	T	D	I,	s	A	L										
INSL6	~	R	ĸ	R	R	G	Y	8	E	ĸ			L	Ŧ	G		T	ĸ	E	z	I,	8	I	A		~								
<u>B Chain</u>	I				5					10					15					20	ł		/	/	25					30				
HIRLX		ĸ	Ħ	ĸ	D	D	v	I	ĸ	L		G	R	E	L	۷	R	A	Q	I	A	r		Ģ	М	8	T	W	8	x	R	\$	L	
H2RLX	Þ	S	W	M	E	E	V	I	ĸ	L	ľ	G	R	E	Ţ,	۷	R	A	Q	I	A	I		G	M	\$	T	W	9	ĸ	R	8	L	
H3RLX		R	A	A	P	Y	G	V	R	Ļ		G	R	E	F	I	R	A	V	I	F	T		G	G	\$	R	W						
Insulin					F	v	N	Q	ы	L		G	8	H	L	v	E	A	L	Y	L	v	đ	G	E	R	G	F	F	¥	T	₽	ĸ	T
IGFI						G	₽	Ë	·۳	L		G	A	E	L	V	Э	A	I,	Q	F	v	•	G	D	R	Ģ	F	Y	F	N	ĸ	₽	T
IGF2				¥	R	₽	8	Ē	T	L	ł	G	Ģ	R	L	V	D	T	I,	Q	F	v		G	₽	R	G	F	¥	F	8	R	₽	λ
INSL3		₽	T	₽	£	М	R	E	K	L	÷	Ġ	H	H	F	V	R	A	I,	V	R	۷		Ģ	G	₽	R	W	8	Т	E	A		
INSL4		Ë	8	L	A	λ	Ľ	L	R	G		G	₽	R	F	Ģ	ĸ	H	I,	L	8	Y		₽	м	₽	E.	ĸ	Т	F	T	T	T	p.
INSL5	~	۷	R	8	ĸ	E	S	۷	R	L		G	L	3	X	1	R	T	V	I	¥	I		A	8	S	R	W	R	R	H	L	E	G
INSL6	~	S	D	I	S	8	А	R	ĸ	L		Ģ	R	Y	×.	V	ĸ	E	I	E	ĸ	L		G	н	A	N	W	\$	Q	F	R	F	E.

.`

Chapter 1: General Introduction

#### 1.2.3 Sources of relaxin

The primary source of relaxin synthesis varies slightly between species, being the corpus luteum in rodents, pigs, cows and humans, and the placenta in cats, dogs, horses and rabbits (Sherwood, 1994), with the highest levels detected during pregnancy. Local synthesis of relaxin has been identified in other reproductive tissues, including ovaries, uterus and mammary glands, as well as in a variety of non-reproductive tissues (see Section 1.4) suggesting a number of roles for relaxin.

#### **1.2.3.1** Extraction of native relaxin

Early methods of extraction of relaxin from sow corpora lutea involved acid alcohol extraction, sodium chloride crystallisation and precipitation with picric acid (Fevold *et al.*, 1930), to yield approximately 30 mg of roughly purified hormone per kg corpora lutea. Forty years later developments in technology allowed highly purified PRLX to be prepared. This involved extraction and precipitation of protein from sow ovaries with acid-acctone, gel filtration on a Sephadex G-50 followed by further purification and concentration of the relaxin peptide with ion exchange chromatography on CM-cellulose (Sherwood *et al.*, 1974), yielding approximately 35 mg of a highly pure PRLX preparation per kg ovary. The purification of PRLX led to the determination of the physicochemical properties and the successful extraction of relaxin from other species including rat (Sherwood, 1979), sand tiger shark (Reinig *et al.*, 1981), rabbit (Eldridge *et al.*, 1985) and horse (Stewart *et al.*, 1986).

#### **1.2.3.2** Synthetic production of relaxia peptides

As extraction of relaxin from tissue sources resulted in small yields and since human relaxin could not be easily prepared by this method, synthetic methods were developed. The synthetic production of relaxin peptides has also allowed investigation of the structure-activity relationships of relaxin. The first relaxin peptide to be synthesised was PRLX (Tregear *et al.*, 1982), followed by H1RLX

Chapter 1: General Introduction

(Hudson et al., 1983) and H2RLX (Hudson et al., 1984). Further developments to the synthesis of relaxin have resulted in the current methodology which includes the solid-phase synthesis of the separate A and B chains, purification of the crude peptides, combination of the A and B chains and folding under oxidative conditions with the final product analysed for purity and correct folding of the peptide (Büllesbach et al., 1994a; Büllesbach et al., 1991; Rembiesa et al., 1993; Tang et al., 2003). However, synthetic production of relaxin peptides is not without its problems, with low yields of peptide due to substantial losses during the combination of A and B chains as a result of the poor solubility of the B chain (Büllesbach et al., 1994a; Tregear et al., 1995). The synthetic conditions for each relaxin peptide often differs due to the complexity and variability of the individual chain sequences, thus increasing the production time (Tregear et al., 2001).

#### **1.2.3.3** Production of relaxin by recombinant methods

To obtain amounts of relaxin peptide larger than that available by native extraction or synthetic means, recombinant methods were developed. Early methods included the combination of natural mRNA and synthetic DNA constructs to produce pro-PRLX in an E.coli culture (Stewart et al., 1983). This was developed further with the incorporation of pro-PRLX DNA template into an E.coli expression vector (Reddy et al., 1992), leading to the successful production of a pro-PRLX peptide. The pressure to develop a large scale, cost effective method for production of recombinant human relaxin was increasing due to the demand for use in clinical trials. Methods for recombinant production of human relaxin were developed by Genentech Inc. (CA, USA) and involved the separate production of the relaxin A and B chains in E.coli followed by the extraction and purification of large quantities of peptide (Canova-Davis et al., 1991). The purified chains were combined and refolded with yields of approximately 60% relative to the amount of B chain incorporated into the reaction. A further advance was made with the development of a one chain process with the A and B chains connected by a "mini-C" peptide to produce a pro-relaxin form of the H2RLX peptide (Vandlen et al., 1995), a method successfully employed in the production of recombinant mini-C pro-insulin (Wetzel et al., 1981). The extracted mini-C pro-H2RLX peptide was then refolded and the mini-C peptide cleaved. The

synthesis with the mini-C peptide allowed correct folding of the recombinant peptide.

#### 1.2.4 Structure-activity relationships of relaxin

The structural features of relaxin that defined the affinity and efficacy of the peptide and its receptor were examined by bioassays (see Section 1.3). The large range of relaxin sequence diversity between species has highlighted the importance of the highly conserved residues and researchers have exploited these structural features to investigate their role in relaxin activity.

#### 1.2.4.1 A Chain

While the A chain appeared to have little direct interaction with the receptor due to its distal position to the B chain binding site (Eigenbrot *et al.*, 1991), its importance was established in maintaining structural integrity of the relaxin peptide complex. Shortening of the A chain by the removal of 3 or more amino acids from the N terminus caused a structural change to the relaxin peptide, leading to a subsequent loss of relaxin bioactivity in the mouse interpubic ligament assay (Büllesbach *et al.*, 1986b). Further investigation revealed that the identity of the N terminal amino acids is not as important for maintaining the structural integrity of the peptide as the presence of an N terminal helix (Büllesbach *et al.*, 1987). The importance of the A chain in the relaxin peptide was highlighted by the analysis of the activity of the separate A and B chains, revealing that the individual chains have no biological activity (Køkouris *et al.*, 1992; Tan *et al.*, 1998).

Substitution of highly conserved basic amino acids in the relaxin A chain with citrulline residues had no effect on the biological activity of the peptide (Rembiesa *et al.*, 1993), while substitution of the glycine (A14) residue with isoleucine dramatically reduced relaxin bioactivity by 100 fold (Büllesbach *et al.*, 1994a). Disruption of the internal A chain disulfide bond by substitution of the cysteine (A10 & A15) residues with alanine resulted in a significant decrease in biological activity (Büllesbach *et al.*, 1995a). While the above substitutions of A chain residues did

20月前になるので、二部に下いた。19月1日、1月前が回復国際地域ないのないないないで、19月1日日、19月1日、19

affect relaxin-like bioactivity, it was concluded that the decrease in bioactivity was indicative of the disruption of structural integrity of the relaxin peptide rather than a role of these amino acids in interactions with the receptor.

#### 1.2.4.2 B Chain

Analysis of PRLX purified from pregnant sow ovaries revealed three forms of the hormone that differed in the length of the B chain at the carboxyl terminus (Sherwood *et al.*, 1974). The native form of PRLX was believed to have a B chain of 31 amino acids, while versions with a shortened B chain of either 29 or 28 amino acids were attributed to partial degradation during the isolation and purification process (Walsh *et al.*, 1980), although all three forms of PRLX were bioactive (Anderson, 1984; Sherwood *et al.*, 1974). This led to the examination of the role of the length of the B chain. Synthetic PRLX with three or seven amino acids removed from the N-terminus and C-terminus respectively showed no decrease in relaxin bioactivity, whereas the removal of two further amino acids from the C-terminus reduced relaxin bioactivity (Tregear *et al.*, 1983; Tregear *et al.*, 1982).

Based on the three-dimensional structure of insulin and with the completion of the xray crystallography of relaxin (Eigenbrot *et al.*, 1991), it was proposed that two highly conserved arginine (B13 & B17) residues in the B chain, which protrude into the surrounding water, would be crucial for interaction with the receptor (Büllesbach *et al.*, 1988). Substitution of these residues, separately and together, with either positively charged lysine or uncharged citrulline dramatically reduced receptor interaction and bioactivity of the relaxin peptide (Büllesbach *et al.*, 1991; Büllesbach *et al.*, 1992), indicating the importance of the arginine residues. However the specificity with which the altered relaxin peptides still interacted with the receptor suggested that there were other amino acids also involved in relaxin binding. Comparison of the known relaxin sequences from a variety of species yielded three highly conserved amino acids within the B chain binding site area; glycine B12, valine B16 and isoleucine B20 (Figure 1.2). Manipulation of these amino acids resulted in the exclusion of glycine B12 and valine B16 as active sites, and the unveiling of isoleucine (B20) as a critical component of relaxin binding (Büllesbach *ci al.*, 2000). The relaxin binding "cassette", comprised of Arg-X-X-Arg-X-X-Ile, is positioned along the B chain alpha-helix allowing the arginine and isoleucine side chains to protrude and interact with the relaxin receptor in a unique "lock and key" mechanism.

#### 1.3 Bioassays for relaxin

The original observations that relaxin caused relaxation of the pubic symphysis, elongation of the interpubic ligament and myometrial relaxation, led to the development of the three bioassays used to determine relaxin activity for more than 30 years. They include the guinea pig pubis symphysis palpation assay, mouse interpubic ligament assay and the uterine relaxation assay.

#### 1.3.1 In vivo functional assays

#### 1.3.1.1 Guinea pig pubis symphysis palpation assay

The first relaxin bioassay was based on the ability of relaxin to elongate the pubic symphysis of the guinea-pig (Fevold et al., 1930). Virgin guinea-pigs in full oestrus were injected subcutaneously with a relaxin preparation. Ten to twelve hours later the pelvic ligaments were gently palpated vertically while holding the symphysis pubis stationary. The original guinea pig unit (GPU) was taken as the minimum amount of hormone needed to loosen the ligaments within ten to twelve hours (Fevold et al., 1930). To ensure a quantitative assay, the definition of the GPU was redefined as the amount of hormone required to induce relaxation of the symphysis pubis within six hours in 70% of castrated female guinea pigs weighing between 350 - 800g, pre-treated daily with 0.85 µg oestradiol for 4 days (Abramowitz et al., 1944). Although the assay now maintained a quantitative aspect, the overall validity of the assay had not improved due to the subjective nature of the analysis and the desensitization of tissues when relaxin was used repeatedly (Noall et al., 1956; Steinetz et al., 1982). To allow for a more objective assessment of relaxin activity, the palpation analysis was performed by two or more independent researchers under blind conditions and scored on a flexibility scale from 0 indicating no flexibility to 6
showing extreme flexibility and compared against a reference standard (Kroc *et al.*, 1959). Although useful for its ability to show the activity of relaxin from a variety of species (Büllesbach *et al.*, 1986a; O'Byrne *et al.*, 1978; Reinig *et al.*, 1981), the guinea pig pubis symphysis palpation assay still remains imprecise and expensive with 20 or more animals per dose needed to reduce error limits to less than 50% (Büllesbach *et al.*, 1994b; O'Byrne *et al.*, 1978; Reinig *et al.*, 1981; Steinetz *et al.*, 1969).

# 1.3.1.2 Mouse interpubic ligament elongation assay

The mouse interpubic ligament elongation assay was the first bioassay to be standardized and accepted as a relaxin bioassay that could be used for comparative studies without the large discrepancies that occurred in the guinea pig pubis symphysis palpation assay. Oestrogen-primed mice were subcutaneously injected with relaxin or vehicle, killed 18-24 hours later, the interpubic ligament dissected and measured against reference standards (Steinetz *et al.*, 1960). This bioassay allowed the reproducible assessment of relaxin activity to be determined in a dose-dependent manner and has been extensively used to characterise the activity of relaxin and synthetic analogues (Büllesbach *et al.*, 1996; Büllesbach *et al.*, 1990; Büllesbach *et al.*, 1993; Sherwood, 1994). Its simplicity and specificity make this bioassay more appealing, however it still has the same drawbacks as the guinea pig pubis symphysis palpation with regard to variation among animals, needing 20 mice per dose to reduce the error limits to less than 50%.

# 1.3.2 In vitro functional assays

### 1.3.2.1 Uterine contractility/relaxation assay

The *in vitro* uterine contractility or relaxation assay are essentially two complementary versions of the same bioassay, modified from an original *in vivo* bioassay based on the observation that relaxin extracts could decrease spontaneous uterine contractions in guinea pigs (Krantz *et al.*, 1950). The contractility bioassay utilises the spontaneous contractions produced in isolated uterine tissue from guinea pigs or rodents, which diminish in the presence of relaxin (Anderson, 1984; MacLennan *et al.*, 1991; Oliver *et al.*, 1978; Petersen *et al.*, 1991). The advantage of this bioassay over the *in vivo* bioassay is that it is sensitive and the relaxin response is immediate rather than over several hours. However the ability to quantitate relaxin activity is not easy since both the frequency and amplitude of the contractions is not reliable, making a stable baseline hard to establish. Endogenous peptides, such as oxytocin, have also been shown to interfere with this assay (Fields *et al.*, 1983).

The uterine relaxation bioassay removes the dependence on the spontaneous contractions by contracting the tissue artificially using a low dose of prostaglandin  $F_2$  alpha (PGF<sub>2α</sub>) (Nishikori *et al.*, 1983), electrical stimulation (Brenner *et al.*, 1984b) or a depolarizing potassium solution (St-Louis, 1982) and then relaxing the tissue with increasing concentrations of relaxin. Taking out the spontaneous component of the bioassay has allowed a more accurate and reliable quantitation of relaxin activity to be achieved.

# 1.3.2.2 Rat isolated atrial bioassay

More recently the rat isolated atrial bloassay was developed based on the observation that relaxin binding sites were described in the atria (Osheroff *et al.*, 1992; Tan *et al.*, 1999). Both lobes of the heart atria were isolated from the ventricles, the right

なるとは、「ある」なるなどに、「ある」の言葉がない。「「「「な」」の「ない」のない。「ない」の「ない」ので、「ない」のない。「ない」のない。「ない」のない。「ない」のない。「ない」のない。「ない」のない。

atria allowed to beat spontaneously to allow the determination of drug induced chronotropic effects, while the left atria was stimulated and the force of contraction measured to allow positive inotropic effects to be studied (Kakouris *et al.*, 1992; Tan *et al.*, 2002; Tan *et al.*, 1998; Ward *et al.*, 1992). Relaxin caused a slow developing, dose-dependent positive chronotropic and inotropic response which stabilised within 5 minutes. The response to relaxin was the result of a direct interaction between relaxin and its receptor, without the need of steroid treatment pre-priming, and there was no difference in results obtained in atria isolated from male or female rats (Kakouris *et al.*, 1992). The isolated atrial bioassay has proven to be a sensitive, reproducible and efficient bioassay, as an entire concentration-response curve to relaxin can be achieved using a single animal, therefore decreasing animal usage.

# 1.3.2.3 Cellular functional assays

More sensitive and precise bioassays were needed to accommodate the low levels of relaxin available from tissue extracts and the rapid advances in peptides synthesised for relaxin structure-activity relationships. Relaxin caused z dose-dependent increase in cAMP in newborn rhesus monkey uterus cells (Kramer *et al.*, 1990), rat myometrial cells (Hsu *et al.*, 1985) and human endometrial cells (Fei *et al.*, 1990; Osheroff *et al.*, 1990). The bioassays were sensitive, specific and the cAMP response to relaxin could be inhibited by the addition of anti-relaxin antibodies (Fei *ei al.*, 1990; Kramer *et al.*, 1990). Limitations of these primary cell cultures include a relatively short life span, a scarcity of starting tissues and a degree of variability between cultures. These problems were solved by the use of a human monocyte cell line, THP-1, that naturally expresses relaxin receptors. Relaxin stimulates a cAMP response in THP-1 cells by binding specifically to receptors on the cell surface (Parsell *et al.*, 1996).

# 1.4 The physiological actions of relaxin

The actions of relaxin on the pregnant female reproductive tract were responsible for its discovery and isolation in 1926 (Hisaw, 1926), and for many years functional research focused on the role of relaxin in female reproductive organs. However, relaxin is becoming more than just a hormone of pregnancy, with research identifying a variety of functions in both reproductive and non-reproductive organs (Table 1.1).

### **1.4.1** Female reproductive tissues

The discovery of relaxin as a reproductive hormone has led to extensive study of its role in the rat and porcine reproductive systems. The primary source of circulating relaxin in most species is the corpus luteum and radioimmunoassays have determined serum levels to be highest during pregnancy (Sherwood *et al.*, 1980). The serum levels of relaxin vary slightly between species. The general pattern is that levels remain low during the first and second trimesters of pregnancy but dramatically increase in the third trimester before falling rapidly following birth (Sherwood *et al.*, 1975a; Sherwood *et al.*, 1980; Stewart *et al.*, 1992), correlating with the influence of relaxin on parturition.

The development of an anti-relaxin antibody allowed researchers to investigate the actions of relaxin in the reproductive process. The elimination of circulating relaxin caused disruption of birth and a decrease in the delivery of live rat pups (Hwang *et al.*, 1989), hindered development of mammary glands and nipples (Hwang *et al.*, 1991), and reduced growth and softening of the cervix and vagina (Zhao *et al.*, 1996). Similar results were found in pigs (Cho *et al.*, 1998; Min *et al.*, 1997; Winn *et al.*, 1994). The development of a relaxin gene knock out mouse that cannot express a functional relaxin (Zhao *et al.*, 1999) has allowed a detailed investigation of the physiological role of relaxin and is described in Section 1.4.6.

Direct administration of relaxin to isolated myometrial tissue from rats caused a dose-dependant inhibition of contractile activity (Petersen et al., 1991), as well as reducing oxytocin induced spasm by preventing the release of Ca2+ from intracellular stores and the influx of extracellular Ca<sup>2+</sup> (Hughes et al., 1995). Similar results were found with isolated pig myometrial tissue, where exposure to relaxin caused a dose-dependent decrease of spontaneous contractility (MacLennan et al., 1991; MacLennan et al., 1995; MacLennan et al., 1986a). It has been demonstrated in myometrial tissue obtained from a non-pregnant rat that relaxin receptors are regulated by oestrogen (Mercado-Simmen et al., 1982). However, during pregnancy, the concentration of relaxin receptors determined in rat myometrial membrane-enriched fractions using [125] dorelaxin do not correlate with oestrogen levels. Administration of exogenous relaxin significantly decreased relaxin receptor expression in the myometrium during late pregnancy (Mercado-Simmen et al., 1982), which could account for the lack of relaxin activity seen in myometrial tissue obtained from late pregnancy in some studies (Longo et al., 2003).

In humans however, the role of relaxin is uncertain with relaxin exhibiting less influence on remodelling for parturition but appearing to be more important in the errly stages of pregnancy. Relaxin binding sites have been shown in both the endometrial and myometrial layers of the human uterus, although functional studies indicate that relaxin fails to inhibit the spontaneous contractility of human myometrial strips (MacLennan *et al.*, 1991; MacLennan *et al.*, 1995; MacLennan *et al.*, 1986a; Petersen *et al.*, 1991). Plasma levels of relaxin rise dramatically upon implantation, remain high for the first trimester and then steadily decrease until parturition (Eddie *et al.*, 1986). Relaxin appears to have little influence on the cervix as clinical trials reported that topically applied relaxin failed to affect any aspect of cervical ripening compared to controls (Brennand *et al.*, 1997), even though the cervix has been shown to contain relaxin binding sites (Kohsaka *et al.*, 1998). One study reported a positive correlation between relaxin serum levels and cervical growth during the early stages of the second trimester of pregnancy (Eppel *et al.*, 1999), with little correlation found at term.

The function of relaxin in a non-pregnant menstrual cycle appears to be influenced by the phase of the menstrual cycle. Relaxin in the uterus is thought to be influenced by progesterone and regulation of relaxin secretion is cycle dependent (Yki-Jarvinen *et al.*, 1985). Relaxin differentially regulates the secretion of vascular endothelial growth factor (VEGF) in isolated uterine cells with a negative influence in the proliferative phase, while in the secretory phase, relaxin treatment stimulated VEGF secretion (Palejwala *et al.*, 2002). The positive effect of relaxin on secretory phase VEGF secretion is thought to be responsible for an increased incidence of menometrorrhagia in relaxin-treated clinical trial patients (Unemori *et al.*, 1999)

### 1.4.2 Male reproductive tissues

Relaxin was first identified in the male when a crude extract taken from rooster testis demonstrated specific relaxin-like activity in the mouse pubis symphysis bioassay that could be inhibited with rabbit anti-PRLX antiserum (Steinetz *et al.*, 1964). The presence of an immunoreactive relaxin-like protein was readily identified in human male seminal plasma (Loumaye *et al.*, 1980), although the amount of this protein varied between the samples. The peptide was confirmed as relaxin and identified as a likely product of the prostate gland (Essig *et al.*, 1982). Expression of both H1RLX and H2RLX mRNA transcripts were identified in the prostate (Hansell *et al.*, 1991) however only H2RLX peptide was identified (Winslow *et al.*, 1992), again questioning the physiological relevance of H1RLX. Human chorionic gonadotropin (hCG) has been shown to influence relaxin secretion in females (Quagliarello *et al.*, 1980), as injection of hCG into a non-pregnant female produces detectable levels of relaxin in the serum a couple days later. In males, hCG has no effect on relaxin secretion as ejaculate from male subjects with no hCG contains detectable amounts of relaxin (Colon *et al.*, 1994).

Early physiological studies showed that PRLX improved motility of washed human sperm (Lessing *et al.*, 1986). and improved the ability of the sperm to penetrate bovine cervical mucus (Brenner *et al.*, 1984a). Studies with human relaxin have shown that treatment of washed human sperm with Ing/ml of human relaxin increased penetration significantly better than when treated with 100ng/ml of PRLX

19

日本をようにはたいでいたがないのでので、「ない」のないで、

(Colon *et al.*, 1989). However, treatment of washed human sperm with relaxin did not have a greater or lesser effect on penetration when compared to human sperm in semen, indicating that fresh, normal semen has an optimum amount of relaxin, and that the addition of more relaxin has no further effect. Another possible role for relaxin in the male may be connected with the development of prostatic hyperplasia, as the expression of relaxin gene transcripts and the presence of relaxin binding sites increases with age in the rat prostate (Hornsby *et al.*, 2001).

# 1.4.3 Cardiovascular system

An increase in the heart rate of pregation women occurs within the same time frame as changes in the plasma relaxin concentrations (Stewart et al., 1990). It was therefore hypothesized that relaxin may have a direct effect on the heart. It was also observed that anaesthetised female rats injected with PRLX, intravenously into a cannulated femoral vein, responded with a subsequent increase in blood pressure and heart rate (Parry et al., 1990). Relaxin binding sites were identified in the atrium of both male and female rats (Osheroff et al., 1992), and further characterisation demonstrated no difference in affinity for relaxin but a greater density of receptors in the female than the male (Tan et al., 1999). Oestrogen treatment did not alter relaxin binding in atria from female rats as in the uterus, but did reduce the density of relaxin binding sites in atria from male rats. The rat isolated atrial bioassay (Section 1.3.2.2) provided evidence of a direct effect of relaxin to cause positive inotropic and chronotropic responses, with relaxin more potent than angiotensin II, isoprenaline and adrepaline (Kakouris et al., 1992). Gene transcripts of both H1RLX and H2RLX were identified in human atrial and ventricle tissue (Dschietzig et al., 2001) however local synthesis of the relaxin peptides has yet to be comlimed (Samuel et al., 2003a).

The influence of relexin on the cardiovascular system extends beyond direct stimulation of the heart. Relaxin may regulate vascular tone as administration of relaxin increased coronary flow in guinea pig isolated hearts (Bani-Sacchi *et al.*, 1995), caused a delayed increase in renal plasma flow comparable to that seen in pregnancy (Danielson *et al.*, 1999) and induced potent dilation of small systemic

resistance arteries (Fisher *et al.*, 2002). The vasodilator effects of relaxin have been suggested to be: a direct consequence of relaxin stimulation (Fisher *et al.*, 2002); mediated through nitric oxide production (Bani *et al.*, 1988; Danielson *et al.*, 1999), endothelin production (Danielson *et al.*, 2000) and atrial natriuretic peptide secretion (Toth *et al.*, 1996); endothelium dependent (Reid *et al.*, 2001); and vary between vascular beds (Samuel *et al.*, 2003a).

While the mechanism of action of relaxin in the cardiovascular system remains unclear, there is increasing evidence that relaxin may play a protective role in heart failure and in chronic heart disease. In guinea pigs with induced myocardial damage, relaxin significantly increased coronary flow, improved heart contractility and decreased the generation of markers associated with myocardial damage (Masini *et al.*, 1997). In patients with chronic heart failure, there was a significant increase in the synthesis of relaxin gene transcripts and in relaxin plasma levels, which correlated directly with the severity of the disease (Dschietzig *et al.*, 2001). In a rat model of myocardial infarction the inotropic response to relaxin was diminished while the chronotropic response was unaffected (Kompa *et al.*, 2002). Gene transcription of Rat3RLX was also shown to be significantly increased in this model of myocardial infarction, further suggesting a protective role for relaxins in the development of heart disease.

# 1.4.4 Brain

The first indication of a central role for relaxin came with the observation that central administration of relaxin inhibited the release of oxytocin, and affected milk ejection during suckling in the anaesthetized lactating rat (Summerlee *et al.*, 1984). Further investigation indicated that the subfornical organ, situated in the ventricular system, was important in mediating the inhibitory action of relaxin, since lesions to the subfornical organ prevented the suppression of oxytocin release (Summerlee *et al.*, 1987). Injection of relaxin directly into the lateral and dorsal portions of the third ventricle induced a sustained rise in blood pressure while injection into the ventral portion of the third ventricle caused a short rise in blood pressure (Mumford *et al.*, 1989). Lesions to the subfornical organ prevented the sustained pressor action

of relaxin but did not prevent the acute rise in blood pressure resulting from relaxin administration into the ventral portion, indicating a biphasic action of centrally administered relaxin. The angiotensin II system was thought to be involved in the central actions of relaxin (Parry *et al.*, 1990). Blockade of the central angiotensin system prevented the pressor response to icv relaxin, but only partially blocked the pressor response to an iv administration of relaxin (Parry *et al.*, 1991), with the remaining pressor response removed by a peripheral vasopressin antagonist. Blockade of the central angiotensin system reduced the relaxin-stimulated release of vasopressin and oxytocin (Geddes *et al.*, 1994) and water consumption of rats (Sinnayah *et al.*, 1999).

Binding studies have revealed high concentrations of relaxin binding sites in several circumventricular organs, including the subfornical organ and the organum vasculosum of the lamina terminalis (OVLT), areas known to be associated with control of blood pressure and fluid balance, and in the magnocellular regions of the paraventricular and supraoptic nuclei, regions responsible for the release of oxytocin and vasopressin (Osheroff et al., 1991). Relaxin binding sites have been identified in layer V of the cerebral cortex of rat brain (Osheroff et al., 1991; Tan et al., 1999) although the physiological function of these relaxin binding sites is still unclear. Fos immunoreactivity studies confirmed results from the binding studies since icv injection of relaxin caused fos-positive neurons to appear in the outer layer of the subfornical organ, dorsal part of the OVLT, throughout the median preoptic nucleus, supraoptic nucleus and the hypothalamic paraventricular nucleus (McKinley et al., 1997). Neurons in the outer layer of the subfornical organ and the dorsal part of the OVLT have efferent connections to the supraoptic nucleus and the hypothalamic paraventricular nucleus (McKinley et al., 1998), implying that it is possible that relaxin releases oxytocin and vasopressin from the supraoptic and paraventricular nuclei indirectly via the subfornical organ and the OVLT. The administration of anti-relaxin antibodies into the circumventricular organs of the rat brain caused a specific central effect to disrupt the timing of birth (Summerlee et al., 1998), whereas peripheral administration of anti-relaxin antibodies had no influence on delivery or survival of pups (Hwang et al., 1989).

Since relaxin has a specific central role and cannot cross the blood-brain barrier, it seems likely that relaxin is produced locally in the brain. Low levels of relaxin mRNA have been identified in the anterior olfactory nucleus, dentate gyrus and neocortex (Burazin et al., 2001; Osheroff et al., 1993), suggesting that the relaxin peptide is localised to few areas while the relaxin receptor is widely distributed This has led to the hypothesis that the relaxin peptide is within the brain. synthesised in the cell bodies and transported to nerve terminals with relaxin binding sites possibly indicating the location of peptide release (Burazin et al., 2001). With the identification of the third human relaxin gene, an equivalent rat relaxin gene 3 (Rat3RLX) was isolated (Burazin et al., 2002) and found to be abundantly expressed in the ventromedial dorsal tegmental nucleus (vmDTg) while low levels of rat relaxin gene 1 (Rat1RLX) were found in areas previously reported to express relaxin mRNA (Osheroff et al., 1993). Relaxin receptors are also expressed in the vmDTg (Burazin et al., 2002), an area shown to connect afferents and efferents with forebrain areas (Goto et al., 2001) leading to the hypothesis that the vmDTg was integrated into the brainstem network that regulates behaviour. This information, combined with the localisation of relaxin binding sites in the cerebral cortex (Tan et al., 1999), suggests that relaxin may influence behaviour (Burazin et al., 2002). The ratio of Rat1RLX to Rat3RLX expression in rat brain has led to the hypothesis that Rat3RLX may be the predominant centrally acting relaxin while Rat1RLX may be the predominant peripherally acting relaxin.

# 1.4.5 Connective tissue and cell growth

Relaxin remodels connective tissue to cause the elongation of the pubic symphysis and facilitate parturition, but little was known about the effects of relaxin on connective tissue outside the reproductive system. Relaxin administration to a cell culture of human dermal fibroblasts produced an increase in the secretion of metalloproteinase, procollagenase and a decrease in the expression of the tissue inhibitor of metalloproteinase expression (Unemori *et al.*, 1990). A decrease in expression of interstitial collagens was also noted, resulting in an overall increase in collagen matrix turnover and a reduction in overall tissue collagen content. Addition of transforming growth factor beta (TGF- $\beta$ ) and interleukin 1-beta (IL-1 $\beta$ ) combined

with indomethacin, which is known to increase collagen secretion in these fibroblast cells, failed to inhibit relaxin induced collagen turnover, indicating that relaxin could modulate collagen synthesis in presence of collagen over-expression (Unemori *et al.*, 1990).

Similar results were obtained in scleroderma fibroblast (Unemori *et al.*, 1992) and lung fibroblast cell cultures (Unemori *et al.*, 1996), as collagen over-expression was reduced by relaxin treatment, while relaxin did not markedly alter the basal collagen levels. Relaxin treatment of an *in vivo* murine model of bleomycin-induced lung fibrosis showed that relaxin prevented and reversed the effects of fibrosis (Unemori *et al.*, 1996), without any effect on non-fibrotic lung tissue. A similar result was obtained in a cell culture model of rat liver fibrosis, with relaxin treatment decreasing collagen synthesis and deposition by decreasing the secretion of two collagenase inhibitors that prevent collagen breakdown, TIMP-1 and TIMP-2 (Williams *et al.*, 2001). An *in vivo* model of renal interstitial fibrosis treated with relaxin showed renal protection with a subsequent decrease in renal disease symptoms, including serum creatinine and proteinuria (Garber *et al.*, 2001; McDonald *et al.*, 2003).

Based on its specific anti-fibrotic actions, relaxin has been trialled for the treatment of Scleroderma (Seibold *et al.*, 1998). Phase II clinical trials suggested that relaxin reduced scleroderma-related skin thickening, leading to an improvement of forced vital capacity and hand function (Seibold *et al.*, 2000). Although the results from Phase III clinical trials showed a similar trend, the relaxin treated patients were not significantly different from placebo (Erikson *et al.*, 2001).

# 1.4.6 Relaxin gene knockout mouse

Recently, inactivation of the relaxin gene by homologous recombination in embryonic stem cells was used to develop a relaxin knockout mouse (relaxin -/- mouse) (Zhao *et al.*, 1999). The phenotype of the relaxin -/- mice displayed no gross anatomical or behavioural differences compared to the normal wild type mouse. Both the male and female relaxin -/- mice were fertile and produced normal litters in

Ŀ

direct contrast to anti-relaxin antibody studies performed in rats (see Section 1.4.1), although male -/- mice displayed stunted growth of the testis, epididymis and prostate compared to age matched wildtype mice (Samuel *et al.*, 2003b). In agreement with the anti-relaxin antibody studies, the mammary gland and nipple development in relaxin -/- female mice was impaired, partly due to a lack of connective tissue development, which resulted in pups being unable to suckle and survive unless fostered to wild type mothers (Zhao *et al.*, 1999). Closer inspection of the pubic symphysis showed that while relaxin -/- mice had an elongated interpubic ligament compared to virgin mice, it was significantly smaller than wild type mice, reiterating the importance of relaxin in this function.

The lack of mammary gland and nipple development led to a further investigation of the influence of relaxin on collagen regulation (see Section 1.4.5). It was found that relaxin -/- mice failed to regulate collagen degradation in the nipple and vagina while the collagen fibres of cervix and mammary glands appeared more densely packed compared to wild-type mice (Zhao *et al.*, 2000). Kelaxin -/- mice showed an increase in age-related development of fibrosis in non-reproductive tissues, including the lung and kidneys, when compared to age-matched wild-type mice (Samuel *et al.*, 2003c; Samuel *et al.*, 2004). The influence of relaxin on fibrosis was further highlighted when the fibrosis in relaxin -/- mice was shown to be reversed to wild type mice levels following relaxin treatment. The relaxin -/- mouse has provided an important scientific tool for elucidating the role of relaxin in the regulation of collagen and in the development of fibrosis.

# Table 1.1

A brief summary of the rat and human physiological systems that express the relaxin receptor, the relaxin peptide and the function of relaxin in these systems. N/A denotes information not available

'n,

Physiological System	Species	Relaxin receptors expressed	Source of relaxin peptide	Functional relevance
Reproductive system	Rat - female	Myometrium, cervix, vagina, nipples, mammary glands, pubic symphysis ligament (Weiss et al., 1982; Oshcroff et al., 1990; Kuenzi et al., 1995; Yang et al. 1992)	Corpus luteum (Sherwood, 1994)	Maintenance of pregnancy, parturition, mammary gland and nipple development.
	Rat - male	Prostate (Homsby et al., 2001)	Prostate (Gunnersen et al., 1995; Hornsby et al., 2001)	Involved in the development of testis, epididymis and prostate.
	Human - female	Fallopian tube, uterus, cervix, mammary glands and nipples (Tang et al., 1995; Kohsaka et al., 1998)	Corpus luteum (Sakbun et al., 1990; Johnson et al., 1991)	Decidualisation, possibly involved in implantation.
	Human - male	LGR7 mRNA transcripts identified in prostate and testis (Hsu et al., 2002)	Prostate (Essig et al., 1982; Winslow et al., 1992)	Possibly involved in sperm motility.
Cardiovascular system	Rat	Atria (Osheroff et al., 1992; Tan et al., 1999)	Rat3RLX mRNA detected in in atria and ventricles (Samuet et al., 2003c)	Influences blood pressure and heart rate, protection against myocardial infarction.
	Human	No binding sites identified in heart, receptors identified through functional studies in systemic resistance arteries (Tan 1999; Fisher et al 2002)	H1RLX and H2RLX mRNA identified in the atrium and ventricle (Dschietzig <i>et al.</i> , 2001)	Cardiovascular regulation through dilator actions, possible protection against chronic heart failure.
Central nervous system	Rat	OVLT, hypothalamic paraventricular and supraoptic nuclei, cerebral cortex layer V, vmDTg (Osheroff et al., 1991; McKinley et al., 1997; Tan et al., 1999)	Anterior olfactory nucleus, dentate gyrus, neocortex, vmDTg (Osheroff et al, 1993; Burazin et al, 2002)	Stimulates oxytocin and vasopressin release, influences behaviour and reproductive system.
	Human	Deep layer of the parietal, entorhinal and frontal cortex (Tan 1999)	N/A	N/A

\*\*\*\*\*\*\*\*\*\*\*\*\*\*

Chapter 1: General Introduction

26

. . .

418 - 500 (\* 19. 25) (\* 19. 25)

.

# 1.5 Relaxin receptor binding assays

Relaxin radioligands were developed to examine the characteristics and localisation of relaxin receptors in various tissues.

# 1.5.1 Specific relaxin radioligands

# 1.5.1.1 Radioiodinated relaxin

[<sup>125</sup>I] labelled PRLX was originally developed to assess serum levels of relaxin using radioimmunoassay (Sherwood *et al.*, 1975b). Synthesis of [<sup>125</sup>I]-PRLX involved the incorporation of a tyrosine residue into the porcine relaxin sequence to allow radio-iodination to occur, with the resulting [<sup>125</sup>I]-PRLX peptide appearing to be biologically active in an immunological assay. Simultaneously another method was developed that utilised the iodination of an acylating agent, known as the Bolton & Hunter iodination method (Bolton *et al.*, 1973), which was then reacted with PRLX to produce a [<sup>125</sup>I]-PRLX. This method does not involve exposing the peptide to the harsh treatment inherent in the first method and does not rely on the presence of a tyrosine. [<sup>125</sup>I]-PRLX was then purified on reverse phase HPLC (Yang *et al.*, 1992), although the stability of the ligand was decreased (Osheroff *et al.*, 1990).

Although [<sup>125</sup>I]-PRLX has been used extensively to localise relaxin receptors, the properties of the ligand are altered slightly by the addition of the <sup>125</sup>I. The iodine is thought to cause conformational changes to the relaxin peptide and the labelled peptide has a lower binding affinity (Yang *et al.*, 1992) and reduced bioactivity (Cheah *et al.*, 1980). Iodination of H2RLX has advantages over PRLX as it naturally contains a tyrosine residue in the A chain (A3; Figure 1.2) and incorporation of the iodine does not affect the binding affinity (Palejwala *et al.*, 1998) although the bioactivity of this peptide has not been determined. One common problem of iodinated relaxin ligands is that the specific activity can vary considerably between preparations, ranging from  $80 - 185 \,\mu\text{Ci}/\mu\text{g}$  (Gates *et al.*,

1981; Palejwala et al., 1998; Sherwood et al., 1975b) as the unlabelled relaxin peptide can not be easily separated from the labelled relaxin peptide.

# 1.5.1.2 Phosphorylated relaxin

High specific activity labelled relaxin was produced using cAMP-dependent protein kinase to phosphorylate H2RLX at the serine (B32) with  $[\gamma^{-3^2}P]$ -ATP, resulting in [<sup>32</sup>P]-H2RLX (Osheroff *et al.*, 1990). Phosphorylation was specific to the serine (B32) and attempts to phosphorylate H2RLX with B chain lengths of 29 and 31 residues failed. [<sup>32</sup>P]-H2RLX has a specific activity of >5000 Ci/mmol (Osheroff *et al.*, 1990) and unlike [<sup>125</sup>I]-RLX, the unlabeled relaxin peptide can be separated from the labelled relaxin peptid<sup>12</sup> using cation exchange HPLC (Section 2.3.1), maintaining consistent specific activity between preparations. H2RLX but not other insulin-like factors competed with [<sup>32</sup>P]-H2RLX binding to relaxin receptors and biological activity was unaffected by phosphorylation. Although the half life of this ligand is shorter than that of iodinated relaxin, the stability and specificity of [<sup>32</sup>P]-H2RLX made it a favourable option.

To increase the resolution of autoradiography performed with  $[^{32}P]$ -H2RLX, a modification was made by substituting the  $[\gamma - {}^{32}P]$ -ATP with  $[\gamma - {}^{33}P]$ -ATP to produce  $[^{33}P]$ -H2RLX (Tan *et al.*, 1999). The sharpness of the autoradiography images were improved due to the lower energy of  $[^{33}P]$   $\beta$  particle emission (Hudson, 1993).

### 1.5.2 In vivo localisation of relaxin receptors

Early relaxin receptor localisation studies were performed *in vivo* by injecting whole animals with [ $^{125}$ I]-RLX and examining individual tissues for ligand localisation that can be displaced by administration of unlabeled hormone. Injection of [ $^{125}$ I] PRLX into female guinea pigs identified ligand localised to the uterine cervix, pubic symphysis and to a lesser degree to the uterine horns that could be blocked with administration of unlabeled PRLX (Gates *et al.*, 1981). There was no localisation of the ligand into other tissues, such as mammary gland, ovary, liver, spleen and connective tissue. Similar results were obtained in rats where localisation in tissues なるなくして大きななななの時代に構成された。ためなどのでのないので、「ないないない」ので、「ないたい」で、「ないない」で、「ないない」で、「ないない」、

other than the uterus was not displaced by unlabeled relaxin indicating non-specific uptake of the ligand (Cheah *et al.*, 1980). Injection of [ $^{125}$ I]-RLX into mice caused uptake of the ligand into the kidneys, with radioactivity also in heart and other major vessels but not in the uterus or pubic joint (O'Byrne *et al.*, 1982). This assay was acknowledged to be unreliable due to metabolism of the radioligand (Cheah *et al.*, 1980) and the difficulty of transporting a large ligand across the blood vessel walls into the target tissues (O'Byrne *et al.*, 1982).

# 1.5.3 In vitro localisation of relaxin receptors

Receptor autoradiography performed on isolated tissue sections allows fine-detail localisation of relaxin receptors within tissues and has proven to be a reliable assay that can be optimised to ensure specific binding of the radioligand. Exposure of tissue sections to ligands has identified relaxin binding sites throughout the reproductive system of rodents, including the myometrial and endometrial layers of the uterus (Osheroff et al., 1990; Tan et al., 1999; Weiss et al., 1982), in the cervix and vaginal tract (Kuenzi et al., 1995; Osheroff et al., 1990; Weiss et al., 1982), within the epithelial cells of the mammary gland as well as the epithelial and smooth muscle cells of nipples (Kuenzi et al., 1995). High resolution autoradiography demonstrated relaxin binding sites in the peripheral layers of the ovary and within the pubic symphysis ligament (Yang et al., 1992). Sites were also identified in the heart, localised to the atrium (Osheroff et al., 1992; Tan et al., 1999) and in specific areas in the brain, including the fifth layer of the cortex and the subfornical organ within the OVLT (Tan et al., 1999). These studies provide evidence that supports the wide range of physiological functions of relaxin (Section 1.4) that could not be confirmed by the in vivo receptor localisation studies.

Autoradiographic localisation of relaxin receptors in the human is limited to a few studies, possibly due to the difficulty in obtaining human tissues. Relaxin binding sites were identified using biotinylated PRLX throughout the reproductive organs, including the fallopian tube (Tang *et al.*, 1995), both the endometrial and myometrial layers of the uterus, cervix, vagina and mammary glands and nipples (Kohsaka *et al.*, 1998). However functional responses to relaxin in human tissues do not support all

日本日本にたちのためのためのないのでの

areas identified with relaxin binding, with for example minimal physiological function found in isolated myometrial tissue where PRLX and HRLX failed to inhibit the spontaneous contractions produced by isolated human myometrial strips (MacLennan *et al.*, 1991; MacLennan *et al.*, 1995; MacLennan *et al.*, 1986a; Petersen *et al.*, 1991). Autoradiographic studies of human heart sections failed to demonstrate specific relaxin binding sites, while receptors were identified in brain sections, specific to the parietal and entorhinal cortexs and within a deep cerebral cortical layer (Tan, 1999). A recent study has identified specific relaxin binding within the synovial lining cells, stromal fibroblasts and blood vessels lining in the female anterior cruciate ligament while no relaxin receptors were identified in the male ligament (Dragoo *et al.*, 2003), demonstrating a sex-specific localisation of relaxin receptors in a non-sex dependant tissue that has not been reported in animals.

# 1.5.4 Pharmacokinetic binding assays

Membrane preparations have been used to characterise relaxin receptors in a variety of tissue and cell types. Cell culture has allowed the identification of relaxin receptor in human foetal membranes (Garibay-Tupas *et al.*, 1995; Koay *et al.*, 1986) thus identifying a novel target for relaxin in the development of the foetus. Binding of [ $^{32}P$ ]-H2RLX to human isolated uterine cells provided the first indication of the size of the relaxin receptor, calculated to be ~220kDa (Osheroff *et al.*, 1995). Relaxin bound to a single site from which it could not be displaced by insulin, IGF's or INSL3, indicating that relaxin modulates cell function through a distinct relaxin receptor (Parsell *et al.*, 1996).

The combination of cellular binding assays and functional bioassays has provided valuable insights into the role of relaxin (Section 1.3). Human isolated endometrial stromal and glandular epithelial cells, and lower uterine fibroblast cells have been shown to express a functional relaxin receptor (Palejwala *et al.*, 1998; Palejwala *et al.*, 2002) that appears to stimulate different signalling pathways depending on the cell preparation. This indicates a variable role for relaxin within the reproductive system that can be further investigated using the cellular bioassays described in Section 1.3.2.3.

# 1.6 Identification of the relaxin receptor

Although relaxin was one of the first reproductive hormones to be identified, the isolation and characterisation of the relaxin receptor eluded researchers for decades. Since relaxin is related to insulin and other members of the insulin-gene superfamily which signal through tyrosine kinases, it was initially proposed that the relaxin receptor would also be a tyrosine kinase, related to but distinctly different from the insulin receptor. The relaxin receptor was reported to be comprised of two components, of 220 and 36 kDa, similar to the insulin receptor (Osheroff *et al.*, 1995). The same study reported that binding of  $[^{32}P]$ -relaxin to human uterine cells was unaffected by GTP, indicating that the relaxin-receptor was unlikely to be a G-protein coupled receptor.

However, and surprisingly given the indications noted above, the relaxin receptor was recently identified as an orphan leucine-rich repeat-containing G-protein coupled receptor (LGR). Two LGR orphan receptors, previously cloned as LGR7 & LGR8, were identified as receptors that could interact and be activated by relaxin (Hsu *et al.*, 2002). The original relaxin receptor appeared to be LGR7 as relaxin displayed similar affinity and potency to that previously described in functional and binding studies. The LGR8 receptor, while displaying a lower affinity and potency i + iNSL3, and appears to be the INSL3 receptor.

# 1.6.1 Leucine-rich repeat-containing G-protein coupled receptor family

LGR receptors are large G-protein coupled receptors containing the typical seven transmembrane domains with a unique large ectodomain that includes the leucinerich repeat (LRR) regions (Figure 1.4). The ectodomain is important for interaction with the large proteins that target these receptors. The LGR family consists of three subfamilies of receptors; Type A LGRs include the luteinizing hormone receptor (LHR), follicle-stimulating hormone receptor (FSHR) and the thyroid-stimulating hormone receptor (TSHR); Type B LGRs include the orphan LGR4, LGR5 & LGR6

receptors; Type C LGRs consisting of the relaxin receptor (LGR7) and the INSL3 receptor (LGR8) (Hsu, 2003). Each group of LGRs is classified according to their ectodomain structure. Type A LGRs ectodomains contain 9 LRR, Type B contain 17 LRR in their ectodomains, while Type C LGRs contain 9 LRRs with a unique N-terminal low-density lipoprotein (LDL) receptor-like cysteine-rich domain (Figure 1.5).

# 1.6.2 LGR7, the putative relaxin receptor

Comparison of recent research on the LGR7 receptor with previously reported data on the relaxin receptor suggests that they are one and the same receptor. LGR7 mRNA transcripts have been identified in a variety of tissues that are known to display relaxin binding or functional responses, including ovary, uterus, brain and heart (Hsu *et al.*, 2000). The LGR7 receptor transiently expressed in HEK293T cells produces cAMP when stimulated by PRLX (Hsu *et al.*, 2002) with similar potency to that reported in the THP-1 cell line which endogenously expresses the relaxin receptor (Eartsch *et al.*, 2001).

# 1.7 Aims of the present study

The aim of this study is to pharmacologically characterise the interaction of relaxin with the human LGR7 receptor, and thereby confirm that LGR7 is indeed the relaxin receptor. In order to achieve this aim, a high throughput binding assay was developed, utilising THP-1 cells which endogenously express the human relaxin receptor, CHO cells transiently transfected with the LGR7 receptor and HEK293 cells stably transfected with the LGR7 receptor. To further investigate the structural interaction between HRLX and the relaxin receptor, methods were developed to recombinantly express HRLX peptide and employ molecular cloning techniques to modify the HRLX sequence in order to deduce novel areas within the relaxin peptide that are important for its activity.

To investigate the characteristics of an endogenously expressed human relaxin receptor within a tissue known to display relaxin function, human uterine tissue was

「大学をたいたい」「大学になったいできたい」というないないなんでき

物理部の法律が必要がない。法律が必要が必要がない。

utilised. The expression of relaxin and the relaxin receptor was investigated to gain further insights into the role of relaxin in the human reproductive system.

# Figure 1.4

Schematic diagram of a leucine-rich repeat-containing G-protein coupled receptor, LGR7.

LGR7 has the typical seven transmembrane spanning regions positioned within the cell membrane. The large extracellular domain contains the 9 leucine-rich repeat regions and the unique N-terminal low-density lipoprotein receptor-like cysteine-rich (LDLR) domain.

Schematic diagram kindly provided by Daniel Scott, Howard Florey Institute of Experimental Physiology and Medicine, Vic, Aust.

÷.



# Figure 1.5

Schematic diagram of the three distinct ectodomains that characterise the LGR receptor family.

The three subfamilies of LGR receptors are characterised by the structure of the ectodomain; Subfamily A receptors contains \$ leucine-rich repeat regions (LRR, ), Subfamily B receptors contain 17 LRR, while Subfamily C receptors contain 9 LRRs with a unique N-termina! low-density lipoprotein (LDL, ) receptor-like cysteine-rich domain. Each subfamily contains a distinct hinge region () as denoted by the different colours.







.

**Subfamily B LGR** 



Subfamily C LGR

# **Chapter Two**

# **General Methods**

# 2.1 Celi culture

# 2.1.1 Maintenance of cell lines

All cell lines were obtained from the American Type Culture Collection (Manassas, VA, USA), except BHK cells which were obtained from Invitrogen (Mount Waverley, Vic, Aust). All cell culture was performed in the cell culture laboratory in PS2 laminar flow hoods under sterile conditions using sterilised equipment. Cells were maintained at  $37^{\circ}$ C, 5% CO<sub>2</sub>, 85% humidity in a CO<sub>2</sub> water-jacketed incubator (Forma Scientific, Marietta, OH, USA). The basic cell culture techniques were as described in Freshney (1994). The cell environment was optimised with the appropriate growth media and nutrients to ensure the continuation of viable cell lines (Freshney, 1994).

### 2.1.1.1 Non-adherent cell lines

To split cells grown in suspension and renew media, the growth media was transferred from the tissue culture flask to a sterile centrifuge tube, centrifuged for 5 minutes at 1000g to pellet cells and the supernatant removed. Cells were gently washed with Hepes-buffered saline (HBS; 10mM Hepes, 150mM NaCl, pH 7.4) and resuspended in new media. Cells were counted and the appropriate number of cells was transferred into new flasks with media.

### THP-1 cells

THP-1 (a human monocyte cell line) cells were maintained in RPMI 1640 medium containing glutamine (2 mM), penicillin (100 units/mL), streptomycin (100  $\mu$ g/mL) and heat inactivated foetal bovine serum (FBS; 10% v/v). Optimal growth and cell survival was ensured by maintaining the density of cells within 2x10<sup>5</sup> and 1x10<sup>6</sup> cells/mL.

# 2.1.1.2 Adherent cell lines

As the extent of cell growth varied with the adherent cell lines and composition of the medium, splitting of cells and media changes depended on the cell type. Media changes were conducted by removal of the old media from the tissue culture flask and replacement with new media. Once the cells had reached 80-90% confluency, they were split as follows: medium was removed from the culture flask, the cells gently washed 1-2 times with sterilised phosphate buffered saline (PBS; 140 mM 10 mM  $Na_2HPO_4$ , 1.8 NaCl. 2.7 mM KCl, mM KH<sub>2</sub>PO<sub>4</sub>) and trypsin/ethylenediaminetetraacetic acid (EDTA) (0.1% trypsin w/v, 0.02% EDTA w/v) solution added. Cells were incubated at 37°C for 5-10 minutes. Cells detached from the flask surface and clumps were disaggregated by pipetting to form a cell suspension. Cells were removed and placed in a new flask with new media, or centrifuged for counting and replating.

# BHK cells

BHK (baby hamster kidney) cells were maintained in alpha Modified Eagle's Medium ( $\alpha$ MEM) media containing glutamine (2 mM), penicillin (100 units/mL), streptomycin (100 µg/mL) and FBS (5% v/v).

# CHO-K1 cells

CHO-K1 (chinese hamster ovary) cells were maintained in 50:50 Dulbecco's Modified Eagle's Medium (DMEM): Ham's F-12 medium containing glutamine (2 mM), penicillin (100 units/mL), streptomycin (100  $\mu$ g/rnL) and heat inactivated FBS (10% v/v).

# HEK293T cells

HEK293T (human embryonic kidney) cells were maintained in RPMI 1640 media containing glutamine (2 mM), penicillin (100 units/mL), streptomycin (100  $\mu$ g/mL) and heat inactivated FBS (10% v/v). All tissue culture plates and flasks were coated with poly-L-lysine (0.1 mg/mL) prior to cell culturing to ensure HEK293T cell adhesion.

# 2.1.2 Freezing and thawing of cells

# 2.1.2.1 Freezing of cells

Cells were split according to the cell type, transferred to a sterile centrifuge tube and centrifuged for 5 minutes at 1000g to pellet cells. The supernatant was discarded and the cells resuspended in FBS. FBS containing 20% dimethyl sulfoxide (DMSO) was added to 1 mL resuspended cells/FBS to give a final concentration of DMSO of 10%. Cells were aliquotted (1 mL) into cryotubes, slowly frozen in a -70°C freezer for 24 hours and transferred to liquid nitrogen storage to ensure long term viability of cells.

# 2.1.2.2 Thawing of cells

Media was warmed to 37°C in a tissue culture flask for 30 minutes. Cells were taken from liquid nitrogen storage and immediately placed in water at 37°C. Thawed cells were added to the pre-warmed tissue culture flask and allowed to settle for 2 hours. Media was replaced with fresh media to remove the DMSO used in the freezing process.

# 2.1.3 Cell counting

A haemocytometer and cover slip were cleaned with ethanol and the cover slip placed over channels in the haemocytometer. Cells were lifted if necessary and approximately 100  $\mu$ L of resuspended cells added drop wise to the haemocytometer. Cells were counted under a microscope in the 16 squares and the average number of cells per square calculated. The average number of cells per square was multiplied by 2.5 x 10<sup>5</sup> to give the number of cells/mL.

# 2.1.4 Transient transfection of adherent cells

Approximately 18 hours before transfection, cells were split, counted and plated at  $8\times10^5$  cells/cm<sup>2</sup> in a tissue culture flask with normal media. The transfection solution was prepared under tissue culture conditions and contained the DNA construct (0.136 µg/cm<sup>2</sup>), the transfection reagent LIPOFECTAMINE<sup>TM</sup> (1.088 µL/cm<sup>2</sup>) and the transfection media OPTI-MEM® (22.6 µL/cm<sup>2</sup>). The transfection mix was incubated for 30-45 minutes at room temperature. Cells were washed 1-2 times with OPTI-MEM® and OPTI-MEM® (90 µL/cm<sup>2</sup>) was added to the transfection mix. Cells were then incubated with the transfection mix at 37°C for 4 hours. Serum enriched media (113.4 µL/cm<sup>2</sup>) was added to the cells according to cell type and incubated at 37°C for a further 20 hours. Transfection mix media was replaced with normal media and incubated at 37°C for 24 hours. Transfected cells were used 48 hours after transfection.

# 2.1.5 Production of a cell line stably expressing a foreign protein

Production of a cell line stably expressing the LGR7 receptor involved transiently transfecting cells in 2 wells of a 12-well plate with LGR7 DNA ligated into the pcDNA3.1-Zeo expression vector (Invitrogen, Vic, Aust) containing the zeocin resistance gene. Cells were maintained for 72 hours following transfection, then the contents of both v ells were transferred to a 175 cm<sup>2</sup> tissue culture flask with normal media containing zeocin (800  $\mu$ M) for specific selection. Cells not containing the LGR7 receptor DNA were selectively destroyed by zeocin, while cells containing the desired DNA continued to grow and replicate. Cells were maintained for 1-2 weeks. At 80% confluency, cells were split and resuspended in media containing zeocin at 8 cells/mL. Cells were plated out into 6 x 96-well plates (100  $\mu$ l/well) and incubated at 37°C for 2-3 weeks. Colonies produced from single cells were selected, screened for expression of relaxin receptor and maintained as individual cell lines.

# 2.2 Production of recombinant protein.

The Sindbis Expression System (Invitrogen, Vic Aust) was used to produce recombinant relaxin protein in a mammalian expression system. The pSinHis plasmid (Invitrogen, Vic Aust) and the relaxin peptide sequences were analysed for relevant restriction sites. A Pml I restriction site was designed for insertion into the multiple cloning site of the pSinHis plasmid and into the 5' end of the relaxin peptide segment. Oligonucleotides were designed (Table 2.1) to incorporate the Pml I restriction site and an appropriate sequence for the correct cloning of the relaxin peptide sequence into the pSinHis plasmid (Figure 2.1).

### 2.2.1 Incorporation of Pml I restriction site into the pSinHis plasmid

The pSinHis plasmid sequence was altered to include the Pml I restriction enzyme cleavage site (Figure 2.2) using the QuikChange<sup>TM</sup> Site-Directed Mutagenesis Kit (Stratagene, CA, USA) according to the manufacturer's instructions. Briefly, the mutagenesis reaction (20  $\mu$ L) contained the Pml I restriction site, forward and reverse primers (Table 2.1; 80 ng each), 1x reaction buffer, dNTPs (1 mM), pSinHis A plasmid DNA (100 ng) and Pfu enzyme (2.5 U). The reaction parameters included heating to 92°C for 2 minutes, 18 cycles of [95°C for 30 seconds, 55°C for 1 minute, 68°C for 25 minutes], followed by 25°C for 5 minutes and ending on 4°C. Although this was a set of single amino acid changes, 18 cycles was chosen due to the large size of the plasmid (10 kB). Following the mutagenesis reaction, the DNA was incubated for 2 hours at 37°C with Dpn1 to remove any non-methylated DNA, precipitated and resuspended in sterilised deionised water (10  $\mu$ L).

Table 2.1

Nucleotide sequences of primers used in the production of recombinant relaxin peptides.

Primer Name	Sequence $(5' \Rightarrow 3')$	
Primers for pSinHis H	Pml I site mutagensis	
pSinHis Pml for	CTG GTG GAC AGC ACG TGG GTC GGG ATC TG	
pSinHis Pml rev	CAG ATC CCG ACC CAC GTG CTG TCC ACC AG	
Primers for RLX inse	rt amplification	
RLX for	GGT GGA CAG CAC GTG GGT CGG GAT CTG TAC GAC GAT GAC	
	GAT AAG GAC TCA TGG ATG GAG GAA G	
RLX rev	CGG CCA OTG GCA TGC TAA GCT TCA GCA AAA TCT AGC	
Primers to sequence I	<sup>b</sup> CR-Script/RLX insert ligation	
T3	ΑΑΤ ΤΑΑ CCC TCA CTA AAG GG	
לז	TAA TAC GAC TCA CTA TAG GG	
Primers to sequence p	SinHis plasmid/RLX insert ligation	
pSinHis Seq for	CTA CAA CAC CAC CAC CTC TAG ACA CCA TGG G	
pSinHis Seq rev	CCT CGG AAG TAC ATC GAG TTT TGC TGG TCG G	

CAC CTA GAC CAG TGG CAG AAA TTG TGC CAT CC

pRLX Seq rev CTG AAG GAC TGC TGT CTG CGG CTT CAC

pRLX Seq for

# Figure 2.1

Schematic diagram of the insertion of relaxin peptide DNA into the pSinHis expression vector.

The relaxin insert generated for ligation into the pSinHis multiple cloning site incorporated the EK recognition and cleavage sequence to allow use of the Pml I restriction site and to maintain the EK enzyme restriction site for removal of the polyhistidine tag from the relaxin peptide following purification.



3'

# pSinHis cloning site 5' polyhistidine region Pml I EK recognition site EK cleavage site Sph I pSinHis **RLX** insert Pml i EK recognition site EK cleavage site RLX peptide Sph I 3' 5' pSinHis with RLX insert

43

# Figure 2.2

# Schematic diagram of the Pml I restriction site mutagenesis.

The insertion of the Pml I restriction site involved the mutation of two nucleotides within the pSinHis multiple cloning site sequence.


## 2.2.1.1 Transformation of pSinHis plasmid into DH<sub>5a</sub> cells

The altered pSinHis plasmid was transformed into competent DH<sub>5α</sub> cells (Sharma *et al.*, 1996). The pSinHis DNA (3  $\mu$ L) was combined with 50  $\mu$ L of competent DH<sub>5α</sub> cells, incubated on ice for 5 minutes and transferred to a BioRad Gene Pulser/*E. coli* Pulser cuvette (0.2 cm electrode gap). Electroporation was carried out at 2500 V, 25  $\mu$ F capacitance and 200 Ohms using a Gene Pulser II electroporation system (BioRad Laboratories, CA, USA). Immediately following transformation, 1 mL of pre-warmed SOC media (20 g/L casein hydrolysate, 5 g/L yeast extract, 0.5 g/L NaCl, 2.5 mM KCl, pH 7, 10 mM MgCl<sub>2</sub> and 20 mM glucose) was added to the cuvette. Cuvette contents were gently mixed, transferred to a microcentrifuge tube and incubated in an orbital shaker (37°C for 1 hour). The transformed cells were plated onto agar LB plates (LB broth: 1% tryptone, 0.5% yeast, 1% NaCl, 75 mg/L ampicillin; agar LB plates: LB broth, 1% agar) and incubated at 37°C for 16 hours.

## 2.2.1.2 Colony selection

Single colonies of cells grown on the agar LB plates were selected, picked with sterile toothpicks under sterile conditions and set in miniprep cultures according to the Wizard Plus Miniprep DNA Purification System (Promega, WI, USA; including cell suspension, lysis, neutralization, wash and elution buffers). Briefly, selected colonies were placed in 5 mL LB broth and incubated for 16 hours at 37°C in an orbital shaker. Cells were pelleted (2 minutes, 10,000xg), the supernatant discarded and cells resuspended in cell suspension solution (250  $\mu$ L). Lysis buffer (250  $\mu$ L) was added and the mixture gently inverted. Cell lysis was halted by addition of chilled neutralisation solution (350  $\mu$ L), gently mixed by inversion and the cell debris pelleted (10 minutes, 10,000xg at 4°C). The supernatant containing the DNA was transferred to a miniprep filter tube contained in a collection tube and centrifuged (1 minute, 14,000xg) to harvest the DNA onto the filter. The filtrate was discarded, the filter tube reconnected to the collection tube, and wash buffer (700  $\mu$ L) was added and centrifuged (1 minute, 14,000xg). The filtrate was discarded and

the filter tube centrifuged (1 minute, 14,000xg) to ensure a dry and clean filter. Any additional filtrate was discarded and the filter tube was reconnected to a sterilized Eppendorf tube. DNA was eluted from the filter with additional elution buffer (100  $\mu$ L) and centrifugation (30 seconds, 14,000xg). DNA was stored at -20°C. For long term storage, glycerol stocks of the miniprep cell culture were made (cells in 50% glycerol) and stored at -70°C.

The selected colonies of pSinHis plasmid DNA were checked for insertion of the Pml I restriction site. The purified plasmid concentrations were calculated as described in Section 2.5. Due to the large size of the plasmid (10 kB), 1  $\mu$ g of purified plasmid DNA was digested in the reaction mixture of 1x One-Phor-All buffer, 0.01% BSA, 20 U Xho I enzyme and 40 U Pml I enzyme in a total volume of 100  $\mu$ L (37°C for 2 hours). A sample of the digestion (10  $\mu$ L + 2  $\mu$ L loading dye) was run on a 1.3% agarose gel and appropriate band sizes were visualized for pSinHis containing the Pml I restriction site (450 bp dropout).

## 2.2.1.3 Amplification of altered pSinHis plasmid DNA

The DNA samples containing the correct digestion profile were further amplified with the Wizard Plus Midipreps DNA Purification System (Promega, WI, USA; including cell suspension, lysis, neutralization and wash buffers, purification resin and midicolumns). Briefly, 100 µL of glycerol stocks of the selected DNA samples were placed in 100 mL LB broth and incubated for 16 hours at 37°C in an orbital shaker. Cells were pelleted (10 minutes, 4000xg at 4°C), supernatant discarded and cells resuspended in cell suspension solution (3 mL). Lysis buffer (3 mL) was added, gently mixed by inversion of the tube and cell lysis terminated by addition of Neutralization buffer (3 mL). Cell debris was pelleted (15 minutes, 14,000xg at 4°C) and the supernatant containing the DNA transferred to a separate tube. Purification resin (10 mL) was added and swirled to mix. A Midiprep purification column was attached to a vacuum manifold. The resin and DNA solution was transferred and passed through the column allowing the plasmid DNA to adhere to the filter/resin. After washing with buffer (30 mL) the column was cut from column, the vacuum manifold. Filter/resin containing plasmid DNA was cut from column,

placed into a sterile Eppendorf tube and centrifuged (2 minutes, 10,000xg) to remove excess wash buffer. The resin filter was transferred to a new tube, preheated sterilised deionised water (70°C, 300  $\mu$ L) added and the DNA removed from the exhausted resin by centrifugation (20 seconds, 10,000xg). The purified plasmid concentrations were calculated (Section 2.5) and DNA was stored at -20°C.

## 2.2.2 Production of the relaxin DNA fragment

The relaxin peptide DNA sequence was obtained from Dr Ross Bathgate (Howard Florey Institute of Experimental Physiology and Medicine, Vic, Aust) with the relaxin peptide DNA in the pMAL plasmid. Primers were designed to incorporate a Pml I restriction site, the pre-relaxin sequence of the pSinHis plasmid and the 4' end of the relaxin sequence (Table 2.1) to amplify by PCR a relaxin DNA segment for ligation. The relaxin DNA segment amplification PCR reaction (20  $\mu$ L) contained forward and reverse primers (50 ng each), pMAL vector with relaxin DNA (25 ng), 1x reaction buffer, MgSO<sub>4</sub> (1.5 mM), dNTPs (1 mM). with Pfx polymerase (2.5 Units). The PCR conditions included 35 cycles with an annealing temperature of 62°C to form a blunt end DNA fragment.

The PCR product was run on a 1.3% low melting temperature agarose gel to visualize the DNA fragment. The DNA fragment size of ~500 bp was the expected size of the relaxin DNA fragment.

## 2.2.2.1 Purification of relaxin PCR DNA fragment

The relaxin DNA band was carefully cut from gel and purified using the Wizard PCR Preps DNA Purification System (Promega, WI, USA; including purification resin, buffers and columns). Briefly, the agarose slice of DNA was placed in a sterile Eppendorf tube and melted at 70°C. Resin (1 mL) was added to the melted agarose and gently mixed. A minicolumn was attached to a 3 mL syringe barrel, resin mix was added and pushed through the syringe with DNA adhering to the filter/resin in the column. Isopropanol (80%, 2 mL) was washed though the column, the minicolumn detached, placed in a sterile Eppendorf tube and centrifuged (2

minutes, 10,000xg) to remove excess isopropanol. The Minicolumn was transferred to a clean tube and DNA eluted with 50  $\mu$ l of nuclease-free water (20 seconds, 10,000xg). The DNA concentration was calculated (Section 2.5) and stored at - 20°C.

## 2.2.2.2 Ligation of relaxin DNA fragments into PCR-script.

To ensure correct digestion of relaxin DNA fragment to allow for correct ligation into pSinHis, the relaxin PCR fragment was ligated into PCR-script using a PCR-Script<sup>TM</sup> Amp Electroporation-Competent Cell Cloning Kit (Stratagene, CA, USA; including cloning vectors, reaction buffers, enzymes and selection tools). Briefly, the ligation reaction included 10 ng pPCR-script Amp SK(+) cloning vector, 1x reaction buffer, 0.5 mM rATP, 170 ng relaxin PCR fragment, 5 U Srf I restriction enzyme and 4 U T4 DNA ligase in a 10  $\mu$ L reaction (2 hours at room temperature). The reaction was inactivated by incubation at 65°C for 10 minutes, precipitated and resuspended in 10  $\mu$ L of deionised water.

The ligated DNA was transformed into XLI-Blue MRF competent cells as previously described with  $DH_{5\alpha}$  competent cells (Section 2.2.1.1). X-gal (20) mg/mL) was spread over the agar LB plates 30 minutes prior to plating out transformed cells. The XLI-Blue MRF competent cells on the X-gal will show cells containing the ligated relaxin PCR fragment as white, while cells containing the vector without the relaxin PCR fragment will appear blue in colour, allowing for an accurate selection of colonies containing the ligated DNA. The transformed cells were plated onto the agar LB plates and were placed in an orbital shaker (37°C for 16 hours). Single white colonies of cells grown on the agar LB plates were selected, picked in a sterile environment and set in miniprep cultures according to the Wizard Plus Miniprep DNA Purification System (Promega, WI, USA) as previously described (Section 2.2.1.2). The DNA was purified from the miniprep cultures, the concentration calculated and screened by digestion to ensure that the correct relaxin PCR fragment was present. The digestion reaction mix included 1x NE 1 buffer (New England Biolabs; 10 mM Bis Tris Propane-HCl, 10 mM MgCl, 1 mM DTT, pH 7), 0.01% BSA, 5 U Sph I enzyme, 20 U Pml I enzyme and 1 µg purified DNA

in a total volume of 20  $\mu$ L and was incubated at 37°C for 2 hours. A sample of the digest (10  $\mu$ L + 2  $\mu$ L loading dye) was run on a 1.3% agarose gel and an appropriate band size for the relaxin DNA fragment (~500 bp) was seen in various samples.

## 2.2.3 Ligation of relaxin DNA fragment into the pSinHis plasmid

The altered pSinHis plasmid (5 µg) and PCR-script vector containing the relaxin fragment (2 µg) were digested in a reaction mixture including 1x NE 1 buffer (New England Biolabs), 0.01% BSA, 25 U Sph I enzyme and 100 U Pml I enzyme in a total volume of 100 µL at 37°C for 2 hours. The digested DNA was loaded onto a 1.3% low melting temperature agarose gel where the relaxin DNA fragment bands (~500 bp) and linearised pSinHis DNA bands (~10 kB) were carefully cut from the The DNA was purified using a Freeze and Squeeze phenol chloroform gel. extraction method. The pieces of agarose gel were weighed, cut into small pieces and placed in sterile Eppendorf tubes with an equal volume of phenol (0.1g agarose to 100 µL phenol). The DNA was frozen at -70°C for 10 minutes, centrifuged for 15 minutes (14,000xg at 4°C) and the aqueous phase carefully transferred to a clean tube and the volume determined. Phenol (50% of aqueous phase volume) was added to the aqueous phase, mixed and centrifuged for 5 minutes (14,000xg at 4°C). An equal volume of chloroform was then added (50% of original aqueous phase volume), DNA solution was mixed and centrifuged for 10 minutes (14,000xg at 4°C). The upper phase containing the DNA was transferred to a clean tube. DNA was precipitated by the addition of 1/10 volume of sodium acetate (NaOAc; 3 M, pH5.2) and 2.5 x volume of absolute ethanol, mixed and incubated at -20°C for 20 minutes. DNA was centrifuged for 25 minutes (14,000xg at 4°C), supernatant discarded and the DNA pellet was carefully washed twice with 76% ethanol. The DNA pellet was air dried, resuspended in sterilized deionised water (20 µL), and the concentration calculated (Section 2.5) before storage at -20°C.

The ligation of the relaxin fragment into the pSinHis plasmid was carried out in a reaction mixture of 200 ng pSinHis plasmid DNA, 40 ng relaxin fragment DNA, 1x One-Phor-All buffer, 1 mM ATP and 4 U T4 DNA ligase in a total volume of 20  $\mu$ L

which was incubated overnight at room temperature. Ligation mix was precipitated and resuspended in sterilized deionised water (10  $\mu$ L). Ligation DNA was transformed into DH<sub>5α</sub> competent cells as previously described (Section 2.2.1.1), plated onto agar LB plates, colonies grown overnight, single colonies selected, set in miniprep cultures and DNA purified according to the Wizard Plus Miniprep DNA Purification System (Promega, WI, USA) as previously described (Section 2.2.1.2). The purified ligated DNA was digested to screen the colonies for correct ligation of relaxin fragment into pSinHis. The digestion reaction included 1x Buffer H (Amersham Pharmacia Biotech, England, UK), 0.01% BSA, 0.01% Triton X-100 (Amersham Pharmacia Biotech, England, UK), 5 U Not I enzyme and 0.5  $\mu$ g purified DNA in a total volume of 20  $\mu$ L incubated at 37°C for 2 hours. Digestion with Not I enzyme will only produce a dropout band of ~850 bp if the relaxin fragment is ligated into pSinHis plasmid in the correct orientation. Successful ligation was confirmed in several of the selected colonies when the digested DNA was visualised on a 1.3% agarose gel.

## 2.2.3.1 Sequencing of pSinHis plasmid containing the relaxin construct

Before sequencing, DNA was further purified by polyethylene glycol (PEG) precipitation. Briefly, 0.16 x volume of NaCl (5 M) and 1x total volume of 13% PEG was added to the DNA. The mixture was incubated on ice for 20 minutes, centrifuged for 25 minutes (14,000xg at 4°C), the supernatant discarded and the DNA pellet gently washed twice with 70% ethanol. The pellet was air dried, dissolved in 2/3 x original volume of sterilized deionised water and the concentration of DNA calculated (Section 2.5).

The sequencing amplification reaction mix (20  $\mu$ L) contained PEG purified pSinHis plasmid DNA containing the relaxin insert (800 ng), sequencing primer (4 pmol), BigDye Terminator Mix (v3.1, 4  $\mu$ L; Applied Biosystems, CA, USA) and DMSO (5%). The PCR sequencing parameters were heating at 96°C for 1 minute, 25 cycles of [96°C for 10 sec, 50°C for 5 sec, 60°C for 4 minutes], ending on 4°C. At completion of the PCR reaction, DNA was precipitated with 95% ethanol and NaOAc (0.3 M, pH 5.2), washed with 70% ethanol and air dried at room temperature

before being processed by electrophoresis through a denaturing polyacrylamide gel (Micromon Sequencing Facility, Monash University, Vic, Aust; Sambrook *et al.*, 2001).

## 2.2.4 RNA transcription of DNA templates

The pSinHis plasmid, with and without the relaxin construct (~20  $\mu$ g), was digested with Xho I (2 hours, 37°C) to linearise the DNA. The digestion was terminated with 1/20 volume of 0.5M EDTA and the DNA extracted by phenol/chloroform extraction. The DNA was precipitated with 1/15 x volume 7.5 M ammonium acetate (RNase free, Sigma) and 2x volume 100% ethanol (-20°C for 20 minutes, centrifugation for 25 minutes at 4°C), washed twice with 70% ethanol (400  $\mu$ L, centrifugation for 5 minutes at 4°C) to ensure a clean DNA template for RNA transcription. The pellet was dried at room temperature and resuspended in 15  $\mu$ L RNase-free Tris-EDTA buffer (TE buffer; Promega). DNA concentration was measured (Section 2.5) and adjusted accordingly to obtain a final DNA concentration of 0.6  $\mu$ g/ $\mu$ L.

The RNA transcription method was based on a combination of the 'SFV gene expression system' manual and products (Invitrogen, page 23) and 'Sindbis expression system' manual (Invitrogen, page 17-18). The 50  $\mu$ L reaction was set up in a 1.5 mL tube (autoclaved) containing NTP mix (20  $\mu$ L), 5x SP6 RNA polymerase buffer (10  $\mu$ L), m7G5'ppp5'G RNA cap analogue structure (5  $\mu$ L), 10 mM DTT (5  $\mu$ L), RNasin (40 U/ $\mu$ L; 1.3  $\mu$ L), RNase free TE (3.7  $\mu$ L), linearised plasmid (1.5  $\mu$ g; 2.5  $\mu$ L) and SP6 RNA polymerase (38 U; 2.5  $\mu$ L). The reaction was incubated for 2 hours at 37°C.

A sample of RNA was examined on a 1% agarose gel to check the quality of the RNA. RNA was precipitated with 1/15x volume 7.5 M ammonium acetate (RNase free, Sigma) and 2x volume 100% ethanol (-20°C for 20 minutes, centrifugation for 25 minutes at 4°C), washed twice with 70% ethanol (400 µL, centrifugation for 5

minute at 4°C) and the pellet resuspended in 20  $\mu$ L RNase-free (DEPC treated) dH<sub>2</sub>O. RNA concentration was measured (Section 2.5) and stored at -70°C.

## 2.2.5 Transfection of pSinHis RNA transcripts into BHK cells

Approximately 18 hours before transfection, BHK cells were split, counted and plated at  $8\times10^5$  cells/cm<sup>2</sup> in a tissue culture flask with  $\alpha$ MEM media. The transfection solution included the RNA transcript of pSinHis plasmid containing the relaxin construct (0.17 µg/cm<sup>2</sup>), the transfection reagent DMRIE<sup>TM</sup> (1.37 µL/cm<sup>2</sup>) and the transfection media OPTI-MEM® (120 µL/cm<sup>2</sup>). Cells were washed 1-2 times with OPTI-MEM® and incubated with the transfection mix at 37°C for 4 hours. Following the 4 hours, transfection solution was removed from cells and replaced with serum enriched  $\alpha$ MEM media (120 µL/cm<sup>2</sup>) incubated at 37°C for a further 20 hours. Some transfected BHK cells were harvested after 24 hours following transfection, while the remaining BHK cells were given a media change and were harvested 48 hours after transfection.

## 2.2.6 Protein extraction

The ProBond<sup>TM</sup> Purification System (Invitrogen, Vic, Aust) was used to extract and purify the polyhistidine-RLX protein. The polyhistidine tail displays high affinity for the pre-equilibrated nickel IDA bead column (Scientifix, Vic, Aust), binding the relaxin peptide and removing all excess lysate. The relaxin peptide was cleaved from its His<sub>6</sub> tag and purified from the nickei column (Figure 2.3).

## Figure 2.3

## Purification of polyhistidine tagged-RLX protein.

The expressed pro-relaxin peptide was purified from the BHK cell lysate through a nickel IDA column. The polyhistidine tag attached to the pro-relaxin peptide displayed high affinity for the activated column, binding to the beads and allowing the cell debris to be washed through. The pro-relaxin peptide was then cleaved from the polyhistidine tag with the EKMax<sup>TM</sup> enzyme and eluted from the column.

は代表が見たりですと言語





## 2.2.6.1 Prepare column for purification

The nickel beads in the column were gently resuspended in the storage mixture using a Pasteur pipette and storage solution eluted without allowing the column to dry out. The resin was washed twice with 7 mL sterilized deionised water, equilibrated with 7 mL binding buffer and excess buffer eluted. The equilibration process was repeated three times before protein extraction.

#### 2.2.6.2 Protein purification

Transfected BHK cells were washed with PBS, lifted and transferred into a 15 mL centrifuge tube. Cells were centrifuged, supernatant removed and the cells resuspended in 4 mL binding buffer (20 mM NaPO<sub>4</sub>, 500 mM NaCl; pH 7.6). The cells were lysed by freezing in a dry ice/ethanol bath and thawing in a 42°C water bath. The freeze/thaw method was repeated twice. The cell lysis suspension was passed through an 18-gauge needle four times to shear unwanted DNA. The cell debris was pelleted by centrifugation and the supernatant transferred to a new tube (containing the protein) and kept on ice. The pellet was kept for analysis.

The cell lysate preparation was added to the pre-equilibrated resin, column was sealed and incubated with gentle rocking at room temp for 20 minutes. The supernatant was eluted and kept for analysis. The resin was washed twice with 2x bed volume of binding buffer (pH 7.8), twice with 2x bed volume of wash buffer (20mM NaPO<sub>4</sub>, 500 mM NaCl; pH 6.0) and twice with 2x bed volumes of binding buffer (without NaCl; pH 7.8). After each wash the supernatant was eluted and kept for analysis.

After the final wash, the resin was resuspended in 500  $\mu$ L binding buffer (without NaCl; pH 7.8) and 50  $\mu$ L 10x EKMax buffer (Invitrogen). EKMax enzyme (10 U) was added to the suspension, column was sealed and rocked overnight at room temperature. The supernatant containing the protein was eluted and collected. The

resin was resuspended three times in 2x resin volume of wash buffer and the supernatant containing the fusion protein (His<sub>6</sub> tag) was eluted into a new tube.

The resin was stripped with 3 x column volume of buffer containing 40 mM EDTA (pH 9.0). The resin was recharged by washing with 3-5 volumes dH<sub>2</sub>O, followed by 50 mM NiSO<sub>4</sub> (1 mL NiSO<sub>4</sub> per 1 mL resin). The resin was stored in 50 mM phosphate buffer containing 20% ethanol at 4°C.

## 2.2.7 Analysis of recombinant protein

Separated protein samples were visualised on a 16% polyacrylamide gel (Section 2.7). Protein samples were mixed with an equal volume of 2 x SDS buffer (without dithiothreitol) and boiled for 5 minutes. Samples were loaded (15  $\mu$ L) together with a Prestained Precision Protein Standard marker (PPS, 10 $\mu$ L; BioRad, CA, USA) and H2RLX peptide as a positive control. For the Western blotting and immunostaining protocol, please refer to Section 2.7. Relaxin was visualised on the membrane using a relaxin specific polyclonal primary antibody (1:2000 dilution) and an anti-rabbit secondary antibody (1:1000 dilution).

## 2.3 Radioligand binding studies

## 2.3.1 Preparation of [<sup>33</sup>P]-H2RLX

## 2.3.1.1 Labelling of recombinant H2RLX with $[\gamma^{-33}P]$ -ATP.

H2RLX was labelled with  $[\gamma^{-33}P]$ -adenosine triphosphate (ATP) using the procedure of Osheroff *et al* (1990) with modifications (Figure 2.4). The reaction was carried out in siliconised tubes which contained 4 µg recombinant H2RLX (2 mg/mL stock), 5 µL of reaction buffer (20 mM Tris-HCl, 100 mM NaCl, 20 mM MgCl<sub>2</sub>, pH7.5), 5 µCi  $[\gamma^{-33}P]$ -ATP (specific activity 2000 Ci/mmol) and 400 U of cAMP-dependent protein kinase (cAMP-dPK) isolated from bovine heart. The reaction was carried out in a 37°C water bath for 1 hour and terminated by placing on ice.

## 2.3.1.2 Separation of unreacted $[\gamma^{-33}P]$ -ATP from labelled relaxin.

Separation of unreacted  $[\gamma^{-33}P]$ -ATP from labelled and unlabelled relaxin was accomplished using a Sep-Pak C<sub>18</sub> cartridge (Waters) (Figure 2.4). A 6 mL syringe barrel was attached to the cartridge. The cartridge resin was activated with acetonitrile, making sure that no air bubbles were introduced which hinder the flow through the cartridge. The resin was washed with deionised water, followed by BSA solution (1 mg/mL BSA in 0.1% TFA), and finally allowed to equilibrate with ATP (1 mM ATP in 0.1% TFA solution). The reaction mixture was added to the syringe with a total of 3 mL ATP (1 mM solution), followed by 4 mL 10% acetonitrile in 0.1% TFA and fractions collected (1 mL) in glass test tubes. The labelled and unlabelled relaxin was eluted using 3 mL 80% acetonitrile in 0.1% TFA solution. A sample of each fraction (10  $\mu$ L) was analysed by liquid scintillation spectrometry (TRI-CARB model 1900 TR, Perkin-Elmer, Vic, Aust).

## Figure 2.4

## Schematic diagram of the $[^{33}P]$ labelling of H2RLX

H2RLX was labelled with  $[^{33}P]$ -ATP in a reaction catalysed by cAMP dependent protein kinase (cAMP-dPK) resulting in 'abelled H2RLX, unlabelled H2RLX and free radiolabel  $[^{33}P]$ . The free  $[^{33}P]$  was removed via purification through a Sep-Pak C<sub>18</sub> cartridge. A pure sample of  $[^{33}P]$ -H2RLX was obtained by separating the unlabelled H2RLX from the labelled H2RLX through a PolyCAT A column on a cation exchange-HPLC.



3 Y 1

## 2.3.1.3 Separation of labelled and unlabelled relaxin

The labelled relaxin was separated from the unlabelled relaxin using cation exchange HPLC (Waters) incorporating a PolyCAT A column (Poly LC, USA). The PolyCAT A column is negatively charged, allowing the adhesion of positively charged peptides. Peptides are eluted from the column according to the strength of their positive charge by an increasing salt gradient applying positively charged counterions. The relaxin peptide containing the  $[\gamma^{-33}P]$  has an extra negatively charged phosphorous ion, forcing it to be eluted earlier than the unlabelled relaxin peptide (Figure 2.4).

The PolyCAT A column was equilibrated with Buffer A (25 mM KH<sub>2</sub>FO<sub>4</sub>, 25% acetonitrile, pH 7.4). The radioactive sample from the first elution step (Section 2.3.1.2) was injected onto the column. The column was eluted with a linear gradient of 0 - 0.5 M KCl in Buffer A at 1 mL/minute for a 50 minute period through a spectrophotometer measuring the absorbance at 214 nm. Elution fractions were collected at one-minute intervals in glass test tubes until the excess unlabelled relaxin peptide was detected by its large absorbance peak. A sample of each fraction (10 µL) was analysed by liquid scintillation spectrometry, and the fraction containing the peak radioactivity corresponding to [<sup>33</sup>P]-H2RLX was transferred into a siliconised tube containing 1 mg of BSA, stored at 4°C and used within seven days.

## 2.3.2 Quantitative Autoradiography

#### 2.3.2.1 Tissue preparation

#### Animal tissue preparation

Male Sprague-Dawley rats (200-250g) were anaesthetized with 80% CO<sub>2</sub>/20% O<sub>2</sub> gas mixture and decapitated. The heart was removed and washed in chilled 1:1 0.32M sucrose: Krebs phosphate (119 mM NaCl, 4.8 mM KCl, 1.2 mM MgSO<sub>4</sub>.7H<sub>2</sub>O, 10 mM NaH<sub>2</sub>PO<sub>4</sub>.2H<sub>2</sub>O) solution. The atria were dissected from the ventricles and frozen in isopentane chilled by liquid nitrogen. The brain was dissected, the brain stem and occipital lobe removed and the brain frozen on a cork platform floating in liquid nitrogen. Tissues were stored at  $-70^{\circ}$ C for up to four weeks. Frozen rat atria and brain were mounted in OCT in the cryostat, and 10 µm sections cut at  $-18^{\circ}$ C, and mounted on precleaned microscope slides subbed with poly-1-lysine (0.01% solution). Slide mounted sections were stored at  $-20^{\circ}$ C.

#### Human uterine tissue preparation

Full thickness endometrial tissue was collected by qualified nurses at 27 hysterectomy operations performed on ovulating women suffering from menorrhagia (age range 34-46). Subjects had not received exogenous hormones in the three months prior to hysterectomy, and endometrial tissue appeared normal by routine histological examination. Ethical approval for this study was obtained from the Southern Health Human Research and Ethics Committee B and informed consent was received from each subject. Tissue blocks containing the intact endometrial layer and several mm of myometrium were frozen in liquid nitrogen and stored at – 80°C. Frozen blocks of uterine tissue were mounted in OCT in the cryostat, cut into 10  $\mu$ m sections at -18°C and collected on precleaned microscope slide. subbed with poly-l-lysine (0.01% solution) identified only by their catalogue number. Slides were stored at -20°C. Following the completion of the binding assay, the uterine sections were analysed on ImageQuaNT<sup>TM</sup> (Version 4.1; Molecular Dynamics, U.S.A.) and the data grouped according to the menstrual phase in GraphPad PRISM (GraphPad Inc., San Diego, U.S.A.).

## 2.3.2.2 [<sup>33</sup>P]-H2RLX quantitative autoradiography binding protocol

Slides were removed from storage at -20°C and equilibrated to room temperature in sealed containers with silica gel to prevent development of condensation. A moist chamber was pre-heated to 25°C, and slides were pre-incubated for 30 minutes with the section covered with 100  $\mu$ L of binding buffer (25 mM Hepes, 300 mM KCl) containing 1 mM phenylmethylsulphonyfluoride (PMSF). Excess buffer was removed and the slides were incubated for 90 minutes in 100  $\mu$ L of altered binding buffer (25 mM Hepes, 300 mM KCl, 1 mg/mL BSA) containing [<sup>33</sup>P]-H2RLX ± cold relaxin and relaxin-like analogues. Slides were briefly rinsed in deionised water, washed in binding buffer (2x10 minutes), followed by another brief rinse with deionised water and blown dry. Slides were apposed to a phosphorimaging plate for 2-3 days, developed on a Phosphorimager (Molecular Devices, USA), the image displayed and the density of binding quantified using ImageQuaNT<sup>TM</sup> (Figure 2.5). Data was analysed using GraphPad PRISM (Section 2.8).

## Saturation binding protocol

Saturation binding was determined using [ $^{33}$ P]-H2RLX, in a concentration range of 50 - 1500 pM. Specific binding was calculated by the total binding – non-specific binding (defined by 10  $\mu$ M H2RLX) for each concentration. Data was analysed using GraphPad PRISM (Section 2.8) and the pK<sub>D</sub> and B<sub>max</sub> values calculated.

## **Competition binding protocol**

Competition binding studies were carried out using [<sup>33</sup>P]-H2RLX at a concentration of 100 pM. The ability of relaxin-like analogues to compete with the radioligand for the relaxin receptor was tested over a concentration range of 10 nM – 10  $\mu$ M, with non-specific binding determined using 10  $\mu$ M H2RLX. Data was analysed using GraphPad PRISM (Section 2.8) and pK<sub>i</sub> values calculated.

## Figure 2.5

## Quantitative autoradiography

Tissue sections, cut on a cryostat at 10 $\mu$ m and collected on poly-L-lysine coated slides, were placed in a moist chamber for the binding experiments. [<sup>33</sup>P]-H2RLX, in the presence or absence of cold relaxin analogues, was incubated on the tissue sections for 90 minutes before slides were washed and exposed to a phosphorimager plate for 3 days. The phosphorimager plates were developed and analysed using ImageQuant<sup>TM</sup>.



## 2.3.2.3 Phosphorimaging

Receptor expression and saturation binding images were expressed as arbitrary phosphorimager units that were converted into dpm/mm<sup>2</sup> using  $[\gamma^{-33}P]$ -ATP standards.

## 2.3.2.4 Preparation of [γ-<sup>33</sup>P]-ATP reference standards

Rat skeletal muscle tissue was ground in a pestie at 0°C to produce a smooth thick paste which was weighed out into eleven lots. Serial dilutions of  $[\gamma^{-33}P]$ -ATP were mixed thoroughly with each separate paste lot and then carefully inserted into prepared moulds of aluminium foil. Moulds were snap frozen in liquid nitrogen and stored at -70°C. Frozen tissue standards were mounted in OCT in the cryostat, cut into 10 µm sections at -18°C and either mounted into poly-L-lysine coated microscopes slides to be apposed together with slide-mounted uterus sections or 30 sections of each mould were placed in liquid scintillation vials for direct determination of radioactivity.

## 2.3.2.5 High resolution autoradiography using photographic emulsion

Tissue sections were processed using the  $[^{33}P]$ -H2RLX quantitative autoradiography binding protocol (Section 2.3.2.2), with slides being incubated with 100 pM  $[^{33}P]$ -H2RLX for 90 minutes to determine total binding with non-specific binding defined by incubation with unlabelled H2RLX (1  $\mu$ M). Labelled sections were washed and dried with a stream of air. Slides were stored overnight in a vacuum-sealed container containing silica gel to ensure that they were completely dry before commencing the emulsion protocol.

LM-1 hypercoat emulsion (Amersham Pharmacia Biotech, U.S.A.) was melted at 42°C (under darkroom conditions) for 1 hour and then transferred to a dipping chamber. Each slide was dipped into the emulsion twice, drained and allowed to dry in a light-proof box overnight. Slides were then transferred into a slide box,

wrapped in foil to prevent exposure to light and stored at 4°C for 3 weeks. Prior to development, slides were equilibrated at room temperature for 1 hour. Emulsion was developed in D-19 developer (5 minutes; Schreiber Photographics, Vic, Aust), stop solution (1 minute; Schreiber Photographics), fixative (10 minutes; Schreiber Photographics) and gently rinsed in running water (15 minutes). Slides were air-dried and examined under a dark-field microscope. For histological analysis, slides were stained with Gill's haematoxylin (distilled water 730 mL, ethylene glycol 250 mL, haematoxylin 2g, sodium iodate 17.6g and glacial acetic acid 20 mL) and Eosin (1% solution).

## 2.3.3 High through-put cellular binding assay

#### 2.3.3.1 Cell preparation

## THP-1 cells

From a large flask of cells, a sample was removed with a glass pipette and counted (Section 2.1.3) with Trypan Blue dye (0.5%; Trace Biosciences, NSW, Aust) to identify the proportion of live cells. The remaining cells were pelleted (5 minutes, 1000xg), media removed and the pellet was washed with sterile HBS to remove the dead cells sitting loosely on the cell pellet. The cell pellet was resuspended in binding buffer (20 mM HEPES, 1.5 mM CaCl<sub>2</sub>, 50 mM NaCl, 1% BSA, 0.1 mg/mL Lysine, 0.01% NaN<sub>3</sub>, pH 7.5) and recounted to confirm that the concentration of cells was approximately  $4x10^6$  cells/100 µL.

## HEK293T cells

Cells were washed and lifted from the tissue culture flask using 0.45 mM EDTA solution, pelleted (5 minutes, 1000xg), the supernatant removed and the cells resuspended in binding buffer (20 mM HEPES, 1.5 mM CaCl<sub>2</sub>, 50 mM NaCl, 1% BSA, 0.1 mg/mL Lysine, 0.01% NaN<sub>3</sub>, pH 7.5).

## 2.3.3.2 [<sup>33</sup>P]-H2RLX cell binding assay

A high through-put cell binding assay was set up in a 96 well plate format. The 96 well plates were coated with Sigmacote® to prevent peptide binding. Final assay volume was 100  $\mu$ L/well, consisting of [<sup>33</sup>P]-H2RLX (20  $\mu$ L) ± cold relaxin and relaxin-like analogues (10  $\mu$ L) with resuspended cells (50  $\mu$ L), which was incubated for 90 minutes at 25°C. Cells were harvested onto GF/C filters (Skudtek Scientific, Vic, Aust) that had been pre-soaked in 0.5% polyethylenimine (PEI, 4°C overnight). Filters were dried (42°C), liquid scintillant added (30  $\mu$ L/well) and the filters incubated overnight at room temperature. Filters were counted on a Topcount scintillation counter (Perkin-Elmer, Vic, Aust) and data was analysed using GraphPad PRISM (Section 2.8).

## Saturaties binding protocol

Saturation binding was determined using [ $^{33}$ P]-H2RLX, in a concentration range of 25 pM – 3 nM. Specific binding was calculated by the total binding – non-specific binding (10  $\mu$ M H2RLX) for each concentration. Protein concentration was determined as described in Section 2.6 and data was expressed as fmol/mg protein. Data was analysed and the pK<sub>D</sub> and B<sub>max</sub> values were calculated using GraphPad PRISM (Section 2.8).

## Competition binding protocol

Competition binding was determined using [ $^{33}$ P]-H2RLX at a concentration of 100-200 pM. The ability of relaxin-like analogues to compete with the radioligand for the relaxin receptor was tested over a concentration range of 10 nM – 3  $\mu$ M, with non-specific binding determined using 10  $\mu$ M H2RLX. Data was analysed using GraphPad PRISM (Section 2.8) and pK<sub>i</sub> values calculated.

## 2.4 Cellular functional assays

## 2.4.3 cAMP assay

CHO-K1 cells were transiently transfected with LGR7 receptor cDNA (see Section 2.1.4). 16 hours prior to experiment, cells were plated at  $1 \times 10^5$  cells/well into 96-well tissue culture plates in normal growth media (see Section 2.1.1.2). Incubation medium was prepared (phenol-red free media, 0.1% FBS, 2 mM IBMX, 1  $\mu$ M forskolin) and drug dilutions prepared in PBS. Growth media was aspirated from cells and incubation media added (90  $\mu$ L/well). Cells were exposed to relaxin or relaxin analogues (10  $\mu$ L; 10 pM – 30 nM) in duplicate wells, and incubated at 37°C for 30 minutes. Stimulation was halted by the addition 0.5M HCl (25  $\mu$ L/well). Plates were stored at -20°C until analysis.

cAMP was determined using the NEN FiashPlate RIA (Perkin-Elmer, Vic, Aust) according to the manufacturers instructions. Briefly, cAMP standards were diluted in water (0.1 – 15 pmol/well) and the experimental samples defrosted. Assay detection buffer was prepared (50 mM sodium acetate trihydrate, 0.8mM EDTA, pH 6.2) and added to the FlashPlate, 90  $\mu$ L/well for unknown samples and 80  $\mu$ L/well for cAMP standards. Unknowns (10  $\mu$ L/well) and cAMP standards (20  $\mu$ L/well) were added to the FiashPlate. [<sup>125</sup>I]-cAMP was diluted 1:100 in tracer diluent (50 mM sodium acetate trihydrate, 0.1% NaN<sub>3</sub>, 12 mM calcium chloride, pH 6.2) and aliquotted into FlashPlates (100  $\mu$ L/well). The FlashPlates were sealed, incubated at room temperature overnight and read on a Topcount scintillation counter (Perkin-Elmer, Vic, Aust). Data was analysed using GraphPad PRISM (Section 2.8).

#### 2.4.2 AlphaScreen<sup>™</sup> cAMP assay

HEK293T cells stably expressing the LGR7 or LGR8 receptors were plated into a 96-well tissue culture plate at  $1 \times 10^4$  cells/well in normal culture media and incubated for 24 hours under normal cell culture conditions. Media was aspirated and assay stimulation buffer (20µL; Hank's balanced salt solution (HBSS), 1 mg/mL BSA, 0.5 65

mM IBMX, 0.5 M HEPES, pH 7.4) added to each well. Cells were exposed to relaxin and relaxin analogues (20 $\mu$ L; 1 pM - 1  $\mu$ M) for 30 minutes (37°C), and stimulation was performed in duplicate wells. Stimulation solution was aspirated, cells were lysed with 100  $\mu$ L lysis buffer (0.5 M HEPES, 1 mg/mL BSA, 0.5 mM IBMX, 0.3% Tween-20, pH 7.4) and snap frozen at -70°C. Plate was stored at -20°C.

cAMP accumulation was analysed using the cAidP Alphascreen Kit (Perkin-Elmer, Vic, Aust) according to the manufacturers instruction with modifications. Experimental samples were defrosted and cAMP standards (1  $\mu$ M - 10 pM) diluted in detection buffer (0.4% HBSS, 3mM HEPES, 0.2% Tween 20, 0.1% BSA, pH 7.4). 5  $\mu$ L of unknown samples and cAMP standards were transferred into a white 384-well plate. Acceptor bead (anti-cAMP acceptor bead diluted in detection buffer) and Donor bead (streptavidin donor beads diluted in detection buffer, 1M biotinylated cAMP) solutions were prepared under light limiting conditions in light sealed containers and keep on ice. In minimal light, 5  $\mu$ L of Acceptor bead solution was added to each well and incubated in the dark for 30 minutes. Also in light limited conditions, the Donor bead solution was added (15  $\mu$ L/well), the plate sealed and incubated in the dark overnight. The plate was read by the Fusion<sup>TM</sup>  $\alpha$  microplate reader (Perkin-Elmer, Vic, Aust). Data was analysed using GraphPad PRISM (Section 2.8).

## 2.5 Determination of RNA and DNA concentration

The absorbance of DNA and RNA samples were measured by UV spectrophotometer (Ultrospec 2000, Pharmacia Biotech). A reference standard of sterilised water was set before measuring absorbance of the samples at 260 and 280 nm. Concentration was calculated using a constant of 50 ng/ $\mu$ L for double strand DNA, 20 ng/ $\mu$ L for single strand DNA and 40 ng/ $\mu$ L for RNA. Furity of the samples was determined using a ratio of the absorbance readings, 260/280 nm, with a ratio of 1.8 depicting a pure sample.

## 2.6 Protein determination

Protein concentration was determined by the Lowry method (Lowry *et al.*, 1951). Cell samples were diluted 10-20 fold, and a 50  $\mu$ L sample made up to 150  $\mu$ L with distilled water. BSA standards (0 – 150  $\mu$ g) were made up in 150  $\mu$ L water. All unknowns and standards were measured in duplicate. Samples were digested with 2 M NaOH (50  $\mu$ L) and incubated at room temperature for 30 minutes. Alkaline copper solution (0.01% CuSO<sub>4</sub>, 0.02% NaK Tartrate, 2% anhydrous Na<sub>2</sub>CO<sub>3</sub>; 1 mL) was added and the incubation continued for 10 minutes. Folin's reagent (30%, 100  $\mu$ L) was added, the samples vortexed and covered with foil. The colour change was allowed to develop in the dark for 30 minutes. Absorbances were measured at a wavelength of 750 nm on a Novaspec II spectrophotometer (Pharmacia LKB Biochrom, England). BSA standards were used to produce a standard curve (GraphPad PRISM) and unknown protein values interpolated from this curve.

## 2.7 Protein analysis

Protein samples were visualized for Western blotting and immunostaining (Harlow *et al.*, 1988). Polyacrylamide gels (10%-16% acrylamide) were produced in the Mini Protean 3 electrophoresis module (BioRad, Hercules, CA, USA). The separation gel (12% acrylamide gel) comprised 30% acrylamide (4.8 mL), Tris/SDS buffer (1.5 M Tris, pH 8.8, 0.4% SDS; 6 mL), 5% glycerol (0.8 mL), 6 mg ammonium peroxide sulphate and TEMED (5  $\mu$ L). The separation gel was allowed to set before assembly of the stacking gel, comprised of 30% acrylamide (0.5 mL), Tris/SDS buffer (0.5 M Tris, pH 6.8, 0.4% SDS; 2.5 mL), distilled water (2 mL), bromophenol blue solution (4 mg in 0.5M Tris, pH 6.8; 20  $\mu$ L) 2.5 mg ammonium peroxide sulphate and TEMED (5  $\mu$ L). Once set, gels were assembled into the electrophoresis module with running buffer (25 mM Tris, 0.2 M glycine, 0.1% SDS), samples loaded with a PPS marker and separated at a constant 150 V (2 hours).

Following electrophoresis, the separation gel was soaked in transfer buffer 2 (25 mM Tris, 20% methanol, pH 10.4) for 5-10 minutes with gentle shaking. The separation

gel was electrotransferred onto a Hybond-C Extra nitrocellulose membrane (0.45  $\mu$ m pore size) with a semi-dry blotter (CBS Scientific, Del Mar, CA, USA) for 30 minutes at a constant current in the range 2-2.5 mA/cm<sup>2</sup> of membrane. Electrotransfer consisted of separation gel, nitrocellulose membrane and Whatman 3MM paper in the order of: positive electrode, 6 sheets of 3MM paper pre-soaked in transfer buffer 1 (0.3 M Tris, 20% methanol, pH 10.4), 3 sheets of 3MM paper pre-soaked in transfer buffer 2, nitrocellulose membrane, separation gel, 6 sheets of 3MM paper pre-soaked in transfer buffer 3 (38 mM Tris, 10 mM  $\beta$ -alanine, 20% methanol, pH 9.4) and negative electrode.

At the completion of the electrotransfer the membrane was washed in Tris Buffer Saline (TBS; 20 mM Tris, 140 mM NaCl, pH 7.6) for 5 minutes. To eliminate nonspecific binding the membrane was soaked in blocking buffer (TBS, 5% skim powdered milk, 0.1% Tween 20; 1 hour, room temp) and then washed for 3x5 minutes with TBS-T (TBS, 0.1% Tween 20). The membrane was incubated by gently shaking overnight (4°C) with primary antibody diluted in primary antibody buffer (TBS-T, 5% Fraction V BSA). The membrane was washed with TBS-T (3x5 minutes) and primary antibody detected with horse radish peroxidase (HRP) linked secondary antibody diluted in blocking buffer (1 hour at room temperature). Membrane was washed with TBS-T (3x5 minutes) before being exposed with enhanced chemiluminescence, Lumi-light Western Blotting Substrate (Roche, IN, USA) according to the manufacturer's instructions. The membrane was exposed to Hyperfilm ECL film (Amersham Bioscience) under darkroom conditions and developed in D-19 developer (1-5 minutes; Schreiber Photographics, Vic, Aust), stop solution (1 minute; Schreiber Photographics), fixative (5 minutes; Schreiber Photographics) and gently rinsed in water (5 minutes). Film was analysed using ImageQuaNT<sup>TM</sup>.

\$

## 2.8 Data analysis

Quantitative autoradiography tissue sections were analysed on ImageQuaNT<sup>TM</sup> (Version 4.1; Molecular Dynamics, U.S.A.) with density of binding expressed as arbitrary phosphorimager units and converted into dpm/mm<sup>2</sup> using  $[\gamma^{-33}P]$ -ATP standards (Section 2.3.2.4).

Saturation binding data from quantitative autoradiography and cellular binding assays were performed in duplicate and entered into GraphPad PRISM (GraphPad Inc., San Diego, U.S.A.). Specific binding was calculated as total binding - non-specific binding for each concentration of radioligand and the cell protein concentration was determined where applicable (Section 2.6). Data was plotted as concentration of [<sup>33</sup>P]-H2RLX (M) vs specific binding as dpm/mm<sup>2</sup> for quantitative autoradiography or fmol/mg protein for cellular binding assays, as mean  $\pm$  S.E.M. of n experiments. The radioligand affinity (pK<sub>D</sub>) and receptor expression (B<sub>max</sub>) were calculated by GraphPad PRISM.

Competition binding data from quantitative autoradiography and cellular binding assays were performed in duplicate and entered into GraphPad PRISM. The specific binding was calculated as total binding - non-specific binding for each concentration of cold relaxin analogue and was calculated as a percentage of total specific binding. The data was plotted as concentration of relaxin analogue (M) vs % specific binding, as mean  $\pm$  S.E.M. of n experiments. The affinity of the relaxin analogues (pK<sub>i</sub>) was calculated in GraphPad PRISM using the Cheng & Prusoff (1973) equation.

For the cAMP FlashPlate and cAMP AlphaScreen<sup>TM</sup> assays, the concentration of cAMP in each sample was calculated against the cAMP standard curve. The cAMP assay was performed in duplicate and data was converted to a percentage of maximum cAMP response determined by forskolin for the FlashPlate assay and determined by H2RLX for the AlphaScreen assay. Data was plotted as concentration of the agonist vs % maximal response as mean  $\pm$  S.E.M. of n experiments and the pEC<sub>50</sub> was calculated in GraphPad PRISM.

## 2.9 Drugs and reagents

The following drugs were gifts: recombinant human gene 2 RLX (H2RLX, Connetic Corporation, CA, USA); all synthetic relaxin and INSL3 analogues (Dr J. Wade, Howard Florey Institute of Experimental Physiology and Medicine, Vic, Aust); DH<sub>5 $\alpha$ </sub> cells (Dr D. Bulach, formerly from Department of Microbiology, Monash University); stably transfected HEK293T cells expressing the LGR7 or the LGR8 receptors (Ms S. Layfield, Howard Florey Institute of Experimental Physiology and Medicine, Vic, Aust); relaxin polyclonal antibody, relaxin and LGR7 DNA constructs (Dr R. Bathgate, Howard Florey Institute of Experimental Physiology and Medicine, Vic, Aust).

 $[\gamma^{-33}P]$ -ATP, dNTPs, LM-1 Drugs and reagents were purchased as follows: hypercoat emulsion, Hybond-C Extra nitrocellulose membrane, Hyperfilm ECL film, Not I restriction enzyme, Xho I restriction enzyme (Amersham Pharmacia Biotech, Uppsala, Sweden); BigDye Terminator Mix v3.1 (Applied Biosystems, CA, USA); acrylamide (30%), ammonium peroxide sulphate, TEMED (Bio-Rad Laboratories, Hercules, CA, USA); Fraction V BSA (Boehringer Manneheim, Germany); skimmilk powder (Diploma, Vic, Aust); polyethylene glycol (PEG) (ICN Biomedicals Inc, OH, USA); 3MM Whatmann paper (Interpath, West Heidelberg, Vic, Aust); DMRIETM, LIPOFECTAMINETM, OPTI-MEM®, Pfx polymerase, ProBondTM Purification System, SFV gene expression system, Sindbis expression system, SDS, X-gal, yeast extract, zeocin (Invitrogen, Mount Waverley, Vic, Aust); HRP-linked anti-rabbit antibody, Pml I restriction enzyme, Sph I restriction enzyme (New England Biolabs, Beverly, MA, USA); agar bacteriological (agar No. 1), casein hydrolysate, tryptone (Oxoid Limited, Basingstoke, Hampshire, England); liquid scintillant Microscint O (Perkin-Elmer, Mount Waverley, Vic, Aust); cAMP-dPK, RNase-free TE, Wizard PCR Preps DNA Purification System, Wizard Plus Midipreps DNA Purification System, Wizard Plus Miniprep DNA purification System (Promega, Madison, WI, USA); agarose LE, ampicillin, Lumi-light Western Blotting Substrate (Roche Diagnostics, IN, USA); nickel IDA agarose beads (Scientifix, Cheltenham, Vic, Aust); ammonium acetate (RNase free), ATP, BSA,

DMSO, eosin, Folin and Ciocalteu's phenol reagent, forskolin, glycerol, haematoxylin, IBMX, pertussis toxin, PMSF, polyethylenimine (PEI), poly-L-lysine, Sigmacote® sodium azide, Tween 20 (Sigma Chemical Company, St.Louis, MO, USA); GF/C filters and filter paper (Skudtek Scientific, Mount Waverley, Vic, Aust); PCR Script<sup>TM</sup> Amp Electroporation-Competent Cell Cloning Kit, QuikChange<sup>TM</sup> Site-directed Mutagensis Kit (Stratagene, CA, USA); αMEM media, DMEM/Hams F12 media, glutamine, penicillin/ streptomycin, RPMI 1640 media, trypan blue, trypsin/EDTA solution (Trace Biosciences, Castle Hill, NSW, Australia);

All other chemicals and reagents were of either analytical or laboratory grade.

# **Chapter Three**

## Structure-activity relationships of relaxin analogues at the human relaxin receptor

## 3.1 Introduction

To pharmacologically characterise the structural interaction between relaxin and its receptor, functional and binding assays were used to screen different relaxin analogues for relaxin-like activity. Historically, the most commonly used functional assay for the determination of the potency and efficacy of relaxin analogues was the mouse interpubic ligament assay where administration of relaxin caused a specific elongation of the interpubic ligament (Section 1.4.1.2). The ability of the analogues to produce relaxin-like activity was correlated with the ability of the analogues to compete with binding in mouse brain membrane preparations (Schwabe et al., 1994). Examination of species homologues of relaxin showed that H2RLX and PRLX were of equal potency, while HIRLX and mouse RLX were only partially active (Schwabe et al., 1994) at the mouse relaxin receptor. Interestingly, H2RLX, H1RLX and PRLX all displayed high affinity in the mouse brain receptor binding assay, while mouse RLX displayed significantly weaker affinity for the receptor (Schwabe et al., 1994). Modification to the H2RLX A chain sequence reduced relaxin activity (Section 1.2.4.1), an action thought to reflect the disruption of the structural integrity of the peptide rather than a direct role of these amino acids in interacting with the receptor. Modifications to the B chain of H2RLX (Section 1.2.4.2) highlighted the importance of the conserved cysteine residues in maintaining peptide integrity and relaxin activity (Schwabe et al., 1994), and the important binding motif comprising of Arg(B13)-X-X-X-Arg(B17)-X-X-Ile(B20) (Figure 1.2) positioned along the B chain  $\alpha$ -helix (Büllesbach et al., 2000; Büllesbach et al., 1992). The highly conserved glycine residues immediately adjacent to the cysteine residues are critical for correct structural folding of insulin and it has been hypothesised that they maintain the same role in relaxin (Shire et al., 1991).

More recently, interactions between relaxin analogues and the rat relaxin receptor have been explored using the rat atrial bioassay and quantitative autoradiography using [<sup>33</sup>P]-H2RLX, and sections of rat uterine or atrial tissue. Both natural isoforms of H2RLX (B33 & B29), H1RLX (B33) and PRLX were equally effective at producing chronotropic and inotropic responses in the atrial bioassay, while RatRLX

73

Chapter 3: structure-activity relationships of relaxin analogues at the human relaxin receptor and H1RLX (B29) were less effective (Tan *et al.*, 1998). Competition studies in sections of rat atria using quantitative autoradiography indicated that H2RLX and PRLX have equal binding affinity for the rat receptor while RatRLX had lower affinity (Tan *et al.*, 1999). From the results obtained at the mouse and rat relaxin receptors, it appears that the native ligand for each receptor appears to show the weakest interaction. Of the analogues tested, mouse RLX and RatRLX displayed the lowest affinity and efficacy at the mouse and rat receptors respectively, while H2RLX and PRLX analogues consistently produced the strongest interaction.

The important binding motif identified in the B chain, Arg(B13)-X-X-X-Arg(B17)-X-X-Ile(B20), is also found in a close relative of relaxin, INSL3. However in INSL3 the arginine residues are displaced 4 amino acids to the C-terminus, X(B13)-X-X-X-Arg(B17)-X-X-Val(B20)-Arg(B21). The Ile(B20) is not as highly conserved as the arginine residues but is only replaced with large hydrophobic residues, such as valine as seen in PRLX and also in the INSL3 sequence (Büllesbach et al., 2000). As ruminants do not express a functional relaxin gene and INSL3 is thought to be a replacement for relaxin in these animals (Bathgate et al., 2001), the sheep INSL3 was modified to contain a relaxin binding motif in the correct position. Four INSL3 analogues were designed and synthesised with minimal amino acid modification; INSL3-A His(ArgB13), INSL3-D His(ArgB13)-His(GluB14)-Phe(LysB15), INSL3-E His(ArgB13) & Arg(AlaB21), INSL3-G Arg(AlaB21) (Claasz et al., 2001; Dawson et al., 2001). At the rat relaxin receptor INSL3-A and INSL3-D containing relaxin and INSL3 arginine residues were found to have higher relaxin activity than the native INSL3 (Tan et al., 2002). Interestingly, the INSL3-E analogue, containing only the relaxin positioned arginines, failed to produce a relaxin-like response in the isolated rat atrial bioassay, as did INSL3-G which contained no correctly positioned arginines. These results could indicate that INSL3 is not a replacement for relaxin in ruminants or that the binding motif in the B chain does not solely determine relaxin activity.

While the interaction between relaxin analogues and the relaxin receptor has been extensively examined in experimental animals, it has proved more difficult to investigate the characteristics of the human relaxin receptor because of the limited

74

 $\langle i \rangle$ 

 $\mathbf{P}$ 

Chapter 3: structure-activity relationships of relaxin analogues at the human relaxin receptor availability of human tissue. However a THF-1 cell line derived from human monocyte cells was found to express a receptor that specifically bound relaxin yet had little affinity for other members of the insulin-gene super family (Parsell *et al.*, 1996). The aim of this study is to examine the interaction of various relaxin and INSL3 analogues at the human relaxin receptor expressed in THP-1 cells.

## 3.2 Methods

The affinities of various analogues of human relaxin, PRLX and RatRLX, as well as several sheep INSL3 analogues (Figure 3.1) were examined at the human relaxin receptor. Due to the difficulty and limited amount of INSL3 analogues synthesised, only native sheep INSL3 and the INSL3-D and INSL3-G analogues were available for investigation.

#### **3.2.1** Development of the high through-put 96 well binding assay

## 3.2.1.1 Harvesting filter pre-treatment

A major problem associated with the use of  $[^{33}P]$ -H2RLX as a radioligand is high non-specific binding to the harvesting filter. The effect of several pre-binding treatments was explored on this non-specific binding, together with the use of two different binding buffers. The pre-treatments included soaking for 30 minutes in cold H2RLX (1µM), Sigmacote® (Sigma Chemical Company, MO, USA), in 0.5% PEI and a non-treated filter as a negative control. The influence of two binding buffers on non-specific binding was also examined (Tan *et al.*, 1999: 25 mM HEPES, 300 mM KCl, pH 7.2, 1 mg/ml BSA; Parsell *et al.*, 1996: 20 mM HEPES, 1.5 mM CaCl<sub>2</sub>, 50 mM NaCl, pH 7.5, 1% BSA, 0.1 mg/ml Lysine, 0.01% NaN<sub>3</sub>).

Binding experiments were conducted in siliconised 96 well plates, in 100  $\mu$ L volume containing 200 pM [<sup>33</sup>P]-H2RLX in binding buffer. Samples were harvested on filter plates pre-treated as described above and counted. All filter treatments reduced non-specific binding to the filter, with 0.5% PEI filter pre-treatment combined with incubation binding buffer #2 being the most effective (Table 3.1) and therefore the filter pre-treatment and binding buffer used in all following experiments. Non-specific binding was further reduced when the harvesting filter plate was pre-soaked in 0.5% PEI at 4°C overnight prior to binding experiments.
# The amino acid sequences of the relaxin and INSL3 analogues.

An amino acid sequence comparison of the relaxin and INSL3 analogues examined in this study. The A chain sequence for the H1RLX, H2RLX and INSL3 analogues remains unchanged while the B chain sequence differs in the corresponding analogues. The cysteine (C) residues that form the disulphide bonds (blue) are aligned and highlighted (pink). The arginine (R) and isoleucine (I) residues important for relaxin-receptor binding (green) are situated in the B-chain. The amino acid substitutions in sheep INSL3 to form the INSL3 analogues are highlighted in the B chain (yellow). Chapter 3: structure-activity relationships of relaxin analogues at the human relaxin receptor

<u>A Chain</u>													E		a a																					
				1				5					- 8 14	0				- E - 14	5				20													
HIRLX				R			v	, . A	L	F					<b>1</b> T.	Ŧ	c					a	LU T		v			I								
H2RLX				ō	. <u>.</u>	Ţ	8	A	L		N	t 1		•	і я	v	6		* 17	- <b>R</b>	. P	я	T.			. 1										
HJRLX				D	v	I	Ä	G	L	9	3	8			ĸ	W	G		8	5	8	Z	I	3	8	L										
PRLX						R	M	T	Ľ	s	Z	7			Q	v	G		I	Ŕ	ĸ	D	I	А	R	L		1								
Ret RLX				Z	8	Ģ	A	L	Ţ	8	£	9			н	I	G		T	R	R	8	I	A	ĸ	L										
SheepINSL3				A	T	A	. V	N	P	X	R	H	r <b>K</b>		L	8	G	2	T	R	0	D	L	L	T	L	1	₽	H	:						
P Chain														ļ																						
<u>D Cuxin</u>								_																												
TION V (04 33)				1				5					10	1		_	_	15		-	_		20			-		25					30			
HIKLA (82-33)					K	*	ĸ	Ð	D	<b>v</b>	I	ĸ	<u>ь</u>		G	S.	E	L	Y		X	Q		A	I	E	G	M	8	Ť	W	8	ĸ	R	8	L
HIKLA (84-49)					ĸ	W	ĸ	D	5	v	I	ĸ	L	I	G		E	L	V		Å	8		A	I		G	M	5	T	W	S				
11 KLA (D2~20)					ĸ	W	K	D	0	v	1	K	. L.		G	F	E	Ŀ	V	A	A	Q	Ľ	A	I		Ģ	M	8							
ПІКLA ( <i>D2-24</i> )					ĸ	W	ĸ	D	B	v	I	ĸ	Ľ		Ģ		£	L	v	Ņ	A	Q		A	I											
H2RLX (B33)				n	8	W	м	E	E	v	I	ĸ	L		G		E	L	v		A	Q	Ì,	λ	I	Ű	G	м	8	т	W	8	ĸ	R	3	ъ
H2RLX (B29)				D	8	W	M	E	E	۷	I	ĸ	L	1	G	i.	z	L	۷		λ	Q	ł	A	I		G	M	8	T	W	8				-
HJRLX					R	A	x	₽	¥	G	v	R	L		G		£	F	τ	i.	A	v		F	T	Ĩ	G	G	8	R	W					
PRLX					0	8	т	N	D	F	r	ĸ	A		G		£	L	v	ŧ.	I.	w	v	R	T	2	a	s	v	8	w	G	R	т		T.
RetRLX	R	V	8	Ľ	E	W	M	D	Q	V	I	9	V		G		G	Y	Å		A	¥	Ż	E	v		G	Ā	8	v	G	R	ľ	Ā	L	-
SheepINSL3					x	x	ଢ	R	A	₽	R	ĸ	L	Į	G	н	н	F	v	8		L	v	R	L		G	G	P	R	W	8	ŝ	R	Ð	G
INSL3-A					A	A	ō	E.	A	₽	E	K	L		G	R	H	F	v	а	X	L	v	R	Ľ	f	G	Ğ	P	R	W	s	8	E	ם	Ğ
INSL3-D					Ā	A	ā	E	Å	P	E	ĸ	L	Ī	G	R	ĸ	L	v		λ	L	v	R	L		G	G	P	8	N	8	8	R	5	ē
INSL3-E					A	A	õ	Z	A	P	E	ĸ	L	J	G	R	H	F	v		λ	L	v	Ä	ī		G	G	- P	R	w	9	a	Ē	D	a
INSL3-G					A	A	ō	E	A	Þ	E	ĸ	Ţ.		G	н	H	r	v		A	т.	v	•	Ŧ.	2	0	ā	Đ			õ	~	7	ñ	

Ċ;

# Table 3.1

# Filter pre-treatment table

Four harvesting filter pre-treatments with two different binding buffers were examined to find the combination that would reduce non-specific [33P]-H2RLX binding to the filter plate. All filter pre-treatments reduced non-specific binding with the most effective combination being the 0.5% PEI pre-treatment with binding buffer 2.

Filter Treatment	Untreated	1 µM H2RLX	Sigmacote®	0.5% PEI
Binding Buffer 1 (Tan <i>et al.</i> , 1999)	100%	20%	37%	20%
Binding Buffer 2 (Parsell et al., 1996)	100%	31%	21%	13%

# Chapter 3: structure-activity relationships of relaxin analogues at the human relaxin receptor

#### 3.2.1.2

The effect of the number of THP-1 cells used per well was examined to maximise specific binding. Since many studies have used membrane preparations for binding experiments, membrane preparations from THP-1 cells and from the entire rat cortex tissue, including the cingulate, frontal, insular, motor, parietal and perirhinal cortices, were also examined. THP-1 cells were prepared as described in Section 2.3.3.1, resuspended in binding buffer and counted. THP-1 cell suspensions were split into four and diluted to achieve a final concentration to give 2, 1.5, 1 and 0.5 million cells per well in a volume of 100  $\mu$ L. The four dilutions were divided into two, with one sample maintained for whole cell binding while the other was processed in a Dounce homogeniser to produce a membrane preparation. Rat cerebral cortex was obtained by anesthetising and killing a rat, and removal of the brain which was placed on ice for 5 minutes. Cortex was dissected from the remaining brain tissue, weighed and homogenised in 50x volume of Tris buffer (10 mM Tris, pH 7.4 @ 4°C, 10 µM PMSF, 250 mM Sucrose). Membrane preparations were resuspended in binding buffer and diluted to deliver 2.5, 1.25, 0.9 and 0.6 mg wet weight membrane homogenate per well in 100 µL.

Experiments were performed in siliconised 96 well plates, total well volume 100 µL containing 200 pM [<sup>33</sup>P]-H2RLX (20  $\mu$ L), the cell or membrane sample (50  $\mu$ L) and non-specific binding was determined with 10 µM H2RLX (10 µL). Samples were harvested onto pre-treated filter plates as described in Section 2.3.3.2, filters were dried (42°C), liquid scintillant added (30 µL/well) and plates were counted. Total binding to non-specific binding ratio with intact THP-1 cells decreased as the cell number decreased, with 2 million cells giving the optimal ratio (Figure 3.2a). THP-1 membranes displayed higher non-specific binding (Figure 3.2b), while rat cortical membranes showed very little specific binding (Figure 3.2c). 2 x 10<sup>6</sup> THP-1 cells/well was chosen for all further experiments.

Chapter 3: structure-activity relationships of relaxin analogues at the human relaxin receptor

# Cell number and setup

÷ .

# Analysis of cell number/well and cell type to obtain optimal specific binding.

The optimal number of cells and type of cell preparation was examined to obtain maximal specific binding, shown by the difference between total binding and non-specific binding . A) the whole cell preparation of THP-1 cells showed optimal specific binding at 2 million and 1.5 million cells/well. B) the THP-1 cell membrane preparation displayed higher non-specific binding compared to the whole cell preparation while C) the rat cortical membrane preparation showed no specific binding.

Chapter 3: struct	
4000-	
3000-	
표 2000·	
1000-	
0.	
B) 4000-	
3000	
E 2000	
1000	
0	
C) 4000	
3000	
변 3 2000	
1000	
0	

#### ture-activity relationships of relaxin analogues at the human relaxin receptor



# 3.2.2 Cell culture

THP-1 cells were grown in suspension and maintained in RPMI 1640 media containing glutamine (2 mM), penicillin (100 units/mL), streptomycin (100  $\mu$ g/mL) and heat inactivated foetal bovine serum (FBS; 10% v/v). Cells were maintained within the density range of 2 x 10<sup>5</sup> and 1 x 10<sup>6</sup> cells/mL to ensure optimal growth and survival. Cells were split and cell culture performed as described in Section 2.1.1.1.

#### 3.2.2.1 THP-1 cell preparation

THP-1 cells were prepared as described in Section 2.3.3.1 on the day of experimentation. Cells resuspended in binding buffer were counted to adjust the concentration of cells to approximately  $4 \times 10^6$  cells/100µL to ensure that the experiments were performed with  $2 \times 10^6$  cells/well. Cells resuspended in binding buffer were kept on ice until plated out into the 96-well plate at 50 µL/well.

# 3.2.3 High through-put 96 well binding assay

The THP-1 cell binding assay was set up as described in Section 2.3.3.2. The total volume used was 100  $\mu$ L/well, comprising 20  $\mu$ L of 200 pM [<sup>33</sup>P]-H2RLX ± 10  $\mu$ L unlabelled relaxin-like analogue and 50  $\mu$ L cell suspension, made up to 100  $\mu$ L with binding buffer. All experiments were performed in duplicate. Following a 90 minute incubation at 25°C, cells were harvested onto GF/C filter plates (Skudtek, Vic, Aust) presoaked overnight (0.5% PEI, 4°C). Filter plates were dried at 42°C, Microscint O (Perkin-Elmer, Vic, Aust) added (30  $\mu$ L/well) and equilibrated overnight. Filters were counted on a Topcount scintillation counter (Perkin-Elmer, Vic, Aust) and data was analysed with GraphPad PRISM (Section 2.8).

# 3.2.3.1 [<sup>33</sup>P]-H2RLX (B33) preparation

H2RLX (B33) was labelled with  $[\gamma^{-33}P]$ -ATP using a procedure based on that of Osheroff *et al* (1990) as described in Section 2.3.1.

# 3.2.3.2 Saturation binding protocol

Saturation binding assays were used to determine the affinity and maximum number of binding sites ( $B_{max}$ ) of human relaxin receptors on THP-1 cells using [<sup>33</sup>P]-H2RLX. Saturation binding assays were performed in a concentration range of 25 pM – 1 nM. Specific binding was calculated by the total binding – non-specific binding (10 µM H2RLX) at each [<sup>33</sup>P]-H2RLX concentration. Protein concentrations were determined as described in Section 2.5.1 and the data was plotted as specific binding (fmol/mg protein) against concentration of radioligand used (pM) (Section 2.8).

#### 3.2.3.3 Competition binding protocol

Competition binding assays were used to calculate the affinity of various relaxin and relaxin-like analogues for the relaxin receptor. Competition binding was performed using [<sup>33</sup>P]-H2RLX at a concentration of 200 pM. As the relaxin analogues were in limited supply, limited concentration ranges were used. The concentration ranges used were; H1, H2, porcine and rat relaxin analogues 100 pM – 30 nM; H3RLX 100 pM – 300 nM; INSL3 1 nM – 300 nM and INSL3-D & -G 10 nM – 3  $\mu$ M. Nonspecific binding was determined using 1  $\mu$ M H2RLX. Results were expressed as a percentage of maximal specific binding against concentration of the relaxin or relaxin-like analogue used (M). pK<sub>i</sub> values were calculated (Section 2.8).

# 3.2.4 Analysis

Results were expressed as mean  $\pm$  S.E.M. of n experiments (Section 2.8). Saturation binding data was analysed with GraphPad PRISM. Radioligand affinity (pK<sub>D</sub>) and receptor expression (B<sub>max</sub>) values were calculated and expressed in pM and fmol/mg protein respectively. Competition binding data was analysed with GraphPad PRISM and relaxin-like affinity (pK<sub>i</sub>) values were calculated by the program using the Cheng & Prusoff (1973) equation. Statistical analysis was completed using GraphPad PRISM statistical programs with the pK<sub>i</sub> data analysed using a Student's ttest and competition curves analysed using a two-way ANOVA. P values of less than or equal to 0.05 were considered significant.

# 3.2.5 Drugs and reagents

All drugs were used as described in Section 2.9. Stock solutions of H1RLX, H2RLX, H3RLX, PRLX and RatRLX and INSL3 analogues were prepared in 0.1% aqueous trifluoroacetic acid (TFA) at a concentration of 10  $\mu$ M. INSL3-D & -G analogue stock solutions were at a concentration of 500  $\mu$ M. All relaxin and relaxin-like analogues were diluted in binding buffer prior to use. All plastic tubes, pipette tips and binding plates used in relaxin binding experiments were coated with SigmaCoat<sup>TM</sup> (Sigma).

# 3.3 Results

# 3.3.1 Characteristics of [<sup>33</sup>P]-H2RLX binding to the human relaxin receptor

[<sup>33</sup>P]-H2RLX binding occurred in a saturable manner to a single binding site expressed in THP-1 cells (Figure 3.3). Expression of the binding site gave a  $B_{max}$  of 1.45 ± 0.62 fmol/mg protein in the whole cell binding assay. [<sup>33</sup>P]-H2RLX showed high affinity for the human relaxin receptor with a pK<sub>D</sub> of 9.37 ± 0.22 (n=4).

#### 3.3.2 Competition for binding at the human relaxin receptor

#### 3.3.2.1 Native human relaxin analogues

All products of the three human relaxin genes displayed affinity for the relaxin receptor (Figure 3.4). Both forms of H1RLX showed high affinity, however H1RLX (B2-33) appeared to have a slightly lower affinity (pK<sub>i</sub> 8.4  $\pm$  0.09, n=6; P<0.04) than the shorter naturally occurring form H1RLX (B2-29) (pK<sub>i</sub> 8.8  $\pm$  0.10, n=6) (Figure 3.4a). The pattern was reversed in the H2RLX peptide (Figure 3.4b) with H2RLX (B33) displaying a slightly higher affinity (pK<sub>i</sub> 9.1  $\pm$  0.16, n=5; P<0.03) for the receptor compared to its shortened form, H2RLX (B29) (pK<sub>i</sub> 8.6  $\pm$  0.11, n=13). H3RLX displayed the lowest, though still high affinity (pK<sub>i</sub> 8.4  $\pm$  0.16, n=8) for the human relaxin receptor (Figure 3.4b). Interestingly H3RLX only displaced 50% of the ligand binding (P<0.0001) while both forms of H1RLX and H2RLX competed for 100% of the binding.

### 3.3.2.2 Species homologues of relaxin

PRLX showed high affinity for the human receptor (pK<sub>i</sub> 8.4  $\pm$  0.09, n=5), not significantly different from H2RLX (B29) (Figure 3.5a), and like the human relaxin completely displaced ligand binding. RatRLX did not compete as effectively, and displayed significantly lower affinity (pK<sub>i</sub> 7.9  $\pm$  0.09, n=7; P<0.0005) than H2RLX 84

(B29) (Figure 3.5b). Due to limited supply of RatRLX the concentration range could not be extended to a level that completely displaced ligand binding.

# **3.3.2.3 Effect of shortening the B chain on affinity of H1RLX**

Shortening of the B-chain of H1RLX from B2-33 to the native B2-29 form appeared to slightly increase the affinity of H1RLX for the human relaxin receptor. Further shortening of the B chain from H1RLX (B2-29) to H1RLX (B2-26) and H1RLX (B2-24) caused a significant decrease in affinity (pK;  $8.1 \pm 0.12$  and approximately estimated 7.5  $\pm$  0.17 respectively, n=6; P<0.004). H1RLX (B2-26) was able to compete for 100% of ligand binding (Figure 3.6a) while the concentration range for H1RLX (B2-24) could not be extended to investigate if it could compete for binding at high concentrations (Figure 3.6b).

#### 3.3.2.4 Relaxin-like affinity of INSL3 analogues

INSL3 was able to compete for [<sup>33</sup>P] H2RLX with low affinity (pK<sub>i</sub> approximately estimated to be  $5.9 \pm 0.32$ ; n=4) (Figure 3.7a). Modification of INSL3 to include a relaxin-like binding motif positioned in the B chain produced INSL3-D which displayed significantly increased relaxin-like interaction with the receptor (P<0.0001), however there was no significant difference in affinity values (pK<sub>i</sub> 6.7 ± 0.11, n=4; P>0.06). Although INSL3-D competed for 100% of ligand binding, it was still 100 fold weaker than H2RLX (B29) (Figure 3.7b). INSL3 was modified to contain neither of the INSL3 or relaxin-like binding motifs. The INSL3-G an logue displayed similar weak relaxin-like interaction (pK<sub>i</sub> 5.4 ± 0.13; n=4) at the human relaxin receptor compared to INSL3, resulting in 1000 fold less affinity than H2RLX (B29) (Figure 3.7c).

[<sup>33</sup>P]-H2RLX saturation binding characteristics at the human relaxin receptor expressed in THP-1 cells.

The specific relaxin radioligand,  $[^{33}P]$ -H2RLX, displayed high affinity (pK<sub>D</sub> 9.37 ± 0.22, n=4) for a single binding site on the relaxin receptor. Natural expression of the relaxin receptor in THP-1 cells gave a B<sub>max</sub> of 1.45 ± 0.62 fmol/mg protein in the whole cell binding assay.



Chapter 3: structure-activity relationships of relaxin analogues at the human relaxin receptor



Competition for binding at the human relaxin receptor expressed in THP-1 cells by the native forms of the human relaxin analogues.

These graphs represent the pooled specific affinity data displayed by the three human relaxin genes for the human relaxin receptor. A) the two forms of human gene 1 relaxin, H1RLX (B2-33) and H1RLX (B2-29), displayed high affinity (pKi 8.4  $\pm$ 0.09 & 8.8  $\pm$  0.10 respectively, n=6) as did the B) two forms of human gene 2 relaxin, H2RLX (B33) (pK<sub>i</sub> 9.6  $\pm$  0.16, n=5) and H2RLX (B29) (pK<sub>i</sub> 8.6  $\pm$  0.11, n=13). The third human relaxin gene, H3RLX, displayed high affinity (pK<sub>i</sub> 8.4  $\pm$ 0.16, n=8) although could only displace 50% of ligand binding (P<0.0001 cf H2RLX).



Chapter 3: structure-activity relationships of relaxin analogues at the human relaxin receptor





87

Competition for binding at the human relaxin receptor expressed in THP-1 cells by species homologues of relaxin.

These graphs represent the pooled specific affinity data displayed by porcine and rat species homologues of relaxin for the human relaxin receptor. The natural ligand for the relaxin receptor, human gene 2 relaxin (H2RLX (B29)), was used for comparison. The species homologues of relaxin showed high affinity for the human relaxin receptor with A) porcine relaxin (PRLX) having similar affinity (pK<sub>i</sub> 8.4  $\pm$  0.09, n=5) to H2RLX, while B) the affinity displayed by the rat homologue (RatRLX; pK<sub>i</sub> 7.9  $\pm$  0.09, n=7) was significantly lower (P<0.005).



Chapter 3: structure-activity relationships of relaxin analogues at the human relaxin receptor

Competition for binding at the human relaxin receptor expressed in THP-1 cells by the human gene 1 relaxin analogues with shortened B chain length.

These graphs represent the pooled specific affinity data displayed by the human gene 1 relaxin analogues for the human relaxin receptor. The proposed natural form of human gene 1 relaxin (H1RLX), H1RLX (B2-29), was used for comparison. Shortening of the B chain by A) 3 amino acids, H1RLX (B2-26) reduced relaxin affinity for the receptor (pKi 8.1  $\pm$  0.12, n=6; P<0.004) which was further reduced with B) removal of another 2 amino acids, H1RLX (B2-24), (pK<sub>i</sub> estimated to be 7.5 ± 0.17, n=6; P<0.004) compared to H1RLX (B2-29).



+1

Competition for binding at the human relaxin receptor expressed in THP-1 cells by sheep insulin-like peptide 3 analogues.

These graphs represent the pooled specific affinity data displayed by the sheep insulin-like peptide 3 (INSL3) analogues for the human relaxin receptor. The natural ligand for the relaxin receptor, human gene 2 relaxin (H2RLX (B29)), was used for comparison. A) native sheep INSL3 displayed very little relaxin-like affinity (pKi estimated to be 5.9  $\pm$  0.32, n=4). B) insertion of the relaxin positioned arginine and intermediate residues (Figure 3.1), INSL3-D, increased relaxin-like affinity (pKi 6.7 ± 0.11, n=4) while C) removal of arginine residues, INSL3-G, showed extremely weak relaxin-like affinity (pK<sub>i</sub> 5.4  $\pm$  0.13, n=4).



# Table 3.2

Summary of the relaxin and INSL3 analogue affinities displayed for the human relaxin receptor expressed in THP-1 cells. Non-specific binding was determined using 1  $\mu$ M H2RLX (B29).

\* denotes P<0.05, \*\* P<0.005 verus H2RLX (B29); † P<0.05, †† P<0.005 verus H1RLX (B2-29).

Analogue	рК <sub>і</sub>	n	
H1RLX (B2-33)	8.4 ± 0.09 †	6	
H1RLX (B2-29)	$8.8\pm0.10$	6	
H2RLX (B33)	9.1 ± 0.16 *	5	
H2RLX (B29)	8.6±0.11	13	
H3RLX	8.4±0.16	8	
PRLX	8.4 ± 0.09	5	
RatRLX	7.9 ± 0.09 **	7	
HIRLX (B2-26)	8.1 ± 0.12 ††	6	
H1RLX (B2-24)	7.5 ± 0.17 ††	6	
INSL3	5.9 ± 0.32 **	4	
INSL3-D	6.7 ± 0.11 **	4	
INSL3-G	5.4 ± 0.13 **	4	

۰.

## 3.4 Discussion

While structure-activity relationships of various relaxin analogues have been determined at mouse and rat relaxin receptors, few studies have investigated the relationships at the human relaxin receptor. This study examined the ability of various relaxins and relaxin-like analogues to interact with the human relaxin receptor endogenously expressed in THP-1 cells. The high through-put THP-1 cell binding assay allowed the interaction between peptides and the receptor to be investigated using minimal amounts of material.

[<sup>33</sup>P]-H2RLX was used in this study as it has been thoroughly characterised and binds to and stimulates the relaxin receptor in a way not distinguishable from H2RLX (Osheroff *et al.*, 1990; Parsell *et al.*, 1996; Tan *et al.*, 1999). In saturation binding assays, [<sup>33</sup>P]-H2RLX displayed high affinity for the human relaxin receptor, consistent with that previously reported at the human relaxin receptor (Osheroff *et al.*, 1995; Parsell *et al.*, 1996). Expression of the relaxin receptor in THP-1 cells was low, although it is difficult to make a direct comparison with binding studies using membrane preparations. Membrane preparations, in contrast to whole cells, concentrate the receptor protein. Previous reports indicate that THP-1 cells express low levels of relaxin receptor, ~275 receptors/cell (Parsell *et al.*, 1995).

Although H2RLX is the predominant form of relaxin and endogenous H1RLX synthesis has yet to be positively confirmed in any tissue, the H1RLX and H2RLX peptides had similar affinities. H1RLX (B2-29) and H2RLX (B33) displayed similar affinities to H1RLX (B2-33) and H2RLX (B29), revealing no further insights into the specific role of the B chain C terminal end 4 amino acids. The C terminus of the B chain varies considerably with species (Figure 1.2), with the length ranging from B28 to B40. There appears to be no direct correlation between B chain C terminus length and affinity for the receptor as H3RLX and RatRLX, with B chain C terminal lengths of B28 and B32 respectively, showed lower affinity for the receptor than H2RLX (B29 & B33) while PRLX, with a B chain of B2-33, displayed similar

Chapter 3: structure-activity relationships of relaxin analogues at the human relaxin receptor affinity to H2RLX (B29) (Table 3.2). Fro-PRLX had the same affinity and efficacy as PRLX (Tan *et al.*, 1998), as did marmoset monkey pro-RLX (Zarreh-Hoshyari-Khah *et al.*, 2001) indicating that the C-peptide did not interfere with interaction with the receptor, again suggesting that the extended length of the C-terminus has little influence on relaxin function.

All but one of the relaxin analogues investigated in this study were able to displace 100% of  $[^{33}P]$ -H2RLX binding, with the H3RLX competition curve appearing to plateau at 50% of radioligand displacement. This raises an interesting issue as to the interaction of the human relaxin analogues with the relaxin receptor. While all saturation studies have indicated a single binding site, the possibility of two binding sites maintaining the same affinity for the relaxin analogues should not be dismissed. The partial displacement of  $[^{33}P]$ -H2RLX binding by H3RLX may be the result of H3RLX interacting with only one of the binding sites. An alternative explanation could be a self interaction between H3RLX peptides at high concentrations that may induce structural changes to the peptide, interfering with binding at the relaxin receptor.

The rank order of affinity of the relaxin analogues differed between the rat and human receptors. H2RLX (B29) had a higher affinity than H2RLX (B33) for the rat relaxin receptor (Tan *et al.*, 2001a), in contrast to the affinity order determined in this study. PRLX displayed similar affinity to H2RLX (B33), while RatRLX had less affinity than the other relaxin analogues at the rat relaxin receptor. The ability of these relaxin analogues to stimulate the rat receptor was demonstrated by H2RLX (B29 & B33) and H1RLX (B33) displaying similar efficacies to PRLX in the rat isolated atrial bioassay. H1RLX (B2-29) displayed significantly lower efficacy while RatRLX was the least bioactive relaxin analogue (Tan *et al.*, 1998). At the human relaxin receptor, cAMP accumulation assays using THP-1 cells reported an efficacy order of H2RLX > H1RLX > H3RLX (Bathgate *et al.*, 2002). There is no current functional data concerning the activity of PRLX and RatRLX at the human relaxin receptor in THP-1 cells.

Chapter 3: structure-activity relationships of relaxin analogues at the human relaxin receptor Reducing the length of the B chain from the C-terminus has a clear negative influence on relaxin affinity with H1RLX (B2-26) and H1RLX (B2-24) displaying progressively lower affinity for the receptor. These results confirm a previous study showing reduced efficacy in the rat atrial bioassay of H1RLX (B2-26) compared to H1RLX (B2-29), while H1RLX (B2-24) had no relaxin-like activity (Tan *et al.*, 1998). Reduction of amino acids up to B24 exposes the cysteine (B23) that forms one of the interchain disulphide bonds, thus causing disruption to the structural integrity of the peptide. Removal of amino acids from the B chain C-terminus of PRLX also shows that removal to B25 has little effect on the ability of the PRLX peptide to induce relaxin-like activity, with further removal of amino acids abolishing all bioactivity (Tregear *et al.*, 1983). Since only limited amounts of H1RLX were synthesised, functional studies using THP-1 cells could not be performed.

While the extended length of the B chain has little influence on the binding capabilities of the relaxin peptide, the two highly conserved arginine residues at B13 and B17 are crucial for relaxin function (Büllesbach et al., 1992). The same two arginine residues were identified in INSL3 and although these were displaced 4 amino acids towards the C-terminus, this peptide displayed very little affinity for the relaxin receptor. Modification of INSL3 so that the arginine residues resided in the relaxin binding motif position produced INSL3-D, increasing the relaxin-like interaction. However the affinity was still 100 fold less than H2RLX, indicating that factors other than the two arginines were necessary for full relaxin-like activity. A third residue critical for the relaxin binding motif, has been identified as isoleucine (B20) (Büllesbach et al., 2000), but this does not explain the difference in relaxinlike affinity since INSL3 contains a compatible residue, valine B20. Functional data examining the ability of the INSL3 analogues to stimulate cAMP production in THP-1 cells showed that although the inclusion of the partial and full human relaxin binding motif in analogues INSL3-A & -D respectively (Figure 3.1) increased relaxin-like cAMP production compared to native sheep INSL3 (Claasz et al., 2002) it was again 100 fold less than H2RLX. Removal of the extra INSL3 arginine in position B21 to expose the valine (B20) in INSL3-E (Figure 3.1) failed to improve Chapter 3: structure-activity relationships of relaxin analogues at the human relaxin receptor relaxin-like potency, indicating that the third arginine did not hinder relaxin-like interaction with the receptor.

The A chain of relaxin has been examined in relation to its influence in relaxin-like activity and appears to be of little significance other than maintaining peptide structure (see Section 1.2.4.1). However, not all areas important for relaxin function have been identified and these do not appear to be solely contained in the B chain. The A and B chain relaxin/INSL3 chimeras would be an invaluable tool for investigation of the roles of each chain in relaxin-like activity.

In conclusion, this study demonstrated distinct differences in the interaction between relaxin analogues and the rat and human relaxin receptors. H1 and H2RLX peptides appear to interact with the receptor in a similar manner, displacing [<sup>33</sup>P]-H2RLX binding with high affinity, while H3RLX may interact slightly differently as it displayed slightly less affinity and could only displace approximately 50% of radioligand binding, possibly indicating the presence of multiple binding sites. Extension of the relaxin B chain has little influence on relaxin function whereas shortening the B chain to position B24 decreases relaxin binding. The relaxin binding motifs of Arg(B13), Arg(B17) and Ile(B20) are not the only determining factors of relaxin activity as insertion of this motif into the closely related INSL3 peptide had only limited effects on relaxin-like binding to the relaxin receptor. Further investigations using H2RLX/INSL3 chimeras may reveal novel areas important for relaxin function.

# **Chapter Four**

# **Production of recombinant relaxin**

いろもうと言語を

#### 4.1 Introduction

Relaxin was first isolated from sow corpora lutea (Fevold *et al.*, 1930). The peptide is present in relatively low concentrations and yields of 30-35 mg of purified PRLX peptide per kg ovary tissue were reported (Fevold *et al.*, 1930; Sherwood *et al.*, 1974). Extraction from the ovary, corpora lutea or placenta led to the successful identification and characterisation of relaxin from a variety of species, including rat, shark, whale, horse, dog and rabbit (Sherwood, 1994). Extraction from tissue could not be used to obtain all species' homologues of relaxin since the collection of large amounts of human tissues for relaxin extraction was not possible. Therefore a synthetic method of providing human relaxin peptides was developed.

The successful sequencing and subsequent synthetic production of a biologically active relaxin peptide was reported in the early 1980's. From the predetermined prorelaxin peptide sequences of rat and porcine relaxin, a highly conserved area within the connecting C-peptide code was identified and used to identify and isolate the human relaxin gene (Hudson et al., 1983). Following comparison of the human and porcine relaxin sequence (Sherwood et al., 1974) and using the reported similarity between the relaxin and insulin structures (Bedarkar et al., 1977), chemical synthesis of human relaxin produced a biologically active human relaxin peptide that dosedependently inhibited rat uterine contractility (Hudson et al., 1983). The successful chemical synthesis of human relaxin provided methods that allowed synthesis of otherwise unobtainable relaxin peptides and also the development of relaxin analogues to investigate the structure-activity relationships of relaxin. Studies with relaxin analogues have shown that the relaxin A chain is necessary for maintaining relaxin peptide structure while the B chain contains areas important for the interaction of relaxin with its receptor (Section 1.2.4). The pre-processed form of relaxin, pro-relaxin, retains full relaxin affinity and efficacy (Tan et al., 1998; Zarreh-Hoshyari-Khah et al., 2001) in contrast to insulin which only becomes active following cleavage of the connecting C-peptide (Peavy et al., 1984; Yu et al., 1973).

Chapter 4: Production of recombinant relaxin

However, the synthetic production of relaxin peptides is not without problems, since low yields of peptide have been reported due to substantial loss during the combination of A and B chains (Tregear *et al.*, 1995). The complexity and variability of the individual relaxin chain sequences necessitates modifications to the synthetic conditions for each relaxin peptide (Tregear *et al.*, 2001), leading to increased cost and time to produce each relaxin analogue. These limitations, and in particular the small yield of peptides, and the increasing demand for the large scale production of human relaxin required for clinical trials, led to the development of recombinant techniques for the production of relaxin peptides.

Early investigations into recombinant methods for production of relaxin peptides involved a synthetically constructed DNA template of pro-PRLX B chain inserted into a carrier plasmid with the remaining part formed by RT-PCR and inserted with specific restriction sites (Stewart *et al.*, 1983). Analysis of the expressed peptides from an *E.coli* culture on a polyacrylamide gel showed a protein of the expected size that was identified as pro-PRLX with anti-relaxin antibodies, although further characterisation of this protein was not reported. Some years later, the successful production of recombinant pro-PRLX was reported by the insertion of the pro-PRLX DNA template into an *E.coli* expression vector, with the pro-PRLX peptide purified from an *E.coli* culture and manually refolded into the correct relaxin peptide structure (Reddy *et al.*, 1992). The recombinant pro-PRLX could be partially digested with dilute trypsin to cleave the C-peptide and form PRLX, and both forms of the recombinant peptide were shown to be biologically active in the bioassay examining the increase in cAMP accumulation in human uterine endometrial cells.

An early attempt to produce recombinant human relaxin involved the insertion of a human relaxin DNA construct with a modified connecting C-peptide into a yeast expression plasmid (Yang *et al.*, 1993). The extracted relaxin pro-peptide was partially purified and found to be biologically active, although it displayed an efficacy some 20 fold less than synthesised H2RLX in the mouse interpubic ligament bioassay. As a part of a program to develop human relaxin for therapeutic use, Genentech Inc. (CA, USA) were successful in producing a recombinant human pro-relaxin protein containing a unique "mini-C" connecting peptide between the A

Chapter 4: Production of recombinant relaxin

and B chains of the protein (Vandlen *et al.*, 1995), a method previously employed successfully in the production of mini-C pro-insulin (Wetzel *et al.*, 1981). The shortened connecting peptide ensured efficient synthesis and extraction from the *E.coli* culture with the pro-relaxin peptide then folded and processed into the H2RLX peptide. This form of recombinant H2RLX was found to maintain full H2RLX efficacy and affinity in THP-1 cell based assays (Parsell *et al.*, 1996).

A common problem for all recombinant methods using bacteria for synthesis is that the resulting relaxin peptide has to be manually folded following extraction as the bacterial cells do not have the ability to fold a mammalian peptide into its correct structure. Recombinant pro-PRLX synthesised in a CHO cell line was biologically active following extraction without the need for manual refolding (Vu *et al.*, 1993). Although the pro-PRLX was reported to have slightly less efficacy than native PRLX in producing a cAMP response in human endometrial cells, the comparison was made between different studies completed 3 years apart (Fei *et al.*, 1990; Vu *et al.*, 1993) and the concentration of the recombinant pro-PRLX was calculated in dilutions of conditioned media.

The aim of this study was to develop a mammalian recombinant system that would allow efficient production and purification of a recombinantly synthesised and correctly folded pro-H2RLX protein. The recombinant system in conjunction with molecular cloning techniques would be used to investigate novel areas important for relaxin activity within the relaxin peptide.

# 4.2 Methods and Results

The recombinant relaxin protein system was based on a combination of the Sindbis Expression System (Invitrogen, Vic, Aust), the InvitroScript<sup>TM</sup> CAP SP6 *In Vitro* Transcription Kit (Invitrogen, Vic, Aust), the SFV Gene Expression System (Invitrogen, Vic, Aust) and the ProBond<sup>TM</sup> Purification System (Invitrogen, Vic, Aust). Briefly, the H2RLX DNA insert was ligated into the Sindbis Expression system pSinHis plasmid, linearised and used as a template for RNA transcription. The capped RNA transcript containing the relaxin coding region and a polyhistidine (His<sub>6</sub>) tag was transiently transfected into BHK cells, the encoded protein was expressed and obtained by harvesting and lysing the cells (Figure 4.1). The relaxin protein was purified from the cell lysate through a nickel IDA bead column (Scientifix, Vic, Aust) that when activated displays high affinity for polyhistidine proteins, binding the relaxin peptide and removing all excess lysate. The relaxin peptide was cleaved from its His<sub>6</sub> tag with EKMax enzyme (Invitrogen, Vic, Aust) and purified from the nickel column (Figure 2.3)

# Figure 4.1

# Schematic diagram of the methodology to produce recombinant pro-RLX.

The RLX DNA fragment (RLX Peptide) is inserted in the pSinHis plasmid adjacent to the enterokinase cleavage site (EKMax), allowing cleavage of the Anti-Xpress<sup>TM</sup> Epitope and the N-terminal polyhistidine tag ((His)<sub>6</sub>) following purification of the recombinant pro-RLX protein. Apon successful ligation of the RLX DNA into the pSinHis plasmid, the plasmid is linearised at the Xho I restriction enzyme cleavage site (Xho I) and used as a template for the transcription of recombinant RNA. *In vitro* transcription is driven by the SP6 promoter (SP6) while *in vivo* transcription, following transfection into the BHK cells, is initiated by the subgenomic promoter (PSG). The pro-RLX protein is translated and extracted from the BHK cells.

Amp = Ampicillin resistance gene; pMB1 = pMB1 origin.

ń.

÷



# 4.2.1 Ligation of the RLX insert into pSinHis plasmid

The H2RLX insert was produced by PCR using pMAL plasmid as a template (gift from Dr Ross Bathgate, Howard Florey Institute of Experimental Physiology and Medicine, Vic, Aust). The 5' and 3' primers contained a Pml I and Sph I restriction site respectively, to facilitate unidirectional cloning into pSinHis. It was necessary to introduce mutations into the multiple cloning site of pSinHis to create a Pml I restriction site (Figure 4.2)

### 4.2.1.1 Incorporation of Pml I restriction site into pSinHis plasmid

Primers were designed (Table 2.1) and synthesised (Invitrogen, Vic, Aust) to incorporate the Pml I restriction site into the multiple cloning site of the pSinHis plasmid using the QuikChange<sup>TM</sup> Site Directed Mutagenesis Kit (Stratagene, CA, USA). The reaction was performed as described in Section 2.2.1. Mutated pSinHis DNA was precipitated and resuspended in sterilised deionised water before being transformed into *E.coli* DH<sub>5α</sub> cells (Section 2.2.1.1), colonies selected and DNA purified (Section 2.2.1.2). Colonies were screened using a Xho I and Pml I digest and analysed on a 1.3% agarose gel (Section 2.2.1.2). Colonies containing the pSinHis DNA with the Pml I restriction site were identified by release of a 450 bp band (Figure 4.3). Correct pSinHis DNA was amplified as described in Section 2.2.1.3.

#### 4.2.1.2 Preparation of the RLX DNA insert

Primers were designed (Table 4.1) to generate a RLX insert that contained the EK recognition and cleavage sites of the pSinHis plasmid that would be removed with the digestion at the Pml I restriction site (Figure 2.1) and synthesised (Invitrogen, Vic, Aust). The Sph I restriction site was incorporated into the 3' end of the RLX insert. PCR reaction for production of RLX DNA fragment is described in Section 2.2.2. RLX insert was visualised as a 500 bp band on a 1.3% agarose gel (Figure 4.4) and purified from a 1.3% low melting temperature agarose gel (Section 2.2.2.1).

# 4.2.1.3 Ligation of RLX insert into pSinHis plasmid

The altered pSinHis plasmid (5  $\mu$ g) and the relaxin fragment (2  $\mu$ g) were digested separately in a reaction mixture including 1x NE 1 buffer (New England Biolabs, MA, USA), 0.01% BSA, 25 U Sph I enzyme and 100 U Pml I enzyme in a total volume of 100  $\mu$ L (37°C for 2 hours). The digested DNA was loaded onto a 1.3% low melting temperature agarose gel where the relaxin DNA fragment bands (~500 bp) and linearised pSinHis DNA bands (~10 kB) were carefully cut from the gel. DNA was purified using the Wizard PCR Preps DNA Purification System (Promega, WI, USA) as described in Section 2.2.2.1. DNA was eluted with 50  $\mu$ L of nucleasefree water at 70°C for the large plasmid and 60°C for the smaller DNA segment. The purified DNA concentration was calculated (Section 2.5) and was stored at -20°C.

Ligation of the RLX insert into the pSinHis plasmid occurred in a reaction mixture containing 200 ng pSinHis plasmid DNA, 40 ng RLX insert DNA, 1x One-Phor-All buffer, 1 mM ATP and 4 U T4 DNA ligase in a total volume of 50  $\mu$ L. Ligation was incubated overnight at room temperature. Ligation mix was precipitated and resuspended in sterilized deionised water (10  $\mu$ L). Ligated DNA was transformed into DH<sub>5α</sub> competent cells as previously described (Section 2.2.1.1), single colonies were selected and DNA purified using the Wizard Plus Miniprep DNA Purification System (Promega, WI, USA) as previously described (Section 2.2.1.2). DNA was screened using a Pml I and Sph I digest in 30  $\mu$ L reaction and DNA visualised on a 1.3% agarose gel. Unfortunately the ligation was unsuccessful as shown by a failure to produce a ~500 bp RLX insert (Figure 4.5).

# Figure 4.2

# Nucleotide sequence of the pSinHis and RLX insert cloning area.

The nucleotide sequence begins with the start codon (ATG) at the 5' end and includes the polyhistidine tag (orange), the Pml I restriction site (blue), the EKMax enzyme recognition site (green) followed by the start of the RLX B chain nucleotide sequence (purple). At the conclusion of the RLX A chain nucleotide sequence is the Sph I restriction site (pink) used in the insertion of the RLX fragment into pSinHis.

--- denotes that the nucleotide sequence continues.

Chapter 4: Production of recombinant relaxin

5' -- ATG GGG GGT TCT CAT CAT CAT CAT CAT CAT GGT ATG GCT AGC ATG ACT GGT GGA CAG CAC GTG GGT CGG GAT CTG TAC GAC GAT GAC GAT AAG GAC TCA TGG ATG GAG GAA GTT ATT AAA ----- AGA TCT CTT GCT AGA TTT TGC TGA AGC TTA GCA TGC CAC TGG CCG --- 3'

# Figure 4.3

Gel electrophoresis of the pSinHis plasmid mutated to include a Pml I restriction site.

Digestion of colonies containing the Pml I restriction site mutation with Xho I and Pml I restriction enzymes, resulted in the formation of a 455bp band as measured against a 100bp molecular weight marker (Lane 1). The positive samples included samples 1 and 3-6 (Lanes 2, 4-7 respectively) while sample 2 (Lane 3) failed to show a Pml I restriction site mutation.



#### Table 4.1

# The oligonucleotide primers designed for the generation of the RLX insert.

The RLX insert forward oligonucleotide primer included the Pml I restriction site (blue), the EKMax enzyme recognition site (green) and the start of the RLX B chain nucleotide sequence (purple). The RLX insert reverse oligonucleotide primer included the complementary sequence to the Sph I restriction site (pink) and the end of the RLX A chain nucleotide sequence (purple). Chapter 4: Production of recombinant relaxin

Sequence  $(5' \Rightarrow 3')$ 

Primer Name

RLX insert for

RLX insert rev

# CAG CAC GTG GGT CGG GAT CTG TAC GAC GAT GAC GAT AAG GAC TCA TGG ATG GAG GAA ()

# GTG GCA TGC TAA GCT TCA GCA AAA TCT AGC

# Figure 4.4

# Gel electrophoresis of the PCR generated RLX insert.

The RLX construct generated by PCR from the original plasmid, pMAL plasmid, resulted in a 500bp band (Lane 2) as measured against a 100bp molecular weight marker (Lane 1)

**1** 

. 1/ 52

# Chapter 4: Production of recombinant relaxin

 $\overline{\mathbf{\nabla}}$ 

4



# Figure 4.5

Gel electrophoresis of the pSinHis plasmid following the ligation reaction to insert the RLX fragment.

Digestion screen of selected colonies (Lanes 1-3, 5 & 6) with Pml I and Sph I restriction enzymes. None of the colonies screened contained the ligated pSinHis-RLX product that would have been represented by a 500bp band. The molecular weight marker is shown in Lane 4. Lane

890bp

500bp

. .



# 4.2.1.4 Activity of Pml I and Sph I enzymes

Due to the continued failure to demonstrate successful ligation, the restriction enzymes used in the digestion reactions were checked for enzyme activity. A digest incorporating the Sac I restriction enzyme with the Pml I and Sph I enzyme separately were designed, with successful digestion confirmed visually by bands of 2245 bp and 2190 bp respectively. The digest reaction included 1mg pSinHis DNA, 1x NE 1 buffer (New England Biolabs, MA, USA), 0.01% BSA, 10 U Sac I enzyme and either 20 U Pml I or 5 U Sph I in a total volume of 30  $\mu$ L (37°C for 2 hours). The products of the digestion reaction were run on a 1.3% agarose gel (Figure 4.6) and demonstrated that both the Pml I and Sph I enzymes were working and produced the expected bands. The Pml I restriction enzyme however appeared to have lost some activity since three bands were observed, representing the expected DNA band, the fully digested pSinHis plasmid and the partially digested pSinHis plasmid.

# 4,2.1.5 Hybridisation screening for the RLX insert

Due to the observed decrease in Pml I enzyme activity, it was hypothesised that successful ligation may have occurred but at low efficiency. Screening large numbers of colonies by miniprep DNA purification and digestion would be time consuming and expensive, therefore a quick and efficient screen of the agar plate colonies was achieved using a [<sup>32</sup>P]-RLX hybridisation probe.

#### 4.2.1.5.1 Preparation of the RLX DNA probe

RLX insert DNA was used for the relaxin specific probe and to ensure no cross recognition of the pSinHis plasmid, the RLX insert DNA was digested with Not I enzyme to remove the pSinHis sequence at the 5' prime end of the DNA fragment. Digestion of RLX insert included 20  $\mu$ g RLX insert DNA, 0.01% BSA, Not I enzyme buffer, 0.2 U Not I enzyme in a total volume of 70  $\mu$ L (overnight incubation at 37°C). DNA was isolated on a 1.3% low melting temperature gel and purified as described in Section 2.2.2.1.

はなまたは一般ない。 ないたいで、 のため、 のたの、 のため、 のため、 のため、 のたの、 のため、 のため、 のたの、 のたの、 のため、 のた
RLX DNA was labelled with  $[\alpha^{32}P]$ -ATP using Ready-to-go DNA labelling beads (Amersham Pharmacia Biotech, Uppsala, Sweden). 50 ng RLX DNA was denatured at 95°C for 3 minutes, placed on ice for 2 minutes and DNA pelleted. The mixture of pelleted DNA and  $[\alpha^{32}P]$ -ATP in a total volume of 50 µL was added to the DNA labelling beads and mixed thoroughly. Reaction was incubated at 37°C for 30 minutes and stored at -20°C. RLX probe was purified using a Sephadex G50 column. The column was equilibrated with 3 mL Tris/EDTA solution (10 mM Tris-HCl, pH 7.5, 1 mM EDTA) prior to addition of the probe labelling reaction. RLX probe was washed through with 400 µL Tris/EDTA solution with the flow through collected at 3 drops/tube and counted with a hand-held  $\beta$ -counter. Tris/EDTA solution was continually added until the eluant no longer contained radioactivity. Samples containing the first peak of radioactivity were pooled and stored at -20°C.

### 4.2.1.5.2 Hybridisation screen

The ligated pSinHis/RLX DNA was transformed into  $DH_{5\alpha}$  cells and grown on agar plates as previously stated in Section 2.2.1.1. Nitrocellulose filters cut to fit the agar plates were numbered, sparingly moistened with distilled water, carefully placed on top of the agar colonies and marked for orientation. The colony lifts were then placed, colony side up, onto 3MM paper soaked in denaturing solution (0.5 M NaOH, 1.5 M NaCl) for 3 minutes. The colony lifts were transferred onto 3MM paper soaked in neutralizing solution (0.5 M Tris-HCl, 1.5 M NaCl, pH 7.5) for 3 minutes, followed by 3 minutes on 2x SSC soaked 3MM paper and then air dried for 5 minutes. The colony lifts were baked in an 80°C oven for 10 minutes before being placed in a clean Petri dish.

The pre-hybridisation solution included Denhardts solution (50x; 5 mL), 20x SSC (12.5 mL), 10% SDS (2.5 mL), 100 mM ATP (50  $\mu$ L), 10 mg/mL salmon sperm DNA (500  $\mu$ L; denatured at 95°C, 5 minutes) in a total volume of 50 mL. The pre-hybridisation solution was placed into a Petri dish, with each individual piece of nitrocellulose being added and soaked prior to the addition of the next piece. A Petri

Chapter 4: Production of recombinant relaxin

dish containing the filters and solution was wrapped in polyethylene film and incubated at 42°C for 4 hours. Filters were carefully removed from the Petri dish, excess pre-hybridisation solution removed and the  $[^{32}P]$ -RLX DNA probe added. The filters were replaced into the dishes individually, gently mixed with the probe solution (pre-boiled, 5 minutes) and incubated at 42°C overnight.

The nitrocellulose filters were washed sequentially with 2x SSC, 0.1% SDS at room temperature, at 30°C for 30 minutes, at 37°C for 5 minutes before drying at room temperature for 30 minutes. Blots were mounted onto cardboard backing, wrapped in polyethylene and exposed to film (Hyperfilm, Amersham, Sweden) overnight with orientation markers. Film was developed in D-19 developer (5 minutes; Schreiber Photographics, Vic, Aust), stop solution (1 minute; Schreiber Photographics), fixative (10 minutes; Schreiber Photographics), gently rinsed in water (10 minutes) and orientated via the markers. The film was then aligned with the original agar plates, positive colonies identified and selected for screening by Miniprep preparations (Section 2.3.1.2). All colonies were screened using a Not I digest, including I µg ligated DNA, 0.01% BSA, Not I enzyme buffer, 0.01% Triton X-100, 10 U Not I enzyme in a total volume of 50 µL. Not I digestion of pSinHis plasmid containing the RLX insert would result in a fragment of 845 bp with one Not I restriction site contained outside the RLX insert area of the plasmid and the other contained within the RLX insert. None of the positive colonies contained the RLX insert (Figure 4.7).

Gel electrophoresis of the pSinHis plasmid digested by Sph I/Sac I and Pml I/Sac I restriction enzymes.

Samples of pSinHis plasmid were digested with a combination of Sph I/Sac I restriction enzymes (Lanes 2 & 3) and Pml I/Sac I restriction enzymes (Lanes 5 & 6) resulting in the formation of a 2190bp and 2245bp band respectively as measured against the 100bp (Lane 1) or 1kb (Lane 4) molecular weight markers. The two pSinHis plasmid bands shown in Lanes 5 & 6 represent uncut (higher band) and cut (lower band) plasmid, indicating partial digestion of the samples.



# Chapter 4: Production of recombinant relaxin

# Gel electrophoresis of the colonies selected from hybridisation probe.

Positive colonies as indicated by the hybridisation probe (Lanes 2-7 & 9-14) were screen by digestion with Not I restriction enzyme. Colonies containing the ligated pSinHis-RLX plasmid were identified by a 845bp band, which was not shown by any of the samples. The 100bp molecular weight marker is shown in Lanes 1 & 8.





### 4.2.1.6 Regeneration of the RLX insert

The primers for the generation of the RLX insert product by PCR were redesigned (Table 2.1) to lengthen the Pml I and Sph I restriction site cleavage areas to reduce the possibility that the enzymes failed to recognise and cleave the short ends of the RLX insert. The RLX insert DNA was generated and purified as described in Section 2.2.2. The pSinHis plasmid and RLX insert DNA were digested with Pml I and Sph I, isolated, purified and ligated as described in Section 4.2.1.3. Ligated DNA was transformed in DH<sub>5 $\alpha$ </sub> cells and produced very small numbers of colonies on agar plates. All colonies were screened using a Not I digest as described above. The ligation failed with no DNA fragment shown when the digested DNA was visualised on a 1.3% agarose gel (data not shown).

### 4.2.1.7 Blunt-end ligation reaction

Digestion of DNA by the Pml I restriction enzyme results in the formation of a "blunt-end", where the DNA is cleaved cleanly with no overhalling DNA sequence. As the blunt-end could have interfered with successful ligation of the RLX DNA fragment into the pSinHis plasmid, ligation conditions were altered to accommodate a blunt-end ligation. The reaction mixture included 200 ng pSinHis DNA, 40 ng RLX insert DNA, 50 mM Tris (pH 7.5), 10 mM MgCl<sub>2</sub>, 5 mM DTT, 0.1 mg/mL BSA, 5% PEG, 0.5 mM ATP and 4 U T4 DNA ligase in a total volume of 100  $\mu$ L and was incubated overnight at room temperature. Ligated DNA was precipitated, transformed into DH<sub>5α</sub> cells, colonies isolated and screened using a Not I restriction enzyme digestion as described in Section 4.2.1.5.2. Ligation was unsuccessful with no DNA fragment visualised on a 1.3% agarose gel (data not shown).

### 4.2.1.8 Generation of RLX insert from PCR script

To ensure the correct digestion of the RLX insert DNA by Pml I and Sph I enzymes, the PCR RLX insert product was directly ligated into PCR script plasmid according to the PCR-Script<sup>™</sup> Amp Electroporation-Competent Cell Cloning Kit (Stratagene,

Chapter 4: Production of recombinant relaxin

CA, USA). The ligation reaction included 10 ng pPCR script Amp SK(+) cloning vector, 1x PCR script reaction buffer, 0.5 mM rATP, 170 ng RLX PCR DNA product, 5 U Srf I restriction enzyme, 4 U T4 DNA ligase in a total volume of 10  $\mu$ L. The ligation reaction was gently mixed, incubated at room temperature for 2 hours and the reaction halted by a 10 minute incubation at 65°C. DNA was precipitated and transformed into XLI-Blue MRF electrocompetent cells as described for DH<sub>5α</sub> cells (Section 2.2.1.1) except that the agar plates were pre-treated with X-gal prior to plating out transformed cells. The PCR script plasmid containing the RLX insert produced white colonies due to the disruption of the  $\beta$ -galactosidase gene responsible for the blue colour seen in colonies that do not contain the RLX insert. Selected colonies were screened by Miniprep preparations and DNA isolated (Section 2.2.1.2). DNA was digested by PinI I and Sph I restriction enz mes, visualisation on a 1.3% agarose gel and showed that seven of the ten selected colonies contained the RLX insert (Figure 4.8).

### 4.2.1.8.1 Sequencing of PCR script plasmid containing the RLX insert

All of the PCR script clones containing the RLX insert were sequenced to ensure no sequence anomalies had occurred during the processing of the DNA. The sequencing protocol was as described in Section 2.2.3.1, incorporating the T3 and T7 sequencing primers (Table 2.1). Results showed that PCR script-RLX clones #1, #2, #4 and #7 contained sequence errors while clones #3, #5 and #6 were correct. Clone #5 was chosen for further study. PCR-script/RLX clone #5 (2  $\mu$ g) was digested with Pml I and Sph I restriction enzymes, run on a 1.3% low melting temperature gel and the RLX insert purified as previously described (Section 2.2.2.1). The RLX insert (40 ng) and the digested pSinHis plasmid (200 ng) were set up in a blunt-end ligation reaction mixture (Section 4.2.1.7) and incubated at 14°C for 4 hours. Ligated DNA was precipitated, transformed into DH<sub>5α</sub> cells, DNA purified from selected colonies and screened using the Not I digest as previously described above. Visualisation of digested DNA on a 1.3% agarose gel showed no successful ligation of the RLX insert into pSinHis (data not shown).

### Gel electrophoresis of PCR-script following ligation of RLX construct.

Selected colonies were screened (Lanes 2-6 & 8-12), for the successful ligation of the RLX insert into PCR-script, by digestion with Pml I and Sph I restriction enzymes. Positive colonies were identified by a 550bp band (Lanes 2-6, 8 & 9) as measured against the 100bp molecular weight marker (Lanes 1 & 7).



# 4.2.1.9 Freeze 'n Squeeze purification and ligation of RLX insert into the pSinHis plasmid

Since the DNA fragments supported the correctly Pml I and Sph I digested ends, it was possible that there was inhibition of the ligation reaction. Therefore, instead of purifying the digested RLX insert and pSinHis plasmid DNA from the 1.3% low melting temperature gel with the Wizard PCR Preps System, it was isolated using a 'Freeze 'n Squeeze' phenol/chloroform extraction method. DNA bands were cut from the gel, weighed, cut into small pieces and place in an Eppendorf tube. An equal volume of phenol was added (0.1 g gel = 100  $\mu$ L phenol), the mixture frozen at -70°C for 10 minutes and centrifuged at 4°C for 15 minutes. The aqueous phase was transferred into a clean tube, phenol added (200 µL aqueous phase to 100 µL phenol), mixed thoroughly and centrifuged for 5 minutes at 4°C. 100 µL of chloroform was added, and the mixture centrifuged for 10 minutes at 4°C. The top phase containing the isolated DNA fragments was transferred into a clean tube and kept on ice. DNA was precipitated using 1/10 volume of 3M NaOAc (pH 5.2) and 2.5x volume of absolute EtOH, storage at -20°C for 20 minutes and centrifuged for 25 minutes at 4°C (14,000g). The DNA pellet was washed twice with 70% EtOH (500 µL, 10 minute centrifugation at 4°C) before being air-dried and then redissolved in 20 µL sterilised deionised water.

The pSinHis plasmid (200 ng) and RLX insert (40 ng) DNA were set up in a standard ligation reaction that included the DNA fragments, 1 mM ATP, 1x One-Phor-All buffer and 4 U T4 DNA ligase in a total volume of 20  $\mu$ L. The reaction was incubated at room temperature overnight and inactivated by heating to 65°C for 10 minutes. Ligated DNA was precipitated, transformed into DH<sub>5α</sub> cells, DNA purified from selected colonies, screened using the Not I digest and visualised on a 1.3% agarose gel. One of twenty colonies selected was positive for RLX insert into pSinHis plasmid, clone #2 (Figure 4.9a). To check the positive clone before sequencing, clone #2 was digested with Pml I and Sph I as described above and visualised on a 1.3% agarose gel which showed the correct RLX insert dropout of

Chapter 4: Production of recombinant relaxin

~550 bp (Figure 4.9b). The plasmid was isolated from an *E.coli*  $DH_{5\alpha}$  culture and DNA purified using the Wizard Plus Midiprep DNA Purification System (Promega, WI, USA) as described in Section 2.2.1.3. Clone #2 was sequenced as described in Section 2.2.3.1.

### 4.2.2 RNA transcription of DNA templates

To transiently express the protein in the BHK cell line, the DNA templates were used to produce RNA transcripts. The *in vitro* transcribed capped RNA is transfected into the BHK cells and is read as messenger RNA, allowing production of the protein. RNA transcripts were produced from the pSinHis-RLX plasmid and from blank pSinHis plasmid to use as an internal control for the transfection and protein extraction process. The methods for RNA transcription were as described in Section 2.2.4. Briefly, the pSinHis-RLX and pSinHis plasmid were linearised by Xho I restriction enzyme digestion, isolated on a 1.3% low melting temperature gel and purified by the 'Freeze 'n Squeeze' protocol. The linearised DNA was transcribed (Section 2.2.4), and the quality of the RNA transcripts visualised on a 1% agarose gel (Figure 4.10).

# Gel electrophoresis of the pSinHis plasmid following ligation of RLX construct.

A) Selected colonies were screened (Lanes 2-11) by digestion with Not I restriction enzyme. Colonies containing the successfully ligated pSinHis-RLX plasmid were identified by a 845bp band (Lane 3, yellow circle) as measured against a 100bp molecular weight marker (Lane 1). The presence of pSinHis-RLX plasmid in Clone #2 was confirmed by B) digestion with Pml I and Sph I restriction enzymes (Lane 2), resulting in a 550bp band. The 100bp molecular weight marker is shown in Lane 1.



÷

# Gel electrophoresis of the RNA transcripts.

The quality of the transcribed RNA is represented by the clear band, approximately half the size of the DNA template band. The blank pSinHis plasmid is shown in Lane 1, the pSinHis-RLX plasmid is in Lane 3 while the 100bp molecular weight marker is shown in Lane 2. Lane

800bp





### 4.2.3 RLX protein production and extraction

The pSinHis-RLX and pSinHis plasmid RNA transcripts were transfected into BHK cells plated in Petri dishes as described in Section 2.2.5. The transfected BHK cells were harvested at two time points, 24 and 48 hours following transfection. Protein was extracted as described in Section 2.2.6. Samples from every step were saved for analysis, including all waste and washing steps. Briefly, harvested cells were pelleted, lysed and the protein in the supernatant was separated from the cell debris. The His<sub>6</sub>-RLX protein was purified from the cell lysate supernatant using the ProBond Purification System (Invitrogen, Vic, Aust) as described in Section 2.2.6.2. The His<sub>6</sub> tag attached to the RLX protein binds to the pre-equilibrated nickel bead column while the unwanted proteins and cell debris are washed away. The RLX protein is cleaved from the His<sub>6</sub> tag on the column by the EKMax enzyme provided with the pSinHis system, and collected from the column.

### 4.2.4 Analysis of recombinant protein

A sample of each extraction step was mixed with loading dye and loaded onto a 16% polyacrylamide gel as described in Section 2.7. Following electrophoresis (150V, 2 hours) proteins were visualised by staining the gel. The gel was fixed in Rapid Stain Fixing Solution (25% isopropanol, 10% acetic acid), and gently agitated for 15 minutes. The fixing solution was replaced with Rapid Coomassie Blue Stain Solution (0.06% Coomassie blue G-250, 10% acetic acid) and the gel gently agitated overnight. Excess stain was removed by washing in Destaining Solution II (7% acetic acid, 5% methanol) until the gel background was clear. Since there were few protein bands visible on the gel throughout all extraction steps (data not shown) including the cell debris pellet, it appeared that the protein in each sample was too dilute. Therefore the transfection and protein extraction was repeated using a larger scale preparation.

The process was repeated for the larger scale extraction and following electrophoresis, proteins were visualised by staining the gel with a silver staining

Chapter 4: Production of recombinant relaxin

protocol (Silver Stain Plus kit, Bio-Rad, CA, USA), which is more sensitive than the Rapid Coomassie Blue Stain. The gel was soaked in a Fixative Enhancer Solution (50% v/v reagent grade methanol, 10% v/v reagent grade acetic acid, 10% v/v fixative enhancer concentrate) with gentle agitation for 20 minutes. The gel was rinsed with deionised water twice for 10 minutes with gentle agitation while the staining solution was prepared. Staining solution consisted of 35 mL deionised water being continually stirred while 5 mL Silver Complex Solution, 5 mL Reduction Moderator Solution and 5 mL Image Development Reagent were added. Immediately before use, 50 mL Development Accelerator Solution was added and mixed thoroughly before being poured over the gel. The gel was gently agitated in staining solution for 20 minutes or until the protein bands were clearly visible, when the reaction was stopped by transferring the gel into a 5% v/v acetic acid solution for 15 minutes. The gel was rinsed in deionised water for 5 minutes before being dried in cellophane for analysis (Figure 4.11). Results showed many protein bands throughout all extraction steps with a strong band of the RLX protein size (~17 kD) identified in the sample of the cell debris pellet. There was no visible band of the RLX protein size identified in any of the EKMax enzyme cleavage samples where the RLX protein was expected to be isolated.

Relaxin specific antibodies were obtained (Dr R. Bathgate, Howard Florey Institute of Experimental Physiology and Medicine, Vic, Aust) to analyse the Western gel for the expressed recombinant RLX protein. Samples of the protein were loaded onto a 16% polyacrylamide gel, H2RLX protein was loaded for a positive control, and the gel processed as described in Section 2.7. Following electrophoresis the proteins were electrotransferred onto a nitrocellulose membrane which was incubated with the relaxin-specific primary antibody overnight at 4°C. The HRP-linked secondary antibody was applied to the membrane, followed by exposure to enhanced chemiluminescence (ECL) and the proteins were visualised on Hyper ECL film (Figure 4.12). The relaxin antibody clearly identified the H2RLX positive control and a band of the RLX protein size (~17 kD) in the cell debris pellet, in agreement with the results from the silver stained gel (Figure 4.11). No other protein band of appropriate size was identified in any of the other extraction step samples.

#### Silver stain of polyacrylamide gel showing the purification steps.

Each sample obtained from the purification step was screened for proteins of pro-RLX size, ~17 kDa. The samples from each of the purification steps (Section 2.2.6.2) included; cell debris pellet (Lane 2), raw supernatant from cell lysate (Lane 3), binding buffer washes (Lane 4 & 5), wash buffer washes (Lanes 6 & 7) binding buffer without NaCl wash (Lane 8), EKMax enzyme cleavage (Lane 9), wash buffer washes (Lane 12-14), wash buffer at pH 5.5 washes (Lane 15 & 16) and washes to strip the resin beads (Lane 17 & 18). PPS molecular weight markers are shown in Lanes 1 & 11 while kaleidoscope molecular weight markers are in Lanes 10 & 20. Lane 19 is blank. The pro-RLX protein was expected in Lane 9 and was not present, however a protein of the expected 17 kDa size was shown in the cell debris pellet and also in the raw lysate supernatant, Lanes 2 & 3 respectively.

During the drying procedure the gel cracked as seen through Lanes 1-3, 11 and 19-20.





Western blot showing the purification samples probed with relaxin specific antibodies.

To probe for pro-RLX, the Western blot containing various samples from the purification steps was exposed to a relaxin-specific antibody. The purification steps (Section 2.2.6.2) included; cell debris pellet (Lane 2), raw supernatant from cell lysate (Lane 3), binding buffer wash 1 (Lane 4), EKMax enzyme cleavage (Lane 5), wash buffer wash 1 (Lane 6), and wash 1 to strip the resin beads (Lane 7). Molecular weight markers are shown in Lanes 1 & 8 with Lane 9 remaining a blank lane. Recombinant H2RLX was used as a positive control (Lane 10), however too much protein was loaded and during transfer the protein appeared to bleed into the adjoining blank lane.

The relaxin-specific antibody indicated a RLX peptide in the cell debris pellet (Lane 2) of the approritate size (17 kDa) while also appearing to bind to larger proteins of  $\sim$ 20 kDa and 50 kDa. The pro-RLX peptide was expected in Lane 5 and the antibody appeared to show a possible dimer form of the pro-RLX (~35 kDa).







Expected pro-RLX

peptide - size,

17kDa

### 4.3 Discussion

The aim of this study was to develop a method for the recombinant production of relaxin peptides which would encompass molecular modification of the peptide to allow further study of the structural interaction between relaxin and the relaxin receptor. The pSinHis system was chosen as the base of the recombinant methodology for two reasons: it is a mammalian system which would ensure correct folding of the human relaxin peptide and the addition of a polyhistidine tag into the protein of choice allowed for simple purification.

*E.coli* cultures are commonly used for recombinant protein production as they can express large amounts of protein efficiently to meet the demand placed by industry. Various methods for the recombinant production of insulin and relaxin peptides by *E.coli* cultures have been reported, including the expression of the A and B chains separately before being assembled into the native protein (Schmidt et al., 1999; Vandlen et al., 1995; Williams et al., 1982) and the expression of the full peptide with the A and B chains connected by either the full length connecting C peptide (Reddy et al., 1992; Stewart et al., 1983; Williams et al., 1982) or a unique "mini-C" peptide (Vandlen et al., 1995; Wetzel et al., 1981). The common problem and the most costly step in the production of recombinant proteins using *E.coli* cultures is the need to re-fold the peptides into the correct structure by forming the disulphide bonds following purification from the culture. E.coli bacteria cannot read the mammalian coding for the structural folding and therefore the peptide is expressed in a non-functional form. Further development has demonstrated success for in vitro folding of pro-insulin using a redox system before purification from the E.coli cultures (Winter et al., 2002a; Winter et al., 2002b), increasing the obtainable yield of the peptide.

An advantage of producing pro-RLX over pro-insulin peptides is that unlike proinsulin, pro-RLX does not need conversion into RLX to obtain biological activity (Tan *et al.*, 1998; Vu *et al.*, 1993; Zarreh-Hoshyari-Khah *et al.*, 2001). The ultimate purpose of our recombinant system was to produce a variety of pro-H2RLX peptides

Chapter 4: Production of recombinant relaxin

to examine the principle areas of the peptide that determine its relaxin-like activity. As only small yields of each peptide were needed for a pharmacological study, the strategy was to develop a mammalian recombinant system that allowed correctly folded RLX peptides to be isolated, as has been shown in previous studies (Kizawa et al., 2003; Marriott et al., 1992; Vu et al., 1993). The advantage of using the pSinHis recombinant system was the synthesis of the pro-RLX peptide with a polyhistidine tag attached. The polyhistidine tag displays high affinity for activated metal cation beads, allowing efficient purification of the recombinant protein. Polyhistidine tags have been extensively used for fast and efficient extraction of recombinant proteins and receptors (Chaga et al., 1999; Mukhija et al., 1995; Palma et al., 2001; Piatibratov et al., 2000; Theis et al., 2001), including recombinant human pro-insulin (Mackin, 1999). Although the polyhistidine tags do not appear to influence the activity of the recombinant peptides (Mukhija et al., 1995; Piatibratov et al., 2000; Theis et al., 2001), the polyhistidine tag can be cleaved from the pro-RLX peptide in the pSinHis system by the EKMax enzyme, allowing a pure sample of pro-RLX to be obtained.

The analysis of the purification samples was expected to show the location of the pro-H2RLX peptide in the elutant following cleavage of the protein from the polyhistidine tag bound to the nickel beads. Unfortunately the protein staining techniques failed to identify the pro-H2RLX protein (17 kDa) in the expected extraction sample. Coomassie Blue and silver staining techniques suggested with the relaxin antibody detection that the recombinant relaxin protein at the molecular size of ~17 kDa was located within the cell debris pelleted upon removal of cell supernatant. The relaxin antibody however also identified a weak protein in the EK max cleavage sample at the molecular size of ~35 kDa. Previous relaxin recombinant studies have reported the formation of pro-RLX homodimers (Vu *et al.*, 1993) which could account for this 35 kDa protein detected by the relaxin antibodies. Although the larger form of the relaxin protein was positively identified in the correct extraction sample, the majority of the recombinant relaxin synthesised appeared to be in its monomeric form in the cell debris extraction sample.

Chapter 4: Production of recombinant relaxin

A successful extraction of pro-RLX peptides from recombinant cultures can be achieved by the secretion of the protein into the culture media. Previous studies have shown secretion of pro-H2RLX with a shortened C-peptide from yeast cultures (Yang *et al.*, 1993), pro-PRLX from a CHO cell culture (Vu *et al.*, 1993), marmoset monkey pro-RLX from an insert cell culture (Zarreh-Hoshyari-Khah *et al.*, 2001) and H3RLX from a mouse adrenocorticotrophic hormone secreting cell line (AtT20) (Kizawa *et al.*, 2003). The addition of a signal peptide to the recombinant protein sequence encoded for the secretion the pro-H2RLX peptide into the cell culture media may prevent the pro-H2RLX peptide being retained within the cell debris pellet upon extraction.

In conclusion, the mammalian recombinant system developed to produce pro-H2RLX peptides appears to have synthesised the recombinant protein, although it seems to be retained within the cell debris pellet. Further modification of the system to allow secretion of the recombinant peptide into the culture media may allow successful isolation of pro-RLX proteins.

# **Chapter Five**

# Characterisation of the putative relaxin receptor, LGR7

### 5.1 Introduction

Although relaxin was one of the first reproductive hormones to be identified, the isolation and characterisation of the relaxin receptor continued to elude researchers for decades. Since relaxin is related to insulin and other members of 🤊 insulingene superfamily that signal through tyrosine kinase receptors, it initially proposed that the relaxin receptor would also be a tyrosine kinase ed to but distinctly different from the insulin receptor (Büllesbach et al., 1995b). The relaxin receptor was reported to be comprised of two components, a 220 and a 36 kDa fragment, similar to the insulin receptor (Osheroff et al., 1995; Parsell et al., 1996). Binding of [<sup>32</sup>P]-relaxin to human uterine cells was unaffected by GTP, indicating that the relaxin receptor was unlikely to be a G-protein coupled receptor (Osheroff et al., 1995). Stimulation of relaxin receptors on lower human uterine fibroblasts caused an increase in tyrosine phosphorylation of a 220 kDa protein while failing to produce a cAMP response (Palejwala et al., 1998), providing further evidence suggesting a tyrosine kinase receptor rather than a G-protein coupled receptor signalling through G<sub>S</sub>. However, relaxin caused activation of adenylate cyclase in human myometrium membrane preparations. This response was inhibited by Tyrphostin 47, a tyrosine kinase inhibitor, and enhanced by GTP-Gpp[NH]p, a non hydrolysable guanine nucleotide analogue (Kuznetsova et al., 1999). Insulin and IGF-1 stimulated adenylate cyclase activity was also regulated similarly to the relaxin response, further strengthening the idea that the relaxin and insulin receptor were similar and providing evidence of a link between a tyrosine kinase receptor and the adenylate cyclase signalling system.

However, and surprisingly given the indications noted above, the relaxin receptor was recently identified as an orphan leucine-rich repeat-containing G-protein coupled receptor (LGR). PRLX was found to interact and stimulate a cAMP response through an orphan LGR receptor, LGR7 (Hsu *et al.*, 2002), with similar potency to that observed for H2RLX acting at the relaxin receptor in THP-1 cells (Parsell *et al.*, 1996). Interestingly, relaxin also stimulated another orphan LGR receptor, LGR8 (Hsu *et al.*, 2002), but had lower potency than at the LGR7 receptor.

Chapter 5: Characterisation of the putative relaxin receptor, LGR7

It was hypothesised and later confirmed that the LGR8 receptor was the INSL3 receptor rather than a second relaxin receptor (Bogatcheva *et al.*, 2003; Kumagai *et al.*, 2002), and LGR8 knockout mice displayed the same cryptorchidism phenotype as INSL3 knockout mice (Hsu *et al.*, 2002). The INSL3 and LGR8 link has also been demonstrated for human cryptorchidism (Ferlin *et al.*, 2003; Roh *et al.*, 2003).

Although the relaxin receptor did not resemble other members of the insulin gene superfamily, it does resemble other reproductive hormone receptors such as luteinising (LH) and follicle stimulating hormone (FSH) (Hsu *et al.*, 2000; lvell, 2002). LGR receptors are large G-protein coupled receptors containing the typical seven transmembrane domains with a unique large ectodomain that includes the leucire-rich repeat (LRR) regions (Figure 1.4) (Hsu, 2003; Kajava *et al.*, 1995). The LGR family is split into three subfamilies of receptors according to their ectodomain structure (Figure 1.5); subfamily A LGRs includes the LH, FSH and the thyroid-stimulating hormone (TSH) receptors; subfamily B includes the orphan LGR4, LGR5 & LGR6 receptors; and subfamily C consists of the putative relaxin receptor (LGR7) and the INSL3 receptor (LGR8) (Hsu, 2003; Hsu *et al.*, 2000).

There is close similarity between the expression of LGR7 receptor mRNA and expression of relaxin binding sites or responses to relaxin in rat tissues. LGR7 mRNA occurs in various reproductive organs, including ovary, uterus and testis, is present in heart and brain tissue, and in low levels in the kidney, adrenal gland and small intestine (Hsu *et al.*, 2000). Relaxin binding sites are found throughout the rat reproductive system, including cervix (Kuenzi *et al.*, 1995; Weiss *et al.*, 1982) and uterus (Osheroff *et al.*, 1990; Weiss *et al.*, 1982) as well as in areas of the brain and heart (Osheroff *et al.*, 1992; Osheroff *et al.*, 1991; Tan *et al.*, 1999). Whilst no relaxin receptor has been demonstrated in the kidney, relaxin treatment has been shown to reverse the fibrotic symptoms induced by age-related fibrosis in relaxin knockout mice (Samuel *et al.*, 2003c; Samuel *et al.*, 2004).

ないないので、「ないない」となったので、「ないない」を見ていた。

The aim of this study was to examine the pharmacological properties of the putative relaxin receptor, LGR7, using a binding assay with the specific relaxin radioligand [<sup>33</sup>P]-HRLX and a cAMP accumulation assay.

J

### 5.2 Methods

LGR7 receptors were transiently expressed in CHO K1 cells using LGR7 DNA in a pcDNA3.1-Zeo expression vector (Invitrogen, Vic, Aust). The relaxin analogues used to examine relaxin interaction with the LGR7 receptor in binding and cAMP assays included the specific relaxin radioligand, [<sup>33</sup>P]-H2RLX, and the competitive relaxin analogues H2RLX, H3RLX and PRLX. RatINSL3 and human insulin were included as negative controls.

### 5.2.1 Cell culture

CHO K1 cells were maintained in 50:50 Dulbecco's Modified Eagle's Medium (DMEM): Ham's F-12 media containing glutamine (2 mM), penicillin (100 units/mL), streptomycin (100  $\mu$ g/mL) and heat inactivated FBS (10% v/v). CHO K1 cells were maintained below ~80% confluence and split as described in Section 2.1.1.2.

### 5.2.1.1 Transient transfection of CHO K1 cells

CHO K1 cells were transiently transfected with the LGR7 receptor DNA construct in the pcDNA3.1-Zeo expression vector (Invitrogen, Vic, Aust) as described in Section 2.1.4. Briefly, 18 hours before transfection, cells were harvested (Section 2.1.1.2), counted (Section 2.1.3) and plated at 8 x  $10^5$  cells/cm<sup>2</sup> in normal cell media. The transfection reaction was incubated for 30 minutes and cells incubated with the transfection mix at 37°C for 4 hours. Serum enriched medium was added and the incubation continued at 37°C for a further 20 hours. Transfection mix medium was replaced with normal media and the incubation continued at 37°C for 24 hours. Transfected cells were used 48 hours after transfection.

### 5.2.2 96 well binding assay

The CHO K1 cell binding assay was set up as described in Section 2.3.3.2. The total volume used was 100  $\mu$ L/well, comprising 20  $\mu$ L of 100 pM [<sup>33</sup>P]-H2RLX ± 10  $\mu$ L unlabelled relaxin analogue and 50  $\mu$ L cell suspension, made up to 100  $\mu$ L with binding buffer. All experiments were performed in duplicate. Following a 90 minute incubation at 25°C, cells were harvested onto GF/C filter plates (Skudtek, Vic, Aust) pre-soaked overnight (0.5% PEI, 4°C). Filter plates were dried at 42°C, Microscint O (Perkin-Elmer, Vic, Aust) added (30  $\mu$ L/well) and equilibrated overnight. Filters were counted on a Topcount scintillation counter (Perkin-Elmer, Vic, Aust) and data was analysed with GraphPad PRISM (Section 2.8).

# 5.2.2.1 [<sup>33</sup>P]-H2RLX (B33) preparation

H2RLX (B33) was labelled with  $[\gamma^{-33}P]$ -ATP using a procedure based on Osheroff *et al* (1990) as described in Section 2.3.1.

### 5.2.2.2 Saturation binding protocol

Saturation binding studies were used to determine the affinity (pK<sub>D</sub>) of [<sup>33</sup>P]-H2RLX and the maximum number of binding sites (B<sub>max</sub>) of the LGR7 receptor transiently expressed in CHO K1 cells. Saturation binding studies were performed at a concentration range of 25 pM – 1 nM. Specific binding was calculated by the total binding – non-specific binding (10  $\mu$ M H2RLX) for each [<sup>33</sup>P]-H2RLX concentration. Protein concentrations were determined as described in Section 2.6 and the data was plotted as specific binding (fmol/mg protein) against concentration of radioligand used (pM) (Section 2.8).

### 5.2.2.3 Competition binding protocol

Competition binding assays were used to calculate the affinity (pK<sub>i</sub>) of various relaxin analogues for the putative relaxin receptor, LGR7. Competition binding was performed using [ $^{33}$ P]-H2RLX at a concentration of 100 pM. To minimise the amount of relaxin analogues used, which were in limited supply, a restricted range of concentrations was used. The concentration range used for H2, H3 and PRLX analogues was 100 pM – 100 nM and a single concentration of RatINSL3 (100 nM) and human insulin (1  $\mu$ M) were also examined. Non-specific binding was determined using I  $\mu$ M H2RLX. Results were expressed as a percentage of maximal specific binding against concentration of the relaxin or relaxin-like analogue used (M) and pK<sub>i</sub> values were calculated (Section 2.8).

### 5.2.3 cAMP accumulation assay

The ability of the relaxin analogues to stimulate a functional response through the LGR7 receptor was examined using the Perkin-Elmer Flashplate cAMP assay (Section 2.4.1). Briefly, cells were plated into a 96-well tissue culture plate 16 hours before each experiment at 10,000 cells/well. Cells were stimulated in incubation media with 10  $\mu$ L relaxin analogue (10 pM – 30 nM), forskolin (1  $\mu$ M) for the reference response or diluting vehicle (PBS) for basal levels for 30 minutes at 37°C. Stimulation was halted by the addition of 0.5M HCl (25  $\mu$ L/well) and the plate stored at -20°C. cAMP samples were analysed using a NEN FlashPlate RIA (Perkin-Elmer, Vic, Aust) according to the manufacturers instructions (Section 2.4.1). Radioactivity was measured on a Topcount scintillation counter (Perkin-Elmer, Vic, Aust) and data analysed against a cAMP standard curve using GraphPad PRISM (Section 2.8).

-

### 5.2.4 Analysis

Results are expressed as mean  $\pm$  S.E.M. of n experiments (Section 2.8). Saturation binding data was analysed with GraphPad PRISM. Radioligand affinity (pK<sub>D</sub>) and LGR7 receptor expression (B<sub>max</sub>) values were calculated and expressed in pM and fmol/mg protein respectively. Competition binding data was analysed with GraphPad PRISM and the relaxin analogue affinity (pK<sub>i</sub>) values were calculated using the Cheng & Prusoff (1973) equation. cAMP accumulation data was analysed with GraphPad PRISM, calculated as pM cAMP per well, expressed as a percentage of maximal cell response as determined by forskolin stimulation and analogue potency (pEC<sub>50</sub>) values calculated. Statistical analysis was completed using GraphPad PRISM statistical programs with competition binding and functional response curves analysed using a two-way ANOVA. pK<sub>i</sub> values were analysed using a Student's t-test. P values of less than or equal to 0.05 were considered significant.

### 5.2.5 Drugs and reagents

All drugs were used as described in Section 2.9. Stock solutions of H2, H3, PRLX analogues and RatINSL3 were prepared in 0.1% aqueous trifluoroacetic acid (TFA) at a concentration of 10  $\mu$ M. Human insulin (Humulin® R) was supplied by Aza Research Pty. Ltd. (NSW, Aust) at a concentration of 100 units/mL. All drugs were diluted in binding or stimulation buffer prior to use. All plastic tubes, pipette tips and binding plates used in relaxin experiments were coated with Sigmacote® (Sigma Chemical Company, MO, USA).

134

### 5.3 Results

# 5.3.1 Characteristics of [<sup>33</sup>P]-H2RLX binding to the LGR7 receptor

Specific [<sup>33</sup>P]-H2RLX binding occurred in a saturable manner to a single population of binding sites on the LGR7 receptor expressed in CHO K1 cells (Figure 5.1). Determination of the  $B_{max}$  value indicated that the density of LGR7 receptors was 336 ± 73 fmol/mg protein in the whole cell binding assay. [<sup>33</sup>P]-H2RLX showed high affinity for the LGR7 receptor with a pK<sub>D</sub> of 9.33 ± 0.23 (n=4).

# 5.3.2 Competition for [<sup>33</sup>P]-H2RLX binding at the LGR7 receptor

Two forms of human relaxin displayed high affinity for [<sup>33</sup>P]-H2RLX binding at the LGR7 receptor (Figure 5.2a) with H2RLX having the highest affinity with a pK<sub>i</sub> of 9.2  $\pm$  0.19 (n=6) while H3RLX was significantly lower (pK<sub>i</sub> 7.4  $\pm$  0.18, n=5; P<0.004). PRLX also displayed high affinity for the human LGR7 receptor (Figure 5.2b) with a pK<sub>i</sub> of 8.5  $\pm$  0.13 (n=6; P<0.03 cf H2RLX) while both RatINSL3 and Human Insulin failed to compete for [<sup>33</sup>P]-H2RLX binding (n=6 & 5 respectively; Figure 5.2c).

Figure 5.1

[<sup>33</sup>P]-H2RLX saturation binding characteristics at the human LGR7 receptor transiently expressed in CHO-K1 cells.

The specific relaxin radioligand, [ $^{33}$ P]-H2RLX, displayed high affinity (pK<sub>D</sub> 9.33 ± 0.23; n=4) for a single binding site on the LGR7 receptor. Transient expression of the LGR7 receptor in CHO K1 cells gave a  $B_{max}$  of 336 ± 76 fmol/mg protein.





Specific Binding

1000	1500	2000
P]-H2RLX (p	(Me	



# Figure 5.2

ł

Competition for binding at the human LGR7 receptor transiently expressed in CHO-K1 ceils.

These graphs represent the pooled specific affinity data displayed by human and species homologues of relaxin for the human relaxin receptor. Both forms of human relaxin, human gene 2 (H2RLX) and human gene 3 (H3RLX) relaxin, displayed specific affinity for the LGR7 receptor with H2RLX having a higher affinity (pK<sub>i</sub> 9.2  $\pm$  0.19, n=6) than A) H3RLX (pK<sub>i</sub> 7.4  $\pm$  0.18, n=5; P<0.004). B) the porcine relaxin homologue (PRLX) showed high affinity for the human LGR7 receptor (pK<sub>i</sub> 8.5  $\pm$  0.13, n=6; P<0.03 cf H2RLX) while C) structurally related peptides, rat insulin-like peptide 3 (RatINSL3; n=6) and Human Insulin (n=5), failed to compete.



# Table 5.1

Comparison of relaxin and relaxin-like analogue affinity for the naturally expressed human relaxin receptor in THP-1 cells and the putative human relaxin receptor LGR7 transiently expressed in CHO K1 cells. \* denotes P<0.05, \*\* P<0.005 verus H2RLX. N/D denotes affinity not determinable, as the analogue did not compete for binding. Analogue H2RLX H3RLX PRLX SheepINSL3 RatINSL3 Human Insulin Chapter 5: Characterisation of the putative relaxin receptor, LGR7

THP-1	CHO-LGR7
pK <sub>i</sub> (n)	pK <sub>i</sub> (n)
8.6 ± 0.11 (13)	9.2 ± 0.19 (6)
8.4 ± 0.16 (8)	7.4 ± 0.18 (5)**
8.4 ± 0.09 (5)	8.5 ± 0.13 (6)*
5.9 ± 0.32 (4)**	
	N/D (6)
	N/D (5)

### 5.3.3

Figure 5.3c)

Chapter 5: Characterisation of the putative relaxin receptor, LGR7

# Stimulation of cAMP accumulation by the LGR7 receptor

The relaxin analogues that competed with high affinity for binding of [<sup>33</sup>P]-H2RLX at the LGR7 receptor also stimulated a cAMP response. PRLX was the most potent (Figure 5.3b) with a pEC<sub>50</sub> of 9.3  $\pm$  0.08 (n=4), followed by H2RLX 8.6  $\pm$  0.12 (n=4; P<0.03 cf PRLX) and H3RLX 7.7  $\pm$  0.10 (n=4; P<0.03 cf H2RLX) (Figure 5.3a). Interestingly the order of affinity obtained in the [<sup>33</sup>P]-H2RLX binding studies is somewhat different from the order of potency for stimulation of cAMP accumulation with PRLX displaying lower affinity at the LGR7 receptor compared to H2RLX while displaying a higher potency for stimulation of a cAMP response. RatINSL3 and Human Insulin failed to induce a cAMP response above basal levels (n=i;

### Figure 5.3

Stimulation of cAMP accumulation through the human LGR7 receptor transiently expressed in CHO-K1 cells.

These graphs represent the pooled cAMP accumulation data stimulated by the human and species homologues of relaxin through the human relaxin receptor. cAMP accumulation was stimulated by all relaxin analogues with human gene 2 relaxin (H2RLX; pEC<sub>50</sub> 8.6  $\pm$  0.12, n=4) being more potent than A) human gene 3 relaxin (H3RLX; pEC<sub>50</sub> 7.7  $\pm$  0.1, n=4; P<0.03). B) The porcine homologue of relaxin (PRLX) was the most potent relaxin analogue (pEC<sub>50</sub> 9.3  $\pm$  0.08, n=4), being significantly stronger than H2RLX (P<0.03). C) the structurally related peptides, rat insulin-like peptide 3 (RatINSL3; n=4) and Human Insulin (n=4) failed to produce a cAMP response above basal levels.





### 5.4 Discussion

The relaxin receptor has recently been identified as an orphan G-protein coupled receptor, LGR7 (Hsu *et al.*, 2002). The aim of this study was to examine the phasepacological properties of the LGR7 receptor by examining the interaction of several relaxin analogues with the human LGR7 receptor transiently expressed in CHO K1 cells.

The LGR7 receptor displayed high affinity for the relaxin radioligand, [<sup>33</sup>P]-H2RLX. The pK<sub>D</sub> value of 9.33  $\pm$  0.23 for the binding of [<sup>33</sup>P]-H2RLX to the LGR7 receptor was almost identical to the affinity shown in the THP-1 binding assay of 9.37  $\pm$  0.22. This is consistent with previously published reports using THP-1 cells (Parsell *et al.*, 1996) and a human uterine cell line (Osheroff *et al.*, 1995). The binding characteristics of the relaxin receptor do not appear to change markedly with expression levels as THP-1 cells express very low levels of the relaxin receptor with a B<sub>max</sub> of 1.45  $\pm$  0.62 fmol/mg protein (Section 3.3.1) or ~275 receptors/cell (Parsell *et al.*, 1996), and the human uterine cell line expresses low levels of receptors with approximately 1082 receptors/cell (Osheroff *et al.*, 1995) while our transient expression system appeared to express the LGR7 receptor with a B<sub>max</sub> of 336  $\pm$  73 fmol/mg protein. However, these assays were conducted as whole cell assays rather than using membrane preparations, therefore the protein levels reflect the entire cell protein content, diluting the receptor expression values.

Interestingly, the affinity of the relaxin analogues appeared to vary between the LGR7-CHO cell system and the THP-1 cells. Although H2RLX showed a greater affinity for the LGR7 receptor than that shown in the THP-1 binding assay (Table 5.1), H3RLX displayed a lower affinity for the LGR7 receptor than that shown in THP-1 cells. PRLX displayed high affinity for the human LGR7 receptor, similar to the affinity displayed for the relaxin receptor expressed in the THP-1 cell line (Table 5.1), and as previously shown non-relaxin members of the insulin gene superfamily failed to interact with LGR7, as neither RatINSL3 nor human insulin could displace radioligand binding.

Not only did the affinities vary slightly, but H3RLX effectively competed for 100% of binding at the LGR7 receptor whereas it only competed for 50% in the THP-1 binding assay. A possible explanation for this is that low levels of relaxin receptor naturally expressed in the THP-1 cells may be accompanied by expression of other receptors in the cell that may also interact with the relaxin analogues, such as the LGR8 receptor, as both LGR7 and LGR8 mRNA have been identified in THP-1 cells (Nguyen *et al.*, 2003). Our radioligand, [<sup>33</sup>P]-H2RLX, can interact with both the LGR7 and LGR8 receptors (see Chapter 6) while it has been reported that H3RLX displays preferential interaction with LGR7 while displaying no affinity for the LGR8 receptor (Sudo *et al.*, 2003). The transfection of the LGR7 receptor into CHO cells allows the specific interaction between the LGR7 receptor is likely to overshadow any small interaction with other naturally expressed receptors.

Another interesting finding was that PRLX was more potent than the native ligand, H2RLX, for cAMP accumulation while displaying a lower affinity for the LGR7 receptor. This is the first study to demonstrate this discrepancy between the relaxin analogues. A previous study has examined both H2RLX and PRLX (Sudo et al., 2003) and reported an equal potency for the two relaxin analogues at the LGR7 relaxin receptor expressed in HEK293 cells. The conflicting results may reflect slight differences in the interaction of the relaxin analogues at the receptor that is augmented by differences in the signal transduction mechanisms of the two cell lines. A difference in the ability to inhibit cAMP signalling has previously been shown between CHO K1 and HEK 293 cells. The cAMP response in CHO K1 cells was inhibited with greater potency than the response in HEK293 cells (Zhang et al., 1995), possibly indicating a difference in strength of coupling to the cAMP pathway between the two cell lines. H3RLX was the weakest of the relaxin analogues in stimulating a cAMP response, in accordance with published reports (Sudo et al., 2003). Also in accordance with the binding studies, neither RatINSL3 nor human insulin could stimulate a cAMP response through the LGR7 receptor.
Chapter 5: Characterisation of the putative relaxin receptor, LGR7

In conclusion, the relaxin receptor has been positively identified as the LGR7 receptor. H2RLX and PRLX displayed high affinity and potency for the LGR7 receptor, while H3RLX interaction was significantly weaker. Rat INSL3 and human insulin failed to maintain either affinity or efficacy for the relaxin LGR7 receptor. These results correlate well with previous studies using THP-1 cells that have been found to express the LGR7 receptor (Nguyen *et al.*, 2003).

# **Chapter Six**

# Characterisation of LGR7 & LGR8 receptors

#### 6.1 Introduction

The protracted search for the relaxin receptor concluded with the characterisation of the relaxin LGR7 receptor (Hsu *et al.*, 2002). However, it was also demonstrated that relaxin interacted with a second receptor, LGR8. Relaxin displayed higher affinity for LGR7 than LGR8 but the interaction with LGR8 was shown to be a specific relaxin interaction that could not be replicated by insulin, IGF's or glucagon (Hsu *et al.*, 2002; Sudo *et al.*, 2003). The LGR7 receptor behaved in the same manner and showed the same distribution as the relaxin receptor (Chapter 5) leading to the conclusion that LGR7 was the classical relaxin receptor as defined by functional and binding studies, leaving the relevance of the interaction between relaxin and the LGR8 receptor unclear.

The LGR7 receptor was isolated from EST sequences and classified as a new subclass of LGR. Subfamily C LGR's are characterised by a unique low-density lipoprotein (LDL) domain within the N-terminus of the ectodomain (Hsu, 2003; Hsu *et al.*, 2000). LGR7 has approximately 24% homology with Subfamily A LGR's, including the well characterised LH, FSH and TSH receptors, and only slightly higher homology (< 50%) with the Subfamily B LGR's, three orphan receptors known as LGR4, LGR5 and LGR6 (Hsu, 2003). Recently, another orphan LGR isolated from EST sequences was found to have > 90% homology with the LGR7, including the characteristic Subfamily C LDL domain. The new receptor became the second Subfamily C LGR, and was chronologically labelled the LGR8 receptor (Hsu, 2003; Hsu *et al.*, 2002).

Further analysis of the LGR8 sequence positively identified it as the human homologue to the mouse GREAT receptor (Hsu *et al.*, 2002). GREAT receptor gene knockout mice (Gorlov *et al.*, 2002; Overbeek *et al.*, 2001) displayed the phenotype of abnormal testis decent, uncannily similar to INSL3 knockout mice (Adham *et al.*, 2000; Nef *et al.*, 1999; Zimmermann *et al.*, 1999). The relaxin-related peptide, INSL3, was found to be another ligand for the LGR8 receptor in both binding and functional studies (Bogatcheva *et al.*, 2003; Kumagai *et al.*, 2002; Sudo *et al.*, 2003). Chapter of: Characterisation of LGR7 & LGR8 receptors

All three species homologues of INSL3, human, porcine and rat INSL3, as well as biotinylated porcine INSL3 and PRLX were capable of stimulating an LGR8 cAMP response with similar potency (Kumagai *et al.*, 2002). At the LGR7 receptor however only PRLX produced a cAMP response and INSL3 analogues had no efficacy. However not all relaxin analogues interact with LGR8 as one study has shown the preferential interaction of H3RLX with LGR7 while displaying no affinity or potency at LGR8 (Sudo *et al.*, 2003).

The confirmation of LGR7 as the relaxin receptor, LGR8 as the INSL3 receptor and the differential interaction displayed by human relaxin analogues raised interesting questions. Do other relaxin analogues interact differentially with the LGR7 and LGR8 receptors and does interactic.) translate into functional signalling? The aim of this study was to investigate the binding characteristics of and functional response to various species homologues of relaxin at the LGR7 and LGR8 receptors.

#### 6.2 Methods

LGR7 and LGR8 receptors were transfected and stably expressed in an adherent cell line. The relaxin analogues used included H2RLX, PRLX, RatRLX and rhesus monkey relaxin (RMRLX), as well as the human homologue of INSL3 (HINSL3).

# 6.2.1 Production of a cell line stably expressing the LGR7 and the LGR8 receptors

Transfection of the LGR7 and LGR8 receptor cDNA in the pcDNA3.1-Zeo expression vector (Invitrogen, CA, USA) into the CHO K1 cell line was preformed using a LIPOFECTAMINE<sup>TM</sup> transfection method as described in Section 2.1.4. Development into a stable cell line was as described in Section 2.1.5. Screening of the CHO K1 cell colonies for expression levels of the LGR7 and LGR8 receptors showed extremely low levels of receptor expression (data not shown), resulting in colonies that were unusable for further investigation.

HEK293 cell lines stably expressing the LGR7 and LGR8 receptors were successfully produced as described in Section 2.1.5 and kindly provided by our collaborator Sharon Layfield at the Howard Florey Institute of Experimental Physiology and Medicine, Vic, Aust.

#### 6.2.2 Cell Culture

HEK293 cells stably expressing either the LGR7 or LGR8 receptors were maintained in RPMI 1640 media containing glutamine (2 mM), penicillin (100 units/mL), streptomycin (100  $\mu$ g/mL), heat inactivated FBS (10% v/v) and zeocin (200 $\mu$ M). All tissue culture plates and flasks were coated with poly-L-lysine (0.1 mg/mL) prior to cell culture to ensure cell adhesion. HEK293 cells were maintained below ~80% confluence and split as described in Section 2.1.1.2.

### 6.2.3 High through-put 96 well binding assay

The HEK293 LGR7/LGR8 cell binding assay was set up as described in Section 2.3.3. The total volume used was 100  $\mu$ L/well, comprising 20  $\mu$ L of [<sup>33</sup>P]-H2RLX ± 10  $\mu$ L un!abelled relaxin or relaxin-like analogue and 50  $\mu$ L cell suspension, made up to 100  $\mu$ L with binding buffer. All experiments were performed in duplicate. Following a 90 minute incubation at 25°C, cells were harvested onto GF/C filter plates (Skudtek, Vic, Aust) pre-soaked overnight (0.5% PEI, 4°C). Filter plates were dried at 42°C, Microscint O (PerkinElmer, Melb, Aust) added (30  $\mu$ L/well) and equilibrated overnight. Filters were counted on a Topcount scintillation counter (PerkinElmer, Melb, Aust) and data was analysed with GraphPad PRISM (Section 2.8).

# 6.2.3.1 [<sup>33</sup>P]-H2RLX (B33) preparation

H2RLX (B33) was labelled with  $[\gamma^{-33}P]$ -ATP using a procedure based on Osheroff *et al* (1990) as described in Section 2.3.1.

#### 6.2.3.2 Saturation binding protocol

Saturation binding studies were used to determine the affinity ( $pK_D$ ) of [<sup>33</sup>P]-H2RLX and number of LGR7 and LGR8 binding sites ( $B_{max}$ ) expressed in HEK293 cells (Section 2.8). Saturation binding studies were performed using a concentration range of 50 pM – 2.5 nM. Specific binding was calculated by the total binding – non-specific binding (10  $\mu$ M H2RLX) at each [<sup>33</sup>P]-H2RLX concentration. Protein concentrations were determined as described in Section 2.5.1 and the data was plotted as specific binding (fmol/mg protein) against concentration of radioligand used (pM).

#### 6.2.3.3 Competition binding protocol

Competition binding assays allowed the calculation of the affinity (pK<sub>i</sub>) of various relaxin analogues for the LGR7 and the LGR8 receptors expressed in HEK293 cells (Section 2.8). Competition binding was performed using [<sup>33</sup>P]-H2RLX at a concentration of 100 pM for LGR7 competition studies and two concentrations, 100 pM and 1 nM, for the LGR8 competition studies. The concentration range used for H2RLX, PRLX, Rat%LX, RMRLX and HINSL3 analogues was 1 pM – 100 nM, with the exception of a single concentration of HINSL3 (100 nM) examined at the LGR7 receptor. Non-specific binding was determined using 1  $\mu$ M H2RLX for the LGR7 receptor and 1  $\mu$ M HINSL3 for the LGR8 receptor. Results were expressed as a percentage of maximal specific binding and pK<sub>i</sub> values were calculated (Section 2.8).

#### 6.2.4 cAMP accumulation assay

The ability of the relaxin analogues to stimulate a functional response through the LGR7 or LGR8 receptors was examined with the AlphaScreen<sup>TM</sup> cAMP accumulation assay (Section 2.4.2). Briefly, cells were plated into a 96-well tissue culture plate 24 hours before each experiment at 10,000 cells/well. Cells were incubated in stimulation buffer containing 10  $\mu$ L relaxin analogue (10 pM – 1  $\mu$ M) or vehicle (PBS) for 30 minutes (37°C). Stimulation mix was removed, the cells lysed and stored in lysis buffer (-20°C). cAMP samples were analysed using an AlphaScreen<sup>TM</sup> cAMP detection kit (PerkinElmer, Melb, Aust) according to the manufacturers instructions (Section 2.4.2). Detection plates were read by Fusion<sup>TM</sup>  $\alpha$  microplate reader (PerkinElmer, Melb, Aust) and data were analysed against a cAMP standard curve with GraphPad PRISM (Section 2.8).

### 6.2.5 Analysis

Results are expressed as mean  $\pm$  S.E.M. of n experiments (Section 2.8). Saturation binding data were analysed with GraphPad PRISM. Radioligand affinity (pK<sub>D</sub>) and LGR7 or LGR8 receptor expression (B<sub>max</sub>) values were calculated and expressed in pM and fmol/mg protein respectively. Competition binding data were analysed with GraphPad PRISM and relaxin analogue affinity (pK<sub>i</sub>) values were calculated using the Cheng & Prusoff (1973) equation. cAMP accumulation data were analysed with GraphPad PRISM and analogue potency (pEC<sub>50</sub>) values calculated (Section 2.8). Statistical analysis was completed using GraphPad PRISM statistical programs with competition binding and functional response curves analysed using a two-way ANOVA. pK<sub>i</sub> and pEC<sub>50</sub> values were analysed using a Student's t-test. P values of less than or equal to 0.05 were considered significant.

#### 6.2.6 Drugs and reagents

All drugs were used as described in Section 2.9. Stock solutions of H2RLX, PRLX, RMRLX, RatRLX and HINSL3 analogues were prepared in 0.1% aqueous trifluoroacetic acid (TFA) within the concentration range of 10-100  $\mu$ M. All drugs were diluted in binding or stimulation buffer prior to use. All plastic tubes, pipette tips and binding plates used in relaxin experiments were coated with Sigmacote® (Sigma Chemical Company, MO, USA).

# 6.3 Results

# 6.3.1 Characteristics of [<sup>33</sup>P]-H2RLX binding

### 6.3.1.1 [<sup>33</sup>P]-H2RLX binding characteristics of the LGR7 receptor

[<sup>33</sup>P]-H2RLX binding occurred in a saturable manner to a single population of sites on the LGR7 receptor (Figure 6.1). Analysis of the binding data gave a  $B_{max}$  of 6.7 ± 0.21 fmol/mg protein in the whole cell binding assay, indicating that the LGR7 receptor was expressed at higher levels than in THP-1 cells ( $B_{max}$  1.45 ± 0.62 fmol/mg protein; Chapter 3). [<sup>33</sup>P]-H2RLX showed high affinity for the LGR7 receptor with a pK<sub>D</sub> of 9.5 ± 0.15 (n=4)

# 6.3.1.2 [<sup>33</sup>P]-H2RLX binding characteristics of the LGR8 receptor

 $[^{33}P]$ -H2RLX binding occurred in a saturable manner to the LGR8 receptor (Figure 6.2a), with a calculated pK<sub>D</sub> of 9.0 ± 0.10 (n=6) and B<sub>max</sub> of 13.3 ± 1.5 fmol/mg protein. Although the data conformed to a single binding site model, closer analysis of the data revealed a two site binding pattern. Conversion of the data into a Scatchard plot (Figure 6.2b) resulted in two affinity sites, a higher affinity site (pK<sub>D</sub> 9.3) and a lower affinity site (pK<sub>D</sub> 8.1). However the affinity values could only be calculated from the grouped data (n=6) as many of the individual data sets would not allow the calculation of the two affinity sites. This can probably be explained by the concentration range of [<sup>33</sup>P]-H2RLX used in some experiments which was not high enough to clearly define the second binding site, but also to variability within the data sets. The two binding sites resulted in two B<sub>max</sub> values of 4.35 and 29.34 fmol/mg protein.

[<sup>33</sup>P]-H2RLX saturation binding characteristics at the LGR7 receptor.

The specific relaxin radioligand, [<sup>33</sup>P]-H2RLX, displayed high affinity (pK<sub>D</sub> 9.5  $\pm$  0.15; n=4) for a single binding site on the LGR7 receptor. Stable expression of the LGR7 receptor in HEK293 cells gave a  $B_{max}$  of 6.7 ± 0.21 fmol/mg protein.





Specific Binding

1000	1500	2000	2500
( <sup>33</sup> P)-H	2RLX (n	M)	

# [<sup>33</sup>P]-H2RLX saturation binding characteristics at the LGR8 receptor.

A) the LGR8 receptor stably expressed in HEK293 cells displayed high affinity for  $[^{33}P]$ -H2RLX with a pK<sub>D</sub> 9.0 ± 0.10 (n=6) and B<sub>max</sub> of 13.3 ± 1.5 fmol/mg protein. Conversion of the data B) into a Scatchard plot revealed the presence of two binding sites (n=6). Analysis of the grouped data showed C) a higher and lower affinity binding site, with pK<sub>D</sub> values of 9.3 & 8.1 and  $B_{max}$  values of 4.35 & 29.34 fmol/mg protein respectively.



# 6.3.2 **Competition binding characteristics** 6.3.2.1

Figure 6.3c).

#### 6.3.2.2 Competition for binding at the LGR8 receptor

# Competition for binding at the LGR7 receptor

All relaxin analogues competed for [<sup>33</sup>P]-H2RLX binding and had high affinity for the LGR7 receptor. The order of affinity of RLX analogues for the LGR7 receptor was H2RLX (pK<sub>i</sub> 9.4  $\pm$  0.10, n=6)  $\geq$  RMRLX (pK<sub>i</sub> 9.3  $\pm$  0.10, n=6; Figure 6.3b)  $\geq$ PRLX ( $pK_i$  9.0 ± 0.12, n=6; Figure 6.3a) >> RatRLX ( $pK_i$  7.2 ± 0.18, n=6; P<0.002 cf H2RLX) (Figure 6.3c) whereas HINSL3 failed to compete for binding (n=4;

Competition for binding at the LGR8 receptor was examined using two concentrations of [<sup>33</sup>P]-H2RLX with the aim of further exposing the two LGR8 binding sites. 100 pM [<sup>33</sup>P]-H2RLX was used to examine the higher affinity binding sites while 1 nM was used to expose binding at a larger proportion of the lower affinity sites. The order of affinity at LGR8 (Figure 6.4) was HINSL3 (pK<sub>i</sub> 9.7  $\pm$ 0.15, n=5; P<0.004 cf H2RLX) > H2RLX (pK<sub>i</sub> 8.6  $\pm$  0.14, n=6) ≥ PRLX (pK<sub>i</sub> 8.2  $\pm$ 0.10, n=5 > RMRLX (pK<sub>i</sub> 7.3 ± 0.20, n=4; P<0.01 cf H2RLX). RatRLX showed no affinity for the LGR8 receptor (n=4). Increasing the concentration of [<sup>33</sup>P]-H2RLX failed to significantly alter the affinity of the competitive analogues (Figure 6.5; Table 6.1), although there was a trend for a decrease in the ability of H2RLX (Figure 6.5a) and HINSL3 (Figure 6.5d) to compete for [<sup>33</sup>P]-H2RLX binding.

# Competition for binding at the LGR7 receptor

All species homologues of relaxin displayed high affinity for the LGR7 receptor. Similar affinity was shown by H2RLX ( $pK_i 9.4 \pm 0.10$ , n=6), A) PRLX ( $pK_i 9.0 \pm 0.12$ , n=6) and B) RMRLX ( $pK_i 9.3 \pm 0.10$ , n=6). C) RatRLX displayed the lowest relaxinlike affinity ( $pK_i 7.2 \pm 0.18$ , n=6; P<0.002 cf H2RLX) while HINSL3 failed to compete for the LGR7 receptor (n=4).



Chapter 6: Characterisation of LGR7 & LGR8 receptors

# Competition for [<sup>33</sup>P]-H2RLX (100pM) binding at the LGR8 receptor

H2RLX and A) PRLX analogues displaying significant lower affinity for the LGR8 receptor, pK<sub>i</sub> 8.6  $\pm$  0.14 (n=6; P<0.004 cf HINSL3) and 8.2  $\pm$  0.10 (n=5) respectively. B) RMRLX displayed the lowest affinity, pK<sub>i</sub> 7.3  $\pm$  0.20 (n=4; P<0.01 cf H2RLX), while C) RatRLX failed to compete (n=4). C) HINSL3 displayed the highest affinity with a pK<sub>i</sub> of 9.7  $\pm$  0.19 (n=5).



Chapter 6: Characterisation of LGR7 & LGR8 receptors

Competition for [<sup>33</sup>P]-H2RLX binding at the LGR8 receptor using high and low ligand concentrations.

The affinity of A) H2RLX (n=6), B) PRLX (n=5-6), C) RMRLX (n=4) and D) HINSL3 (n=5) for the LGR8 receptor was not significantly altered by an increase in radioligand concentration. Competition for 100 pM [<sup>33</sup>P]-H2KLX is shown by closed symbols while competition for 1 nM [<sup>33</sup>P]-H2RLX binding is shown by open symbols.





A)

B)

**C**)

D)

**Kspecific binding** 

**%specific binding** 

# Table 6.1

The affinity  $(pK_i)$  of relaxin and HINSL3 analogues for the LGR7 and LGR8 receptors stably expressed in HEK293 cells. Competition for binding at LGR7 occurred at a single site, while affinity for the LGR8 receptor was examined using two concentrations of [<sup>33</sup>P]-H2RLX (100pM and 1nM) with the aim of further exposing the two LGR8 binding sites. \* denotes P<0.01, \*\* P<0.005 versus H2RLX (unpaired t-test). N/D denotes affinity not determinable, as the analogue did not compete for binding.

Analogues	LGR7-HEK	LGR8-HEK	LGR8-HEK	
	pK <sub>i</sub> (n)	рК <sub>і</sub> (n)	рК <sub>і</sub> (n)	
H2RLX	9.4 ± 0.10 (6)	8.6 ± 0.14 (6)	8.2 ± 0.18 (6)	
PRLX	9.0 ± 0.12 (6)	8.2 ± 0.10 (5)	8.2 ± 0.09 (6)	
RatRLX	7.2 ± 0.18 (6)**	N/D (4)	N/D (5)	
RMRLX	9.3 ± 0.10 (6)	7.3 ± 0.20 (4)*	6.9 ± 0.21 (4)*	
HINSL3	N/D (4)	9.7 ± 0.19 (5)**	9.1 ± 0.11 (5)**	

Chapter 6: Characterisation of LGR7 & LGR8 receptors

cAMP acc
Stimulatic receptor s

Figure 6.6c). .

.

Chapter 6: Characterisation of LGR7 & LGR8 receptors

### cumulation assay

# on of cAMP production by activation of the LGR7 stably expressed in HEK293 cells

All species homologues of relaxin activated the LGR7 receptor and stimulated a cAMP response. Initial analysis of the data with a sigmoidal dose-response curve (GraphPad Prism) showed that H2RLX (pEC<sub>50</sub> 9.4  $\pm$  0.42, n=4) and PRLX (pEC<sub>50</sub> 9.4  $\pm$  0.30, n=4) displayed the highest potency at the LGR7 receptor (Figure 6.6a), followed by RMRLX with a pEC<sub>50</sub>  $8.9 \pm 0.16$  (n=3; Figure 6.6b). However there appeared to be anomalies between the data points and the generated curve of best fit, therefore the data was re-analysed and found to conform to a two site model (Figure 6.7). The re-calculation result: in two pEC<sub>50</sub> values for each analogue; H2RLX,  $10.2 \pm 0.11 \& 8.1 \pm 0.19$  (n=4); PRLX,  $9.8 \pm 0.14 \& 8.0 \pm 0.36$  (n=4); RMRLX, 9.9 $\pm$  0.21 & 8.0  $\pm$  0.15 (n=3); with a high affinity state fraction of 0.7, 0.75 and 0.51 respectively. Interestingly RatRLX conformed to a single site model only, displaying the weakest potency with a pEC<sub>50</sub> of 8.2  $\pm$  0.34 (n=4; Figure 6.6c). HINSL3 failed to activate LGR7 and cause a significant cAMP response (n=4;

### Stimulation of cAMP production by activation of the LGR7 receptor

cAMP accumulation was stimulated by all relaxin analogues with H2RLX (pEC<sub>50</sub> 9.4  $\pm$ 0.42; n=4) and A) PRLX (pEC<sub>50</sub> 9.4  $\pm$  0.30; n=4) displaying the highest potency, followed by B) RMRLX (pEC<sub>50</sub> 8.9  $\pm$  0.16; n=4) and C) RatRLX (pEC<sub>50</sub> 8.2  $\pm$  0.34; n=4). HINSL3 (n=4) failed to produce a cAMP response above busal levels.

A) 100-BDOR 2 75-% H2RLX maximum 50-25-0 B) % H2RLX maximum response 100 75-50-25-C) % H2RLX maximum respon 100-75-50-25-



# Stimulation of cAMP production at the LGR7 receptor

Further analysis of the cAMP response produced by the relaxin analogues at the LGR7 receptor showed stimulation in a biphasic manner. A) H2RLX (pEC<sub>50</sub> 10.2  $\pm$  0.11 & 8.1  $\pm$  0.19, n=4); B) PRLX (pEC<sub>50</sub> 9.8  $\pm$  0.14 & 8.0  $\pm$  0.36, n=4); C) RMRLX (pEC<sub>50</sub> 9.9  $\pm$  0.21 & 8.0  $\pm$  0.15, n=3).











Chapter 6: Characterisation of LGR7 & LGR8 receptors

161

l

# 6.3.3.2 Stimula receptor

HINSL3 displayed the highest potency at the LGR8 receptor (pEC<sub>50</sub> 9.1  $\pm$  0.17, n=4; Figure 6.8c) with H2RLX (pEC<sub>50</sub> 7.9  $\pm$  0.16, n=4; P<0.03) and PRLX (pEC<sub>50</sub> 8.1  $\pm$  0.21, n=3; P<0.03) (Figure 6.8a) displaying significantly lower potency. The potency of RMRLX was dramatically reduced at the LGR8 receptor with a pEC<sub>50</sub> of 6.4  $\pm$  0.01 (n=3; P<0.001, Figure 6.8b) while RatRLX failed to stimulate a cAMP response (n=4; Figure 6.8c). Although HINSL3 was the most potent, it failed to produce a maximal cAMP response with the dose-response curve reaching a plateau at approximately 70% of the maximal response to H2RLX.

Chapter 6: Characterisation of LGR7 & LGR8 receptors

# Stimulation of cAMP production by activation of the LGR8 receptor stably expressed in HEK293 cells

# Stimulation of cAMP production by activation of the LGR8 receptor

The rank order of potency for cAMP accumulation was similar to the rank order of affinity with H2RLX (pEC<sub>50</sub> 7.9  $\pm$  0.16, n=4) and A) PRLX (pEC<sub>50</sub> 8.1  $\pm$  0.21, n=4) displaying high potency, followed by B) RMRLX (pEC<sub>50</sub> 6.4  $\pm$  0.01, n=4). C) RatRLX (n=4) failed to produce a cAMP response above basal levels while HINSL3 displayed the highest potency (pEC<sub>50</sub> 9.1  $\pm$  0.17, n=4; P<0.03 cf H2RLX).

÷.



#### 6.4 Discussion

While the relaxin receptor was identified as the LGR7 receptor, the same study also reported an interaction between relaxin and the LGR8 receptor (Hsu *et al.*, 2002). The native ligand for the LGR8 receptor was later shown to be the INSL3 peptide (Bogatcheva *et al.*, 2003; Kumagai *et al.*, 2002), further highlighting the unexpected interaction between this receptor and relaxin. The aim of this study was to examine the interaction between various relaxin analogues and HINSL3 with the LGR7 and LGR8 receptors.

[<sup>33</sup>P]-H2RLX binding was saturable to a single population of sites on the LGR7 receptor whereas the radioligand appeared to bind to two sites at the LGR8 receptor. This was the first indication of a difference between the actions of relaxin at the two receptors. The appearance of two LGR8 binding sites did not result from overexpression of the receptor compared to the LGR7 receptor as the  $B_{\text{max}}$  values revealed similar levels of receptor expression. [<sup>33</sup>P]-H2RLX binding to LGR7 and LGR8 was of high affinity, comparable with previously determined [<sup>33</sup>P]-H2RLX affinity for LGR7 receptors transiently expressed in a CHO-K1 cell system (pKp 9.33  $\pm$  0.23; Chapter 5), and naturally expressed in THP-1 cells (pK<sub>D</sub> 9.37  $\pm$  0.22; Chapter 3). While the LGR8 saturation curve was able to conform to a two site Scatchard plot, it was difficult to show a difference between the sites in competition binding at the two concentrations of radioligand used in this study. The relative lack of difference displayed by the relaxin analogues in the competition assays could be attributed to the unselective labelling of the two sites by [<sup>33</sup>P]-H2RLX. Using the affinity and B<sub>max</sub> values calculated from the two site LGR8 saturation binding model, an estimated fraction of occupancy (FO) was determined for each of the affinity sites. The equation for FO was taken from GraphPad PRISM; FO x  $B_{max} = (B_{max} x)$  $[radioligand])/(K_D + [radioligand])$ . Competition binding experiments conducted with 100pM radioligand should occupy 16% of the higher affinity sites while labelling only 1% of the lower affinity sites. Increasing the concentration of radioligand to 1nM increased the FO to 67% for the high affinity site and 11% for the low affinity site, so that 10% of the total number of sites occupied would be low

Chapter 6: Characterisation of LGR7 & LGR8 receptors

affinity sites. To occupy greater than 80% of the low affinity sites, the concentration of the radioligand would need to be increased to 40 nM.

The generation of LGR7/LGR8 receptor chimeras (Sudo *et al.*, 2003) containing the ectodomain of LGR7 attached to the transmembrane domain of LGR8, LGR7/8 receptor, and the complementary LGR8/7 receptor, have been used to show two separate LGR8 binding sites. Differences in the affinities displayed by relaxin analogues in competition binding studies at the chimeric receptors suggested that a higher affinity LGR8 binding site resided in the receptor ectodomain while a lower affinity site appeared to be located in the transmembrane domain (Halls, 2003; Halls *et al.*, 2003). The ectodomain of the receptor seemed to dictate the majority of the receptor properties with relaxin analogues interacting at LGR7/8 with similar affinity and potency to LGR7 while LGR8/7 followed LGR8 properties. The transmembrane domain does maintain a role in determining the receptor characteristics as further manipulation of the LGR7/8 chimera to contain the individual extracellular loops of LGR7, as well as the ectodomain, revealed the importance of extracellular loop 2 for maintaining H3RLX interaction at LGR7 (Sudo *et al.*, 2003).

While the binding studies showed that LGR7 displayed a single affinity site, the cAMP stimulation assay appeared to conform to a biphasic pattern. The relaxin induced cAMP accumulation is reportedly a biphasic response, involving a rapid increase in cAMP by the classical  $G_S$  stimulated pathway followed by a slightly delayed and sustained increase by a phosphoinositide 3-kinase (PI3K) dependent pathway (Nguyen *et al.*, 2003). Interestingly, not all relaxin analogues appear to stimulate in a biphasic manner as shown by the monophasic response to RatRLX. A possible explanation is that the biphasic response is the result of ligand directed signalling (Kenakin, 1995), where the individual relaxin analogues can differentially direct the coupling of the receptor to various signalling pathways involved in the production of cAMP. This could also explain the difference seen in the maximal responses produced by the relaxin analogues and HINSL3 at the LGR8 receptor. Relaxin signalling through LGR8 may involve coupling to a pathway that INSL3 fails to activate. Further experiments are needed to confirm these results and to

explore the mechanisms involved in relaxin and INSL3 signalling through the LGR7 and LGR8 receptors.

Research is showing that not only do relaxin and INSL3 interact differently at LGR7 and LGR8, but relaxin analogues also interact differently with the receptors. While H2RLX and PRLX showed strong affinity for both LGR7 and LGR8, RatRLX, and to a lesser degree RMRLX, displayed preferential interaction with LGR7 over LGR8. H3RLX has also been reported to interact with the LGR7 receptor while having no affinity for the LGR8 receptor (Sudo *et al.*, 2003). This study also showed that while extracellular loop 2 is important for H3RLX interaction with LGR7, H2RLX and PRLX show equal preference for the extracellular loops of LGR7 and LGR8, as interchanging the LGR7/LGR8 extracellular loops of the LGR7/8 chimera had no affect on either affinity or efficacy for the receptor. Examination of the importance of the extracellular loops to RatRLX may identify a common factor with H3RLX that dictates the selectivity for the LGR7 receptor.

The ability of the human relaxin analogues to interact differentially with receptors has been further highlighted with the identification of two new receptors that display high affinity for H3RLX. The orphan receptors GPCR135 (Liu *et al.*, 2003b) and GPCR142 (Liu *et al.*, 2003a) bind H3RLX but have no affinity for PRLX or other members of the human insulin-relaxin superfamily, including insulin, IGF I & II, INSL3, INSL4 & INSL6 peptides. The GPCR135 and GPCR142 receptors are Type I GPCR's with a short N terminal extracellular domain and couple primarily to the G<sub>i</sub> subunit, completely different to the LGR7 and LGR8 Type III GPCR's with the large N terminal extracellular domain and primarily G<sub>S</sub> coupling. The interaction of H1RLX and H2RLX at the two new receptors was not completely explored although preliminary data suggests that only H3RLX interacts with these receptors (Liu *et al.*, 2003b). An interesting detail to note was that while separated A and B chains of H2RLX fail to interact with the relaxin LGR7 receptor, the separated H3RLX B chain displays affinity for the GPCR135 and GPCR142 receptors, although at some 3 log units lower affinity than the full peptide.

#### Chapter 6: Characterisation of LGR7 & LGR8 receptors

It is difficult to predict the physiological consequences of relaxin interacting at the LGR7 and LGR8 receptors due to the uncertainty of the specific physiological actions of relaxin and INSL3 in the human. Although it has been established that relaxin maintains a role in reproduction that is likely to involve implantation (Chapter 7) and that INSL3 is critical for testicular development (Ferlin et al., 2003; Roh et al., 2003), the consequences of a cross-interaction between the relaxin peptides and the LGR8 receptor is still unknown. Human tissues reported to express LGR8 receptor mRNA transcripts include the brain, kidney and uterine tissues (Hsu et al., 2002) while relaxin mRNA transcripts have been identified in human uterine tissues (Yki-Jarvinen et al., 1983). Although there is little information concerning the expression of relaxin mRNA transcripts in human brain and kidney tissues, studies have shown relaxin mRNA expression in the rat brain and kidney tissues (Burazin et al., 2002; Gunnersen, 1995). While the co-localised expression of relaxin and LGR8 mRNA does allow for cross-interaction, further investigation is needed however to establish that relaxin does interact with the LGR8 receptor in vivo, rather than just in the *in vitro* situation.

In conclusion, H2RLX and PRLX analogues could interact with high affinity and potency at both the LGR7 and LGR8 receptors. RMRLX and RatRLX analogues interacted selectively with the LGR7 receptor, while HINSL3 selectively interacted with the LGR8 receptor. The ability of the relaxin analogues to interact selectively with the two receptors could be further developed as a useful research tool to examine the individual physiological properties of the LGR7 and LGR8 receptors.

# **Chapter Seven**

LGR7 expression in the human uterus

#### 7.1 Introduction

Relaxin is primarily produced by the corpus luteum or placenta and affects connective tissue remodelling in the cervix and vagina in preparation for parturition (Sherwood, 1994). Studies in relaxin knockout mice have also demonstrated an essential role for the peptide in mammary gland development and nipple growth (Zhao *et al.*, 1999). In humans, however, the physiological role of H2RLX is associated more with degradation of foetal membrane collagen at term (Bogic *et al.*, 1997) rather than remodelling of the cervix. Moreover, labour is not prolonged in pregnant women who lack circulating relaxin following ovum donation (Emmi *et cl.*, 1991)

In contrast to many other mammalian species, there is no evidence of a preterm surge in serum relaxin in pregnant women. Maximum concentrations of the hormone (0.8 ng/ml) are measured in the first trimester of pregnancy, after which there is a steady decline to term (Eddie *et al.*, 1986). Low plasma relaxin levels are also detected during the luteal phase of the menstrual cycle, with small increases 6-7 days after the LH surge (Stewart *et al.*, 1990). The source of H2RLX in normally cycling women is the corpus luteum (Goldsmith *et al.*, 1995; Ivell *et al.*, 1989; Khan-Dawood *et al.*, 1989), with substantial gene expression observed in the late luteal phase of the menstrual cycle. However, it has been recently shown that local synthesis of H2RLX occurs in human endometrial stromal and glandular epithelial cells taken from both proliferative and secretory phases of the menstrual cycle (Palejwala *et al.*, 2002). Immunoreactive relaxin has also been localized to decidualised stromal cells in the late secretory phase and in early pregnancy (Bryant-Greenwood *et al.*, 1993) implying that the endometrium provides an additional local source of RLX in women.

Several studies show that relaxin has a broad range of effects on the human endometrium and may mediate stromal cell differentiation and/or vascularization. Relaxin stimulates the production of a number of factors from endometrial cells *in vitro* including prolactin (Telgmann *et al.*, 1998), insulin-like growth factor binding protein-1 (IGFBP-1) (Bell *et al.*, 1991), glycodelin (Tseng *et al.*, 1999) and vascular

Chapter 7: LGR7 expression in the human uterus

endothelial growth factor (VEGF) (Palejwala et al., 2002; Unemori et al., 1999). The effects of relaxin on VEGF secretion differs depending on the stage of the menstrual cycle, with a negative influence observed in the proliferative phase and a stimulatory effect in the secretory phase (Palejwala et al., 2002). Of particular interest is the ability of relaxin to stimulate VEGF expression in the secretory phase of the cycle and the suggestion that the increased incidence of menometrorrhagia in relaxin-treated patients is due to increased VEGF production (Unemori et al., 1999). Although relaxin clearly stimulates cAMP production in cultured human endometrial cells (Fei et al., 1990), little is known about the expression of relaxin receptors and their intracellular signalling pathways. Using biotinylated PRLX, binding sites were demonstrated in both the endometrium and myometrium of women, with substantial binding observed in the luminal and glandular epithelium (Kohsaka et al., 1998). Specific relaxin binding has also been shown in marmoset endometrial stromal cells, with weak signals observed in the epithelium (Einspanier et al., 2001). Biotinylated relaxin-binding is highest in the endometrial stroma during the secretory phase of the cycle and in early pregnancy.

To date the relaxin LGR7 receptor gene transcript has been identified in the human uterus (Hsu *et al.*, 2000) but no assessment of gene expression or receptor localization has been conducted throughout the human menstrual cycle. The aim of this study was to examine and characterise the relaxin receptor, correlating expression with LGR7 gene expression in human uterine tissue throughout the menstrual cycle.

### 7.2 Methods

#### 7.2.1 Human tissue collection and preparation

Full thickness endometrial tissue was collected from 27 hysterectomy operations performed on ovulating women suffering from menorrhagia (age range 34-46) as described in Section 2.3.2.1.

#### 7.2.1.1 Preparation of slide mounted uterus sections

Tissue blocks containing the intact endometrial layer with several mm of myometrium attached were frozen in liquid nitrogen and stored at  $-80^{\circ}$ C. Frozen blocks were oriented so that sections would be cut at right angles to the endometrial luminal surface, sectioned in a cryostat at  $-18^{\circ}$ C and collected on precleaned microscope slides coated with poly-l-lysine. Following routine haematoxylin and eosin staining, tissue samples were sorted into 7 groups according to the phases of the menstrual cycle by an experienced histopathologist using established criteria for the normal menstrual cycle (Noyes *et al.*, 1950). The 7 phases of the menstrual cycle were early proliferative (EP), mid proliferative (MP), late proliferative (LP), early secretory (ES), mid secretory (MS), late secretory (LS) and menstrual (M). Frozen sections (10µm) were then used for quantitative autoradiography to determine expression levels and location of the relaxin receptor.

#### 7.2.1.2 RNA extraction from human uterus tissue samples

RNA samples w re prepared by Dr Gareth Weston (Monash Medical Centre, Vic, Aust). Total RNA was extracted from thin sections of the frozen specimens homogenized in TRIZOL reagent (Invitrogen, Mt. Waverley, Vic, Australia) according to the manufacturer's protocol. Purity of the RNA was enhanced by precipitating the aqueous phase of the TRIZOL preparation with an equal volume of 100% ethanol, followed by a second extraction through a QIAGEN RNeasy column (Qiagen, Hilden, Germany). After resuspension in RNase-free water, RNA was precipitated overnight with ethanol as previously described (Weston *et al.*, 2002).

# 7.2.2 Expression and localisation of the relaxin receptor using [<sup>33</sup>P]-H2RLX

# 7.2.2.1 [<sup>33</sup>P]-H2RLX (B33) preparation

H2RLX was labelled wit <sup>33</sup>P]-ATP using a procedure based on that of Osheroff et al (1990) as describer ction 2.3.1.

# 7.2.2.2 Relaxin receptor expression throughout the phases of the menstrual cycle

Slide-mounted uterus sections were processed as described in Section 2.3.2.2. Briefly, slides were preincubated in 100µL of binding buffer (25mM Hepes and 300mM KCl, pH 7.2) containing phenylmethylsulphonylflouride (PMSF, 1µM) for 30min in a humidity chamber (25°C). The binding buffer was removed and slides were incubated for 90mins with 100µL of binding buffer with BSA (1mg/mL), containing 100 µL [<sup>33</sup>P]-H2RLX. Slides for determining non-specific binding were incubated with H2RLX (1µM). Labelled sections were washed, briefly rinsed in distilled water, dried and apposed to a phosphorimager plates for 3 days with  $[\gamma^{-33}P]$ -ATP reference standards (Section 2.3.2.4). Plates were scanned with PhosphorImager SI<sup>TM</sup> (Molecular Dynamics, U.S.A.) and images were analysed with ImageQuaNT<sup>TM</sup> (Version 4.1; Molecular Dynamics, U.S.A.). Receptor expression images were expressed as arbitrary phosphorimager units that were converted into dpm/mm<sup>2</sup> using the  $[\gamma^{-33}P]$ -ATP reference standards. Uterus block numbers were then organised according to menstrual phase and data grouped using PRISM (GraphPad Inc., San Diego, U.S.A.).

# 7.2.2.3 Localisation of the relaxin receptor using high resolution autoradiography

Slide-mounted uterus sections were processed as described in Section 7.2.2.2. Slides were stored overnight in a vacuum-sealed container with silica gel before

Chapter 7: LGR7 expression in the human uterus

commencing the emulsion protocol (Section 2.3.2.5). LM-1 hype:coat emulsion (Amersham Pharmacia Biotech, U.S.A.) was incubated at 42°C (under darkroom conditions) for 1 hour and then transferred to a dipping chamber. Each slide was dipped into the emulsion twice, drained and allowed to dry in a light-proof box overnight. Slides were then transferred into a slide box, wrapped in foil to prevent exposure to light and stored at 4°C for 3 weeks. Prior to development, slides were equilibrated at room temperature for 1 hour. Emulsion was developed in Kodak D-19 developer (5mins), Kodak stop solution (1min), Kodak fix solution (10mins) and gently rinsed in running water (15mins). Slides were air-dried and examined under a dark-field microscope. For histological analysis, slides were stained with Gill's haematoxylin (distilled water 730mL, ethylene glycol 250mL, haematoxylin 2g, sodium iodate 17.6g and glacial acetic acid 20mL) and Eosin (1% solution).

# 7.2.3 Characterisation of [<sup>33</sup>P]-H2RLX binding in the proliferative and secretory phases.

Slide-mounted uterus sections from the proliferative and secretory phase were processed as described in Section 2.3.2.2. Briefly, slides were preincubated in 100µL of binding buffer containing PMSF (1µM) for 30min in a humidity chamber (25°C). The binding buffer was removed and slides were incubated for 90mins with 100µL of binding buffer with BSA (1mg/mL), containing 50-1500pM [<sup>33</sup>P]-H2RLX for saturation binding or 100pM [<sup>33</sup>P]-H2RLX in the absence or presence of increasing concentrations of unlabelled H2RLX or PRLX (10nM to 10µM) for competition binding. All non-specific binding was defined with H2RLX (1µM). Labelled sections were processed and analysed as described in Section 7.2.2.2. Saturation binding images were expressed as arbitrary phosphorimager units that were converted into dpm/mm<sup>2</sup> using [ $\gamma$ -<sup>33</sup>P]-ATP reference standards (Section 2.3.2.4), while competition binding was expressed as a percentage of total specific binding.

### 7.2.4 Analysis of relaxin, LGR7 and related gene expression

RT-PCR experiments were performed and analysed by Dr Chrishan Samuel and Ms Chongxin Zhao (Howard Florey Institute of Experimental Physiology and Medicine, Vic, Aust). RLX (H1 & H2), LGR7 and VEGF<sub>165</sub> gene expression were first assessed by RT-PCR. Analysis of H3RLX was not conducted in this study because there was no known human tissue that expressed H3RLX and hence no positive control cDNA. Total RNA (1µg) was used for synthesis of first strand cDNA in a 20µl reverse transcription reaction using a RT kit (Promega Corporation, Annandale, NSW, Australia) and random hexamer primers. Between 1-2µg of this cDNA template was used in touch-down PCR reactions (50µl volume) with 0.2U Tag DNA polymerase (Promega) and 100ng/µl of human gene-specific oligonucleotide primers (Geneworks Pty Ltd, Adelaide, Australia). Human LGR7 primers were designed from the human genome sequence (NCBI, Accession No. AF190500), whereas all other relaxin oligonucleotide primers were obtained from previously published sequences; H1RLX (Hudson et al., 1983), H2RLX (Hudson et al., 1984) and human VEGF<sub>165</sub> (Houck et al., 1991). The relaxin and LGR7 oligonucleotide primers used were designed to span intron-exon junctions to distinguish between PCR products derived from mRNA and those derived from contaminating DNA, while the VEGF<sub>165</sub> primers were kindly provided by Dr Elaine Unemori (Connetics Corporation, Palo Alto, CA, USA). All PCR reactions were carried out in a Perkin Elmer Gene Amplifier with an initial denaturation step at 94°C (3 minutes) and annealing temperatures as shown in Table 7.1. GAPDH was used in separate PCR reactions to control for quality and equivalent loading of the cDNA. Aliquots of the PCR products were electrophoresed on 2-4% (w/v) agarose gels stained with ethidium bromide and photographed.

Real-time PCR experiments were conducted and analysed by Dr Laura Parry and Ms Helen Gehring (University of Melbourne, Vic, Aust). Real-time PCR was used to quantify H2RLX, LGR7, VEGF<sub>165</sub> and VEGF<sub>121</sub> gene expression using human genespecific primers and FAM-labelled probes (Keystone Division, Biosource International, Foster City, CA) designed with the Primer Express<sup>TM</sup> computer

•;

.

#### Chapter 7: LGR7 expression in the human uterus

software (Applied Biosystems, Foster City, CA). The human VEGF primers and probes were kindly provided by Dr Elaine Unemori (Connetics Corporation). All primers were designed to span introns to distinguish between PCR products derived from mRNA and those derived from contaminating DNA. For each sample, lug total RNA was reverse transcribed in a 30µL reaction containing 1xTaqMan® buffer (5.5mM MgCl<sub>2</sub>, 500µM dNTPs, 2.5µM oligo d(T), 0.4µL RNase inhibitor and 1.25U/µL MultiScribe<sup>™</sup> reverse transcriptase; Applied Biosystems). For the LGR7 cDNA,  $2.5\mu$ M random hexamers replaced the oligo d(T). A second reaction mix using 30ng total RNA from each sample and a series of endometrium RNA dilutions (100-0.001ng) was prepared for the endogenous reference 18S ribosomal RNA (18S) PCR reactions and to generate the 18S standard curves respectively. First strand cDNA synthesis for all samples was carried out simultaneously at 25°C for 10min, 42°C for 45min and 95°C for 10min with a final cooling temperature a 4°C before storage at -20°C. The PCR reactions were conducted in triplicate using 96-well optical reaction plates (Applied Biosystems), in 25 µl volumes consisting of 1xTaqMan® Universal PCR Master Mix, 0.8 µM forward and reverse primers, 0.4 µM probe and 2.5 µl cDNA template. An ABI PRISM 7700 Sequence Detector (Applied Biosystems) was used with the following conditions: 50°C for 10 min, 95 °C for 10 min followed by 40 cycles of 95°C for 15 sec and 58°C for 1 min. Negative template samples were included in each plate as controls. The relative  $C_T$ standard curve method was used in this study, where  $C_T$  is the cycle number at which DNA amplification is first detected. In the relative standard curve method, specific gene and 18S (endogenous reference) expression are assessed in separate PCR reactions and the C<sub>T</sub> values for both genes are related to those of the endogenous reference standard curve. The amount of specific gene expressed in each sample is calculated by dividing the log (gene of interest concentration) by the log (18S concentration) using the following formula: log (gene concentration) =  $(C_T - b/a)$ where b = y-intercept of the standard curve and a = slope of the standard curve. In this experiment the regression line for the 18S standard curve was y = -3.398x +14.235, R<sup>2</sup>=0.998. The same 18S standards were used on every plate, with interplate variation 2.2 % (Ing standard, n=4 plates).

Table 7.1

Primers, probes and conditions used for RT/PCR

Primer Sequences:	Annealing	Product
	temperature(s)	size (bp)
	used:	
HIRLX forward primer:	58ºC(3 cycles),	250
5'GTCGCGGCCAAATGGAAGGA3'	57°C(3), 56°C(3),	
HIRLX reverse primer:	55ºC(31)	
5'TTGCCTCTCAGATAGGGCTGC3'		
H2RLX forward primer:	55°C(2), 54°C(2),	445
5°GAACCAATTITCCAGAGCAGTCG3'	53°C(2), 52°C(2),	
H2RLX reverse primer:	51°C(2), 50°C(30)	
5'TGTAGAGTTGTCTCTTTTTTCGAG3'		
Human LGR7 forward primer:	58°C(2), 57°C(2),	634
5'GCGCAGCTGTGGATGGAGAG3'	56°C(2), 55°C(2),	
Human LGR7 reverse primer:	54°C(2), 53°C(2),	
5'GGTCTTGTGGTCAGAGTATAG3'	52 <sup>9</sup> C(2), 51 <sup>0</sup> C(2),	
	54 <sup>0</sup> C(25)	
Human VEGF <sub>165</sub> forward primer:	59ºC(5), 58ºC(5),	140
5'CCAGCACATAGGAGAGATG3'	57 <sup>0</sup> C(30)	
Human VEGF <sub>165</sub> reverse primer:		
5'GGAACATTTACACGTCTGC3'		

ないたので、「ないの」で、「ないの」で、「ないの」で、「ない」で、「ない」で、「ない」で、「ない」で、「ない」で、「ない」で、「ない」で、「ない」で、「ない」で、「ない」で、「ない」ので、「ない」の

### 7.2.5 Statistical Analysis

Data was analysed as described in Section 2.8. Quantitative autoradiography results were analysed using one-way ANOVA with a Newman-Keuls post-test. The  $pK_D$ ,  $B_{max}$  and  $pK_i$  data was analysed using a Student's t- Test. Data for mRNA concentrations did not show homogeneity of variance and were log transformed before analysis by one-way ANOVA (SPSS 10.0). The Least Squares Difference method tested for significant differences at the 95% confidence level between stages of the menstrual cycle.

ļ
#### 7.3 Results

# 7.3.1 Changes in binding of [<sup>33</sup>P]-H2RLX with the phases of the menstrual cycle.

Slide mounted sections of uterus from the seven phases of the menstrual cycle were incubated with [<sup>33</sup>P]-H2RLX to determine receptor binding levels and to determine any changes that may occur during the menstrual cycle. Initial studies were conducted blind with a variety of sections from different phases of the cycle and showed strong binding in some sections but not others. Sections with high levels of binding were subsequently shown tc be from the secretory phase of the cycle and those with little binding from the proliferative phase. Systematic studies of uterine sections from all proliferative phases showed very little [<sup>33</sup>P]-H2RLX binding (EP 1.3  $\pm$  0.51, n=4; MP 2.1  $\pm$  0.39, n=5; LP 2.6  $\pm$  1.12 dpm/mm<sup>2</sup>, n=3; Figure 7.1), whereas those from the secretory phase showed a large amount of binding mainly confined to the endometrium (ES 11.6  $\pm$  2.33, n=5, P < 0.05 cf all proliferative phases; MS 10.0, n=2; LS 6.5  $\pm$  0.73, n=6; and M 5.6 dpm/mm<sup>2</sup>, n=2; Figure 7.2). Binding levels were markedly increased in the early secretory phase and then progressively declined towards the menstrual phase although these levels were still higher than in the early proliferative stage.

### 7.3.2 Localisation of the relaxin receptor in uterine sections using emulsion autoradiography

To identify the cells that express the relaxin receptor, [<sup>33</sup>P]-H2RLX labelled uterine sections were coated with silver grain hypercoat emulsion and exposed for 3 weeks. Low levels of silver grains appeared over the endometrium of sections from the E-MP phase with no significant silver grains overlying the myometrium (Figure 7.3a). In contrast, in sections from E-MS, an abundance of silver grains was highly localised to sites in the endometrium, with non-specific levels of silver grains associated with the myometrium (Figure 7.3b). Comparison of corresponding dark

field and light field images showed that the silver grains in the endometrium were localised to the glandular epithelial cells and the luminal epithelium.

# 7.3.3 Characterisation of [<sup>33</sup>P]-H2RLX binding in the proliferative and secretory phases.

The saturation and competition binding properties of [<sup>33</sup>P]-H2RLX to sections from the early-mid proliferative (E-MP) and early-mid secretory (E-MS) were studied to determine if the changes in intensity were related to an alteration in receptor number or affinity. Binding to sections from both the E-MP and E-MS phases was saturable to a single population of binding sites (Figure 7.4) which displayed no significant difference in the binding affinity (E-MP pK<sub>D</sub> 9.16 ± 0.17, n=4; E-MS pK<sub>D</sub> 9.16 ± 0.06, n=4). However, the number of binding sites in E-MS was significantly (P<0.05) higher (B<sub>max</sub> 160 ± 32; n=4) than in the E-MP sections (B<sub>max</sub> 15 ± 4; n=4).

Competition studies were conducted in uterine sections to examine the ability of H2RLX and PRLX to compete for [<sup>33</sup>P]-H2RLX binding. pEC<sub>50</sub> values were calculated from competition curves fitted using Prism and pK<sub>i</sub> values calculated using the Cheng & Prusoff (1974) equation. Sections from the E-MP showed that H2RLX (pK<sub>i</sub> 8.0  $\pm$  0.06, n=4) had a significantly higher affinity for the relaxin receptor in the endometrium than PRLX (pK<sub>i</sub> 7.4  $\pm$  0.21, n=4, P < 0.05). The same pattern was observed in the E-MS with H2RLX (pK<sub>i</sub> 8.3  $\pm$  0.11, n=4) having a significantly higher affinity for the receptor than PRLX (pK<sub>i</sub> 7.3  $\pm$  0.08, n=4, P < 0.05; Figure 7.5). There was no significant difference however in the affinity of either H2RLX or PRLX determined in the E-MP or E-MS phase.

:

Autoradiographic localisation of [<sup>33</sup>P]-H2RLX binding to slide-mounted sections of human uterus at 7 different phases of the menstrual cycle.

The left panels are representative slides of the total binding and right panels of nonspecific [<sup>33</sup>P]-H2RLX binding from each of the phases of the menstrual cycle.

М

## Chapter 7: LGR7 expression in the human uterus



Quantitation of the autoradiographic localisation of [<sup>33</sup>P]-H2RLX binding to slidemounted sections of human uterus at 7 different phases of the menstrual cycle.

The levels of specific binding to the endometrial layer as determined by quantitative autoradiography from each of the phases in the menstrual cycle (mean  $\pm$  SEM). The 7 phases of the menstrual cycle include: early proliferative (EP), mid proliferative (MP), late proliferative (LP), early secretory (ES), mid secretory (MS), late secretory (LS) and menstrual (M). Note the relatively low levels of binding in the proliferative stages, the significant increase that occurs at the start of the secretory phase (\* P<0.05 cf all proliferative stages) and the gradual reduction towards the menstrual phase.



### Chapter 7: LGR7 expression in the human uterus

Qualitative autoradiographic localisation of [<sup>33</sup>P]-H2RLX binding to slide mounted sections of human uterus using photographic emulsion.

Light field (LF) and dark field (DF) images of A) myometrium (M) and endometrium (E) showing silver grains abundant in the glandular epithelial cells of the endometrium (x2 magnification), and B) the endometrial luminal epithelium (x10 magnification).

L

B)

A)



Relaxin receptor characterisation determined using quantitative autoradiography saturation binding of [<sup>33</sup>P]-H2RLX to slide-mounted sections of human uterus.

Sections of uterus tissue were obtained from the A) proliferative (E-MP) and B) secretory stages (E-MS) of the cycle. Saturation binding isotherms show that the binding of [<sup>33</sup>P]-H2RLX is saturable to a single population of sites



Chapter 7: LGR7 expression in the human uterus

Relaxin receptor characterisation determined using quantitative autoradiography competition binding of H2RLX and PRLX to slide-mounted sections of human uterus.

Sections of uterus tissue were obtained from the A) proliferative (E-MP) and B) secretory stages (E-MS) of the cycle. Competition binding isotherms show competition for [<sup>33</sup>P]-H2RLX binding by H2RLX and PRLX O.



Chapter 7: LGR7 expression in the human uterus



#### 7.3.4

H1RLX gene transcripts were only weakly expressed in two samples of the proliferative phase of the cycle (Figure 7.6). Gene transcripts of H2RLX were detected at relatively weak but consistent levels in both the proliferative and secretory stages of the cycle and appear to be the predominant form of relaxin in the human endometrium. Of the two bands observed, the 445bp band corresponds to the expected product for H2RLX while the 330bp band represents an unrelated fragment of DNA. H2RLX was highly expressed in the corpus luteum as expected (Figure 7.6), however no direct comparison can be made between the endometrium and the ovarian samples as different amounts of RNA were used in the first strand cDNA synthesis.

An LGR7 gene transcript (634bp) was detected in the human endometrium, with stronger expression observed in those samples obtained from the secretory phase of the cycle (Figure 7.6). VEGF<sub>165</sub> gene expression was also higher in the secretory phase of the cycle, with seemingly stronger expression observed in the late secretory endometrium (Figure 7.6). An additional positive control was not included as the endometrium is generally considered to be a major source of VEGF.

Quantitative analysis by real-time PCR confirmed that there was a significant (P<0.01 ANOVA) increase in the endometrial LGR7 mRNA concentrations in the early secretory phase compared with all other stages of the cycle (Figure 7.7a). Although LGR7 gene expression decreased in the mid-secretory endometrium, mRNA concentrations remained significantly (P<0.05) higher in the mid secretory, late secretory and menstrual phases compared with the proliferative phase of the cycle. This pattern of LGR7 gene expression paralleled that of RLX binding in the endometrium assessed by quartitative autoradiography (Figure 7.2). The real-time PCR for H2RLX demonstrated a low level of gene expression in the endometrium with C<sub>T</sub> values of between 38-40 (data not shown). Thus, it was not possible to show differences for H2RLX between the phases of the cycle.

#### Analysis of relaxin, LGR7 and related gene expression

Chapter 7: LGR7 expression in the human uterus A comparison of VEGF<sub>165</sub> and VEGF<sub>121</sub> mRNA concentrations in the endometrium throughout the cycle showed an identical pattern of gene expression for both isoforms (Figure 7.7b & c). Concentrations were low in the proliferative phase of the cycle and did not differ significantly (P<0.05) compared with early and mid secretory endometrium. Expression of both VEGF isoforms increased significantly (P<0.05) in the late secretory phase and remained high in the menstrual phase.

Expression of RLX (H1 & H2), LGR7 and VEGF<sub>165</sub> mRNA using RT-PCR on RNA extracted from the human endometrium at various phases of the menstrual cycle.

The first lane is the molecular weight marker, followed by samples from separate patients in the proliferative phase (lanes 1-5), early secretory phase (lanes 6-8), mid secretory phase (lanes 9-11), late secretory phase (12-14) and menstrual phase (lanes 15-16) of the cycle. RNA from human prostate was used as a positive control for H1RLX PCR, human corpus luteum for H2RLX PCR and THP-1 cells for the LGR7 PCR (all lane 17). Water replaced the cDNA in negative control reactions for each PCR (lane 18).

Data obtained from Dr C Samuel, Howard Florey Institute of Experimental Physiology and Medicine, Vic, Aust.





A) LGR7, B) VEGF<sub>165</sub> and C) VEGF<sub>121</sub> mRNA concentrations in the endometrium at five phases of the menstrual cycle.

Data are the mean  $\pm$  SEM gene/18S ratio x 10<sup>3</sup>. (a) significantly (P<0.05) higher compared with all other phases; (b) significantly (P<0.05) higher compared with the proliferative phase; (c) significantly (P<0.05) higher compared with the proliferative phase and early secretory phase.

Data obtained from Dr L Parry, Dept of Zoology, University of Melbourne, Vic, Aust.



1

#### 7.4 Discussion

The aim of the present study was to investigate the sites of expression and the characteristics of the relaxin receptor in the human uterus at different phases throughout the menstrual cycle. While binding sites for biotinylated PRLX have previously been identified in the human uterus (Kohsaka *et al.*, 1998), until recently no studies had been carried out to identify changes in relaxin receptor expression during the menstrual cycle. The relaxin peptide has previously been identified in human endometrial tissue in the secretory phase however no relaxin peptide was detectable in the proliferative phase (Yki-Jarvinen *et al.*, 1983). Relaxin has been shown to differentially influence VEGF secretion in isolated uterine cells taken both from the proliferative and secretory phases (Palejwala *et al.*, 2002), although it was unclear whether the differential regulation occurs at the level of the receptor or the signalling cascade.

The results reported in this study indicate that expression of the relaxin receptor varies in a cyclical manner through the menstrual cycle. A low level of receptor binding was demonstrated during the proliferative stages with a dramatic increase in binding at the time of ovulation. The high levels of binding were maintained during the early secretory (days 15-18 of a standardised 28 day cycle) and mid secretory phase (days 19-24) and then decreased through the late secretory and menstrual chases (days 25-28). Results from the gene expression studies supported the autotadiography and showed a 13-fold increase in LGR7 mRNA following ovulation, which subsequently decreased in the mid-late secretory and menstrual stages of the cycle. The more rapid decline in the LGR7 mRNA compared to the binding probably reflects a more rapid turnover of RNA than receptor protein. This is the first study to identify a cyclic expression of the relaxin receptor in the human Two studies have appeared while our work was under review. uterus. Immunohistochemical staining of endometrial sections from 3 patients showed little if any staining in sections from patients in the proliferative or early secretory phase but intense staining in one patient in the mid-secretory phase (Ivell et al., 2003). Although the other study reported no consistent regulation of LGR7 mRNA during the phases of the menstrual cycle (Luna et al., 2004) the expression levels are 189

expressed as a ratio with GAPDH which is known to be sex-steroid dependent (Wu et al., 2003). Little detail is provided about the patients recruited for these studies, however all patients were normally ovulating women not suffering from endocrine complaints and tissue samples collected were screened by histological examination, similar to the criteria followed in our study, therefore it is unlikely that the conflicting results are due to patient selection. In the present study the quantitative PCR was carried out using 18S RNA as the standard. Given our finding that relaxin receptor is elevated in the secretory phase, and the observation that plasma relaxin levels increase from day 5 to peak at day 10 post ovulation (Stewart et al., 1990), it is possible that a role exists for relaxin in the process of embryo implantation. Analysis of gene expression of the relaxin peptides indicated that the endometrium is not a major source of circulating relaxin, with low to undetectable levels of relaxin mRNA present throughout all stages of the menstrual cycle. Our data are in agreement with recent work (Palejwala et al., 2002) that demonstrated expression of relaxin in the human endometrium using a nested PCR method. The major source of circulating relaxin is the corpus luteum, since it is highly expressed in this tissue (Ivell et al., 1989), high levels of the hormone are found in the ovarian vein (Weiss et al., 1976) and women without ovaries who become pregnant by oocyte donation have no circulating relaxin (Emmi et al., 1991; Johnson et al., 1991). Local production of relaxin from the endometrium may therefore play a significant role in the establishment or maintenance of early pregnancy.

The binding characteristics of the relaxin receptor remained unaltered between the proliferative and the secretory phases, with the  $pK_D$  values for the [<sup>33</sup>P]-H2RLX ligand and the  $pK_i$  values for H2RLX and PRLX being not significantly different with the changing stages of the cycle. The values determined were comparable with those obtained with previous binding studies using THP-1 cells (Tan *et al.*, 2001a), that endogenously express the human relaxin receptor (Parsell *et al.*, 1996), or human uterine cells (Osheroff *et al.*, 1995; Palejwala *et al.*, 1998) indicating that THP-1 cells and the uterus express the same receptor. The consistency in the binding characteristics between the different menstrual cycle phases indicates that the variations in relaxin function during the menstrual cycle, such as its potential influence on VEGF secretion (Palejwala *et al.*, 2002), are not due to changes in receptor conformation or affinity of the relaxin receptors expressed. Relaxin 190

receptor expression in the endometrium during the secretory phase was confined specifically to glandular and luminal epithelial cells. The binding of [<sup>33</sup>P]-H2RLX was appropriate for binding to the relaxin receptor since it was competitively inhibited by both H2RLX and PRLX with high affinity. In contrast to previous studies that showed binding of biotinylated PRLX to blood vessels and myometrium in addition to glandular and luminal epithelium (Kohsaka et al., 1998), the present studies provided no evidence for binding of [<sup>33</sup>P]-H2RLX to these tissues. The differing results may relate to the concentrations of labelled relaxin used in the two studies. In the present study we used  $[^{33}P]$ -H2RLX at a concentration of 100pM, similar to the known pK<sub>D</sub> value for relaxin at the receptor (Tan et al., 1999) whereas the study using biotinylated PRLX used a concentration of approx 0.7µM, several orders of magnitude higher than the  $pK_{D}$ . It is possible that the use of high labelling concentrations could result in binding to low affinity sites such as the related LGR8 receptor which is known to bind relaxin (Hsu et al., 2002). However the lack of binding to stromal cells was surprising given the demonstration of functional responses in these cells in culture (Bell et al., 1991; Einspanier et al., 2001; Palejwala et al., 2002). The reason for the difference may relate to the different conditions experienced by cells in situ and those in culture since human endometrial stromal cells fail to produce a cAMP response to relaxin when first isolated but the response appears gradually over the first 11 days of culture and then increases dramatically from 11-30 days (Tseng et al., 1995). As with stromal cells, studies with human myometrial cell lines (PHM1-41) isolated from term pregnant myometrium show clear evidence of relaxin stimulated cAMP accumulation indicating the presence of relaxin receptors (Dodge et al., 1998). This may also relate to the expression of relaxin receptors in an immortalised cell line under cell culture conditions. In contrast, functional studies of the effects of relaxin in the human uterus show that human myometrium is much less able to respond in vitro to relaxins than either pig or rat myometrium. Porcine relaxin had little relaxant effect on myometrium from pregnant or non-pregnant women (MacLennan et al., 1986a). Later studies with recombinant H2RLX showed little or no relaxant effect on human myometrium excised at hysterectomy or at caesarian section (MacLennan et al., 1991) or on uterine fundus or isthmus at term (Petersen et al., 1991). Studies of spontaneous contractions of human myometrium demonstrated small reductions in

amplitude to relaxin-rich extracts of corpus luteum (Szlachter *et al.*, 1980) or of amplitude and frequency to H2RLX in some oestrogen-primed tissues (MacLennan *et al.*, 1991). This is in direct contrast to studies in pig myometrium or rat uterus where complete inhibition of contraction was observed with H2RLX, PRLX and H1RLX (MacLennan *et al.*, 1991; MacLennan *et al.*, 1995; MacLennan *et al.*, 1986b; Petersen *et al.*, 1991). The lack of significant [<sup>33</sup>P]-H2RLX binding to relaxin receptors in the human myometrium reported here is in accord with the poor relaxant responses to relaxin observed in this tissue.

Immunohistochemical studies demonstrated relaxin in endometrium in the secretory, but not in the proliferative phase nor in myometrium from either phase (Yki-Jarvinen et al., 1983). The hormone has also been localised to decidualised stromal cells in the late secretory phase and in early pregnancy (Palejwala et al., 2002). Recent studies have established that relaxin is synthesised in the endometrium since relaxin mRNA was present in cultured endometrial stromal and glandular epithelial cells with stronger bands seen in cells from the secretory phase and immunoreactive relaxin was detected in culture medium bathing the cells (Palejwala et al., 2002). As there was a lack of evidence for relaxin receptors a sociated with the endometrial vasculature, our study does not support a direct role for relaxin in regulating endometrial vascular function. However, relaxin does increase VEGF expression in glandular epithelial and stromal cells from the secretory phase (Palejwala et al., 2002) suggesting that it may have a role in endometrial vascularisation. The significant increase in relaxin binding in the early secretory phase, and the retention of the elevated binding throughout the secretory phase, suggests a role for relaxin in implantation. Support for this hypothesis comes from a study showing that high levels of relaxin produced in granulosa lutein cell cultures are associated with 100% implantation success rates with IVF, while low levels resulted in an implantation rate of 13% (Stewart et al., 1999). Although this would indicate that the production of relaxin is linked to successful implantation, circulating relaxin is clearly not obligatory for this process since implantation can occur in women without ovaries (Emmi et al., 1991; Johnson et al., 1991). This may indicate that local relaxin may be important, that it is a marker for other factors important in implantation or merely that it is a marker of oocyte quality. VEGF is known to have an important role in the proliferation and vascularization of the uterus lining and while relaxin has been

į

 ų,

4

found to stimulate VEGF secretion in glandular epithelial and stromal cells (Palejwala *et al.*, 2002), the VEGF responsible for uterus growth has been identified in intravascular neutrophils rather than from the glandular epithelial cells (Gargett *et al.*, 2001). The increase in VEGF<sub>165</sub> and VEGF<sub>121</sub> gene expression during the midlate secretory phases supports previous reports showing an increase in VEGF mRNA during this stage of the cycle (Sugino *et al.*, 2002).

In conclusion, we have localised binding sites for relaxin in the human uterus and have demonstrated that they are primarily confined to the glandular and luminal epithelium in the secretory stage of the cycle. Furthermore, the levels of relaxin receptor expression change dramatically during the different phases of the menstrual cycle. The binding characteristics of the receptor were appropriate for binding to the relaxin receptor (LGR7) and were unchanged with differing expression levels. Thus any difference in relaxin function between the phases is a reflection of changes in the intracellular properties of the receptor and not due to changes in receptor characteristics. The location and regulation of relaxin receptor expression that occur during the menstrual cycle suggest a role in the implantation process, although further studies are needed to confirm this hypothesis.

#### 7.5 Acknowledgements

I would like to acknowledge the contribution of; Assoc Prof Peter Rogers and Ms Fiona Ledermann (Centre for Women's Health Research, Monash University, Vic, Aust) for the provision of human uterus samples used in the study, Drs Caroline Gargett and Gareth Weston (Centre for Women's Health Research, Monash University, Vic, Aust) for the provision of the human endometrial mRNA samples, Dr Chrishan Samuel and Ms Chongxin Zhao (Howard Florey Institute of Experimental Physiology and Medicine, Vic, Aust) for conducting and analysing the RT-PCR experiments, Dr Laura Parry and Ms Helen Gehring (University of Melbourne, Vic, Aust) for conducting and analysing the real-time PCR experiments.

This study is currently under review for publication.

: :

: ;

ļ

# **Chapter Eight**

# **General Discussion**

#### 8.1 General Discussion

The interaction of relaxin with its receptor has been, and continues to be, extensively studied in experimental animals such as mice and rats. While such investigations are vital for obtaining information regarding the physiological and pharmacological properties of relaxin, the investigation of these properties at the human relaxin receptor is also important. The work presented in this thesis examines the pharmacological properties of the human relaxin receptor by investigating the interaction with various relaxin analogues in binding and cAMP assays using cell lines endogenously and recombinantly expressing the human relaxin receptor. Investigation into the properties of the physiologically expressed relaxin receptor were also explored in human uterine tissue, with the aim of gaining further insight into the role of relaxin in human reproduction.

The lack of available human tissue for research no longer restricted investigations at the human relaxin receptor as the characterisation of the human monocyte cell line, THP-1, showed endogenous expression of the relaxin receptor (Parsell *et al.*, 1996). The successful development of a high through-put binding assay (Chapter 3) allowed examination of the interaction between various relaxin analogues and the human relaxin receptor. The greatest advantage of using this binding assay over other assays was that very little peptide was needed to complete a binding study, an important factor when only limited amounts of the synthetic relaxin analogues were available.

Work presented in this thesis has shown several factors important in determining relaxin-like affinity for the human relaxin receptor (Chapter 3). The extended length of the relaxin B chain does not influence affinity, however shortening of the relaxin B chain reduces affinity for the relaxin receptor, similar to the reduced efficacy displayed at the rat relaxin receptor (Tan *et al.*, 1998). The reduction in affinity is likely due to instability of the structural integrity of the peptide rather than the loss of particular amino acids. Shortening the B chain to within one amino acid of the cysteine important in the formation of a linking disulphide bond may cause peptide

A second s

1

-

instability, similar to that previously shown with the substitution of amino acids adjacent to a disulphide bond (Büllesbach *et al.*, 1995). Another factor important in determining relaxin-like affinity is the relaxin B chain binding motif (Büllesbach *et al.*, 2000), however this does not solely determine affinity as shown by the INSL3 analogues synthesised with the relaxin binding motif. Affinity for the relaxin receptor was increased by the insertion of the binding motif into the INSL3 peptide, however this interaction was still very weak, indicating either the absence of critical factors yet to be determined or the presence of unknown factors in the INSL3 peptide that hinder interaction with the relaxin receptor. Unfortunately due to the limited amounts of the peptides synthesised, examination of the relaxin and INSL3 analogues in a functional study could not be completed at this time.

Further analysis of the structural-activity properties of relaxin was also hindered by the inability to synthesise novel relaxin analogues, modified to explore new areas. In an attempt to overcome this obstacle, a mammalian recombinant system was developed with the aim of producing various relaxin analogues modified by molecular cloning techniques (Chapter 4). While the recombinant system appeared to synthesis the pro-HRLX peptide, it was not successfully isolated from the cell debris following lysis of the expressing cells. Due to my lack of experience as a peptide chemist, there was a possibility that further attempts to successfully isolate a bioactive recombinant relaxin protein would be very time consuming and little research would be accomplished by the completion of the thesis candidature. Therefore, further development of a recombinant system was temporarily cancelled until completion of this thesis.

The identification of the relaxin receptor as the orphan LGR7 receptor (Hsu *et al.*, 2002) midway through the candidature of this thesis allowed the specific interaction between relaxin and the human relaxin receptor to be investigated in a recombinant system. An important advantage of investigating receptor pharmacology in a cell line recombinantly expressing the relaxin receptor, rather than a cell line that naturally expresses the receptor, is the ability to alter to the expression level of the LGR7 receptor. Increasing the expression of LGR7 above the naturally expressed receptors in the cell line allows a specific examination of the interaction between

ł

ł

relaxin and LGR7 to be completed with minimal interference by other receptors. This was particularly noticeable when comparing the ability of H3RLX to displace  $[^{33}P]$ -H2RLX binding to the relaxin receptor expressed in THP-1 cells, to the LGR7 receptor transiently expressed in CHO K1 cells. While appearing to displace all radioligand binding in the CHO-LGR7 system (Chapter 5), H3RLX could only displace approximately 50% of binding to THP-1 cells (Chapter 3), possibly indicating that H3RLX could not compete for  $[^{33}P]$ -H2RLX binding to another receptor expressed in THP-1 cells. THP-1 cells have been shown to express both LGR7 and LGR8 receptor mRNA transcripts (Nguyen *et al.*, 2003) and while our radioligand displays high affinity for both receptors (Chapter 6), H3RLX displays no affinity for LGR8 (Sudo *et al.*, 2003). Although RatRLX also displayed no affinity for LGR8 (Chapter 3), however experiments using higher concentrations of RatRLX are needed to investigate the maximal displacement of  $[^{33}P]$ -H2RLX binding in THP-1 cells.

Even though INSL3 has the highest homology of the insulin-like peptides to relaxin, studies have consistently reported that INSL3 displays no significant affinity for the relaxin receptor (Claasz et al., 2002; Smith et al., 2001; Tan et al., 2002) (Chapter 3, 5 & 6). In light of these results, the report that relaxin displayed high affinity for the INSL3 receptor was surprising (Bogatcheva et al., 2003; Hsu et al., 2002). Closer analysis of relaxin analogue interaction at LGR7 and LGR8 (Chapter 6) revealed the ability to interact at LGR8 is not universally shared by all relaxin analogues. H2RLX and PRLX displayed high affinity for both receptors (Hsu et al., 2002; Sudo et al., 2003), while RatRLX and H3RLX (Sudo et al., 2003) failed to display affinity for the LGR8 receptor. RMRLX was able to bind to and activate LGR8, however it preferentially interacted with the LGR7 receptor. Work presented in this thesis showed that relaxin analogue binding to LGR7 occurred at a single population of sites while binding to LGR8 appeared to be a two site interaction. Analysis of chimeric LGR7/LGR8 receptors also reported relaxin interacting at two LGR8 binding sites. This was shown by differences in affinity of relaxin analogues for the separated LGR8 ectodomain and transmembrane domain in the chimeric receptors (Halls, 2003), leading to the hypothesis that the higher affinity site is positioned in

the ectodomain while the lower affinity site is in the transmembrane domain. The generation of chimeric receptors that contain only the separate LGR8 ecto- and transmembrane domains would allow closer examination of the two hypothesised LGR8 binding sites. Examination of the specific binding of relaxin and INSL3 at LGR8 with an INSL3 radioligand may also reveal further differences in the ability of these peptides to bind to the LGR8 receptor.

Relaxin analogues are not only interacting differently with the LGR8 receptor, the interaction between relaxin analogues and the LGR7 receptor also appears to be an individualised relationship. Further manipulation of the ecto- and transmembrane domains of the LGR7/8 chimeric receptor to include each LGR7 extracellular loop separately revealed that the presence of extracellular loop 2 is important for the interaction between H3RLX and the LGR7 receptor (Sudo et al., 2003). This was not found to be important however in the interaction with H2RLX and PRLX analogues. Signalling through LGR7 also appeared to be agonist-directed as H2RLX, PRLX and RMRLX analogues stimulated a biphasic cAMP response while the response to RatRLX was monophasic (Chapter 6). Although the cAMP response to relaxin is reportedly a biphasic response comprised of a G<sub>S</sub> protein- and a PI3Kcoupled pathway, this is the first study to show a biphasic cAMP dose-response curve. The ability of relaxin to stimulate the two pathways to produce a cAMP response could also explain the lower efficacy displayed by INSL3 at LGR8 while maintaining the highest potency, as INSL3 may not be able to direct coupling of the receptor to both pathways. Investigation of the cAMP response in the presence of inhibitors to each component is needed to examine the coupling directed by the individual relaxin analogues at LGR7, as well as the difference between relaxin and INSL3 signalling at LGR8.

As the pharmacological differences between the H2RLX and H3RLX analogues become more apparent, the physiological function of these peptides is becoming an intriguing issue. A detailed examination of the expression of relaxin and its receptor in human uterine tissue (Chapter 7) has convincingly shown for the first time the cyclic expression of the relaxin receptor in the endometrial layer of human uterine tissue, with the LGR7 receptor being highly expressed during the early-mid

1

4

secretory phase of the menstrual cycle. Although reports have shown differences in relaxin activity (Palejwala et al., 2002) and receptor expression (Ivell et al., 2003; Luna et al., 2004) between the proliferative and secretory phases, solid evidence that the role of relaxin is influenced by the menstrual cycle has been lacking. The localisation and increase in expression of the relaxin receptor in the glandular and luminal epithelial cells indicates a possible role in implantation. These cells are known to maintain an important role in this process, as many of the factors that influence implantation are produced and secreted via the glands during the secretory phase, and the first contact between the embryo and the uterus takes place via the Implantation is a complex process involving not only luminal epithelium. reproductive and growth hormones but also cytokines and adhesion molecules (Lindhard et al., 2002). Cellular adhesion molecules expressed by the luminal epithelium may play a critical role in the initial attachment of the embryo to the endometrial surface (Enders et al., 1969). Reproductive hormones influence the regulation of adhesion molecules, including glycodelin, an endometrial product of the secretory phase. Progesterone is involved in glycodelin production, and its secretion is regulated by relaxin (Stewart et al., 1997). Heparin binding-epidermal growth factor (HB-EGF) is also an important adhesion factor that facilitates blastocyst implantation (Raab et al., 1996). HB-EGF is expressed in the glandular epithelial cells of the endometrium at increasing levels during the implantation window (Leach et al., 1999), and this pattern was reproduced in an endometrial cell line by combined treatment with oestrogen, progesterone and relaxin (Lessey et c') 2002). HB-EGF expression has been linked to the regulation of other key endometrial factors involved in implantation, including leukaemia inhibitory factor (LIF) and integrin  $\alpha_{v}\beta$ 3, highlighting the complex relationship between implantation factors. Further examination of the relationship between relaxin and various implantation factors could lead to valuable information regarding the specific role relaxin plays in implantation.

The examination of mRNA in the endometrium has revealed local synthesis of H2RLX mRNA while there has been no successful detection of either H1RLX or H3RLX transcripts (Palejwala *et al.*, 2002; Chapter 7). H3RLX mRNA expression was examined in this study and not detected, however as there is currently no human

ų,

Ą

ļ

tissue known to express H3RLX, a positive control for the RT-PCR could not be used and therefore the validity of these results is uncertain. A current hypothesis is that while both relaxin peptides have the potential to function through-out the body, H2RLX is the principle systemic acting relaxin peptide while H3RLX is primarily a centrally acting relaxin peptide. The high affinity displayed by the GPCR135 (Liu *et al.*, 2003b) and GPCR142 (Liu *et al.*, 2003a) receptors for H3RLX, while appearing to show no affinity for H2RLX, confuses this theory slightly as although GPCR135 is abundant in various brain regions with limited expression reported in peripheral tissues, GPCR142 is widely expressed peripherally. A possible explanation is that H3RLX is not the native ligand for GPCR142. Research in rats has shown that expression of Rat3RLX is primarily restricted to the brain (Burazin *et al.*, 2002) and as rats are not capable of expressing a functional GPCR142 receptor (Liu *et al.*, 2003a) while expressing a fully functional Rat3RLX peptide, the relevance of GPCR142 in relaxin gene 3 function is uncertain.

In conclusion, relaxin-like activity at the human relaxin receptor is determined by several factors, including the structural integrity of the peptide and the presence of the relaxin binding motif, however there are other areas important for interaction with the relaxin receptor still to be identified. The interaction between relaxin analogues and a variety of receptors is distinct to each analogue, as shown by the ability with which the various relaxin analogues bind to and activate different receptors. Although the wide variety of receptors that interact with relaxin is confusing in terms of functional relevance, the location and regulation of LGR7 receptor expression in human uterine tissue provides evidence for a role in implantation. Further investigation into the interaction between the human relaxins and the various receptors, as well as their relative expression throughout the body, is needed to provide insight into the specific physiological roles of the individual peptides.

# **Chapter Nine**

## References

- ABRAMOWITZ, A.A., MONEY, W.L., ZARROW, M.X., TALMAGE, R.V.N., KLEINHOLZ, L.H. & HISAW, F.L. (1944). Preparation, biological assay and properties of relaxin. In *Preparation and Assay* of *Relaxin*. pp. 103-114.
- ADHAM, I.M., EMMEN, J.M. & ENGEL, W. (2000). The role of the testicular factor INSL3 in establishing the gonadal position. *Molecular Cell Endocrinology*, **160**, 11-16.
- ANDERSON, M.B. (1984). Bioassay of porcine relaxin components by in vitro inhibition of rat uterine contractions. *Endocrinology*, **114**, 364-368.
- BANI, G., BANI-SACCHI, T., BIGAZZI, M. & BIANCHI, S. (1988). Effects of relaxin on the microvasculature of mouse mammary gland. *Histology & Histopathology*, 3, 337-343.
- BANI-SACCHI, T., BIGAZZI, M., BANI, D., MANNAIONI, P.F. & MASINI, E. (1995). Relaxin-induced increased coronary flow through stimulation of nitric oxide production. *British Journal of Pharmacology*, 116, 1589-1594.
- BARTSCH, O., BARTLICK, B. & IVELL, R. (2001). Relaxin signalling links tyrosine phosphorylation to phosphodiesterase and adenylyl cyclase activity. *Molecular Human Reproduction*, 7, 799-809.
- BATHGATE, R.A., SAMUEL, C.S., BURAZIN, T.C., LAYFIELD, S., CLAASZ, A.A., REYTOMAS, I.G., DAWSON, N.F., ZHAO, C., BOND, C.P., SUMMERS, R.J., PARRY, L.J., WADE, J.D. & TREGEAR, G.W. (2002). Human relaxin gene 3 (H3) and the equivalent mouse relaxin (M3) gene. Novel members of the relaxin peptide family. *Journal of Biological Chemistry*, 277, 1148-57.
- BATHGATE, R.A.D., MONIAC, N., BARTLICK, B., CLAASZ, A.A., DAWSON, N.F., TAN, Y.Y., WADE, J.D., TREGEAR, G.W. & IVELL, R. (2001). The relaxin-like factor (insulin3) is highly expressed in the ruminant ovary: a putative ruminant relaxin? In *Relaxin 2000: Proceedings* of the Third International Conference on Relaxin & Related Peptides. eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. pp. 349-356. Netherlands: Kluwer Academic Publishers.
- BEDARKAR, S., TURNELL, W.G., BLUNDELL, T.L. & SCHWABE, C. (1977). Relaxin has conformational homology with insulin. *Nature*, 270, 449-451.
- BELL, S.C., JACKSON, J.A., ASHMORE, J., ZHU, H.H. & TSENG, L. (1991). Regulation of insulin-like growth factor-binding protein-1 synthesis and secretion by progestin and relaxin in long term cultures of human endometrial stromal cells. *Journal of Clinical Endocrinology & Metabolism*, 72, 1014-1024.
- BOGATCHEVA, N.V., TRUONG, A., FENG, S., ENGEL, W., ADHAM, I.M. & AGOULNIK, A.I. (2003). GREAT/LGR8 is the only receptor for insulin-like 3 peptide. *Molecular Endocrinology*, 17, 2639-2646.
- BOGIC, L.V., YAMAMOTO, S.Y., MILLAR, L.K. & BRYANT-GREENWOOD, G.D. (1997). Developmental regulation of the human relaxin genes in the decidua and placenta: overexpression in the preterm premature rupture of the fetal membranes. *Biology of Reproduction*, **57**, 908-920.

**Chapter 9: References** 

- BOLTON, A.E. & HUNTER, W.M. (1973). The labelling of proteins to high specific radioactivities by conjugation to a <sup>125</sup>I-containing acylating agent. *Biochemical. Journal*, **133**, 529-539.
- BONGERS-BINDER, S., BURGARDT, A., SEEGER, H., VOELTER, W. & LIPPERT, T.H. (1991). Distribution of immunoreactive relaxin in the genital tract and in the mammary gland of nonpregnant women. Clinical & Experimental Obstetrics & Gynecology, 18, 161-164.
- BRENNAND, J.E., CALDER, A.A., LEITCH, C.R., GREER, I.A., CHOU, M.M. & MACKENZIE, I.Z. (1997). Recombinant human relaxin as a cervical ripening agent. British Journal of Obstetrics & Gynaecology, 104, 775-780.
- BRENNER, S.H., LESSING, J.B., SCHOENFELD, C., AMELAR, R.D., DUBIN, L. & WEISS, G. (1984a). Stimulation of human sperm cervical mucus penetration in vitro by relaxin. Fertility & Sterility, 47, 714-716.
- BRENNER, S.H., LESSING, J.B. & WEISS, G. (1984b). The effect of *in vivo* progesterone administration on relaxin-inhibited rat uterine contractions. *American Journal of Obstetrics & Gynecology*, 148, 946-950.
- BRYANT-GREENWOOD, G.D., RUTANEN, E.M., PARTANEN, S., COELHO, T.K. & YAMAMOTO, S.Y. (1993). Sequential appearance of relaxin, prolactin and IGFBP-1 during growth and differentiation of the human endometrium. *Molecular & Cellular Endocrinology*, 95, 23-29.
- BULLESBACH, E.E., GOWAN, L.K., SCHWABE, C., STEINETZ, B.G., O'BYRNE, E. & CALLARD, I.P. (1986a). Isolation, purification, and the sequence of relaxin from spiny dogfish (Squalus acanthias). *European Journal of Biochemistry*, 161, 335-341.
- BÜLLESBACH, E.E. & SCHWABE, C. (1986b). Preparation and properties of porcine relaxin derivatives shortened at the amino terminus of the A-chain. *Biochemistry*, 25, 5998-6004.
- BULLESBACH, E.E. & SCHWABE, C. (1987). Relaxin structure: Quasi allosteric effect of the NH<sub>2</sub> terminal A-chain helix. *Journal of Biological Chemistry*, **262**, 12496-12501.
- BULLESBACH, E.E. & SCHWABE, C. (1988). On the receptor binding site of relaxins. International Journal of Peptide & Protein Research, 32, 361-367.
- BÜLLESBACH, E.E. & SCHWABE, C. (1990). Monobiotinylated relaxins: preparation and chemical properties of the mono(biotinyl-aminohexanoyl) porcine relaxin. International Journal of Peptide & Protein Research, 35, 416-423.
- BÜLLESBACH, E.E. & SCHWABE, C. (1991). Total synthesis of human relaxin and human relaxin derivatives by solid-phase peptide synthesis and site-directed chain combination. *Journal of Biological Chemistry*, 266, 10754-10761.
- BULLESBACH, E.E., YANG, S. & SCHWABE, C. (1992). The receptor-binding site of human relaxin II: a dual prong-binding mechanism. *Journal of Biological Chemistry*, **267**, 22957-22960.
- BÜLLESBACH, E.E. & SCHWABE, C. (1993). Mouse relaxin: synthesis and biological activity of the first relaxin with an unusual crosslinking pattern. *Biochemical and Biophysical Research Communications*, **196**, 311-319.
- BULLESBACH, E.E. & SCHWABE, C. (1994a). Functional importance of the A chain loop in relaxin and insulin. *The Journal of Biological Chemistry*, **269**, 13124-13128.

- BÜLLESBACH, E.E., STEINETZ, B.G. & SCHWABE, C. (1994b). Synthesis and biological properties of guinea pig relaxin. *Endocrine Journal*, 2, 1115-1120.
- BÜLLESBACH, E.E. & SCHWABE, C. (1995a). Structural contribution of the A-chain loop in relaxin. International Journal of Peptide & Protein Research, 46, 238-243.
- BÜLLESBACH, E.E., STEINETZ, B.G. & SCWABE, C. (1995b). On the relaxin receptor. In Progress in Relaxin Research: The proceedings of the second international congress on the homone relaxin. eds MacLennan., A.H., Tregear, G.W. & Bryant-Greenwood, G.D. pp. 145-158. Singapore: World Scientific Publishing.
- BULLESBACH, E.E. & SCHWABE, C. (1996). The chemical synthesis of rat relaxin and the unexpectedly high potency of the synthetic hormone in the mouse. *European Journal of Biochemistry*, 241, 533-537.
- BULLESBACH, E.E. & SCHWABE, C. (2000). The relaxin receptor-binding site geometry suggests a novel gripping mode of interaction. *Journal of Biological Chemistry*, **275**, 35276-35280.
- BURAZIN, T.C., BATHGATE, R.A., MACRIS, M., LAYFIELD, S., GUNDLACH, A.L. & TREGEAR, G.W. (2002). Restricted, but abundant, expression of the novel rat gene-3 (R3) relaxin in the dorsal tegmental region of brain. *Journal of Neurochemistry*, 82, 1553-1557.
- BURAZIN, T.C.D., DAVERN, P.J., MCKINLEY, M.J. & TREGEAR, G.W. (2001). Identification of relaxin and relaxin responsive cells in the rat brain. In *Relaxin 2000: Proceedings of the Third International Conference on Relaxin and Related Peptides.* eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. pp. 209-214. Netherlands: Kluwer Academic Publishers.
- CANOVA-DAVIS, E., KESSLF & T.J., LEE, P.-J., FEI, D.T.W., GRIFFIN, P., STULTS, J.T., WADE, J.D. & RINDERKNECHT, E. (1991). Use of recombinant DNA derived human relaxin to probe the structure of the native protein. *Biochemistry*, **30**, 6006-6013.
- CHAGA, G., BOCHKARIOV, D.E., JOKHADZE, G.G., HOPP, J. & NELSON, P. (1999). Natural polyhistidine affinity tag for purification of recombinant proteins on cobalt(II)carboxymethylaspartate crosslinked agarose. *Journal of Chromatography A.*, 864, 247-256.
- CHEAH, S.H. & SHERWOOD, O.D. (1980). Target tissues for relaxin in the rat: tissue distribution of injected 1251-labeled relaxin and tissue changes in adenosine 3',5'-monophosphate levels after in vitro relaxin incubation. *Endocrinology*, **106**, 1203-1209.
- CHENG, Y.C. & PRUSOFF, W.H. (1973). Relationship between the inhibition constant Ki and the concentration of inhibitor which causes 50% inhibition (IC50) of an enzyme reaction. *Biochemical Pharmacology*, 22, 3099-3108.
- CHO, S.J., DLAMINI, B.J., KLINDT, J., SCHWABE, C., JACOBSON, C.D. & ANDERSON, L.L. (1998). Antiporcine relaxin (antipRLX540) treatment decreases relaxin plasma concentration and disrupts delivery in late pregnant pigs. *Animal Reproduction Science*, 52, 303-316.
- CLAASZ, A.A., BATHGATE, R.A.D., DAWSON, N.F., SUMMERS, R.J., OTVOS, J.L., TREGEAR, G.W. &
   WADE, J.D. (2001). Chemical synthesis and relaxin activity of analogues of ovine Insulin 3 containing specific B-chain residue replacements. In *Relaxin 2000: Proceedings of the Third*

International Conference on Relaxin & Related Peptides. eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. pp. 243-246. Netherlands: Kluwer Academic Publishers.

- CLAASZ, A.A., BOND, C.P., BATHGATE, R.A., OTVOS, L., DAWSON, N.F., SUMMERS, R.J., TREGEAR,
   G.W. & WADE, J.D. (2002). Relaxin-like bioactivity of ovine Insulin 3 (INSL3) analogues.
   European Journal of Biochemistry, 269, 6287-6293.
- COLON, J.M., GAGLIARDI, C., SCHOENFELD, C., AMELAR, R.D., DUBIN, L. & WEISS, G. (1989). Human relaxin stimulates human sperm penetration of bovine cervical mucus. Fertility & Sterility, 52, 340-342.
- COLON, J.M., SANTORO, N., GOLDSMITH, L.T., WHITCOMB, R., MUSKET, D. & WEISS, G. (1994). Relaxin secretion into human semen independent of gonadotropin stimulation. *Biology of Reproduction*, 50, 187-192.
- DANIELSON, L.A., SHERWOOD, O.D. & CONRAD, K.P. (1999). Relaxin is a potent renal vasodilator in conscious rats. *Journal of Clinical Investigation*, 103, 525-533.
- DANIELSON, L.A., KERCHER, L.J. & CONRAD, K.P. (2000). Impact of gender and endothelin on renal vasodilation and hyperfiltration induced by relaxin in conscious rats. *American Journal of Physiology - Regulatory Integrative & Comparative Physiology*, 279, R1298-R1304.
- DAWSON, N.F., MACRIS, M., SUMMERS, R.J., TAN, Y.Y., TREGEAR, G.W. & WADE, J.D. (2001). Chemical synthesis and biological activity of ovine Insulin 3, a relaxin structural homologue. In Relaxin 2000: Proceedings of the Third International Conference on Relaxin and Related Peptides. eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. pp. 247-248. Netherlands: Kluwer Academic Publishers.
- DODGE, K.L. & SANBORN, B.M. (1998). Evidence for inhibition by protein kinase A of receptor/G alpha(q)/phospholipase C (PLC) coupling by a mechanism not involving PLCbeta2. Endocrinology, 139, 2265-2271.
- DRAGOO, J.L., LEE, R.S., BENHAIM, P., FINERMAN, G.A. & HAME, S.L. (2003). Relaxin receptors in the human female anterior cruciate ligament. *American Journal of Sports Medicine*, 31, 577-584.
- DSCHIETZIG, T., RICHTER, C., BARTSCH, C., LAULE, M., ARMBRUSTER, F.P., BAUMANN, G. & STANGL, K. (2001). The pregnancy hormone relaxin is a player in human heart failure. *FASEB Journal*, 15, 2187-2195.
- EDDIE, L.W., BELL, R.J., LESTER, A., GEIER, M., BENNETT, G., JOHNSTON, P.D. & NIALL, H.D. (1986). Radioimmunoassay of relaxin in pregnancy with an analogue of human relaxin. *Lancet*, 1, 1344-1346.
- EDDIE, L.W., SUTTON, B., FITZGERALD, S., BELL, R.J., JOHNSTON, P.D. & TREGEAR, G.W. (1989). Relaxin in paired samples of serum and milk from women after term and preterm delivery. *American Journal of Obstetrics & Gynecology*, 161, 970-973.
- EIGENBROT, C., RANDAL, M., QUAN, C., BURNIER, J., O'CONNELL, L., RINDERKNECHT, E. & KOSSIAKOFF, A.A. (1991). X-ray structure of human relaxin at 1.5 A. Comparison to insulin

٠.

--- -

and implications for receptor binding determinants. Journal of Molecular Biology, 221, 15-21.

- EINSPANIER, A., MULLER, D., LUBBERSTEDT, J., BARTSCH, O., JURDZINSKI, A., FUHRMANN, K. & IVELL, R. (2001). Characterization of relaxin binding in the uterus of the marmoset monkey. Molecular Human Reproduction, 7, 963-970.
- ELDRIDGE, R.K. & FIELDS, P.A. (1985). Rabbit placental relaxin: Purification and immunohistochemical localisation. *Endocrinology*, **117**, 2512-2519.
- EMMI, A.M., SKURNICK, J., GOLDSMITH, L.T., GAGLIARDI, C.L., SCHMIDT, C.L., KLEINBERG, D. & WEISS, G. (1991). Ovarian control of pituitary hormone secretion in early human pregnancy. *Journal of Clinical Endocrinology & Metabolism*, 72, 1359-1363.
- ENDERS, A.C. & SCHLAFKE, S. (1969). Cytological aspects of trophoblast-uterine interaction in early implantation. *American Journal of Anatomy*, **125**, 1-29.
- EPPEL, W., KUCERA, E. & BIEGLMAYER, C. (1999). Relationship of serum levels of endogenous relaxin to cervical size in the second trimester and to cervical repening at term. *British Journal of Obstetrics and Gynaecology*, **106**, 917-923.
- ERIKSON, M.S. & UNEMORI, E.N. (2001). Relaxin clinical trials in systemic sclerosis. In *Relaxin 2000: Proceedings of the Third International Conference on Relaxin & Related Peptides*. eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. pp. 373-381. Netherlands: Kluwer Academic Publishers.
- ESSIG, M., SCHOENFELD, C., D'ELETTO, R., AMELAR, R., DUBIN, L., STEINETZ, B.G., O'BYRNE, E.M.
  & WEISS, G. (1982). Relaxin in human seminal plasma. Annals of the New York Academy of Sciences, 380, 224-230.
- FEI, D.T., GROSS, M.C., LOFGREN, J.L., MORA-WORMS, M. & CHEN, A.B. (1990). Cyclic AMP response to recombinant human relaxin by cultured human endometrial cells--a specific and high throughput in vitro bioassay. Biochemical & Biophysical Research Communications, 170, 214-222.
- FERLIN, A., SIMONATO, M., BARTOLONI, L., RIZZO, G., BETTELLA, A., DOTTORINI, T., DALLAPICCOLA, B. & FORESTA, C. (2003). The INSL3-LGR8/GREAT ligand-receptor pair in human cryptorchidism. Journal of Clinical Endocrinology & Metabolism, 88, 4273-4279.
- FEVOLD, H.L., HISAW, F.L. & MEYER, R.K. (1930). The relaxative hormone of the corpus luteum. Its purification and concentration. *Journel of the American Chemical Society*, **52**, 3340-3348.
- FIELDS, P.A., ELDRIDGE, R.K., FUCHS, A.R., ROBERTS, R.F. & FIELDS, M.J. (1983). Human placental and bovine corpora luteal oxytocin. *Endocrinology*, **112**, 1544-1546.
- FISHER, C., MACLEAN, M., MORECROFT, I., SEED, A., JOHNSTON, F., HILLIER, C. & MCMURRAY, J. (2002). Is the pregnancy hormone relaxin also a vasodilator peptide secreted by the heart. *Circulation*, **106**, 292-295.
- FRESHNEY, R.I. (1994). Culture of animal cells: a manual of basic technique. New York, N.Y., USA.: Wiley-Liss Inc.

- GARBER, S.L., MIROCHNIK, Y., BRECKLIN, C.S., UNEMORI, E.N., SINGH, A.K., SLOBODSKOY, L., GROVE, B.H., ARRUDA, J.A. & DUNEA, G. (2001). Relaxin decreases renal interstitial fibrosis and slows progression of renal disease. *Kidney International*, **59**, 876-882.
- GARGETT, C.E., LEDERMAN, F., HERYANTO, B., GAMBINO, L.S. & ROGERS, P.A.W. (2001). Focal vascular endothelial growth factor correlates with angiogenesis in human endometrium. Role of intravascular neutrophils. *Human Reproduction*, 16, 1065-1075.
- GARIBAY-TUPAS, J.L., MAASKANT, R.A., GREENWOOD, F.C. & BRYANT-GREENWOOD, G.D. (1995). Characteristics of the binding of 32P-labelled human relaxin to the human fetal membranes. Journal of Endocrinology, 145, 441-448.
- GATES, G.S., FLYNN, J.J., RYAN, R.J. & SHERWOOD, O.D. (1981). In vivo uptake of 125 I-Relaxin in the guinea pig. Biology of Reproduction, 25, 549-554.
- GEDDES, B.J., PARRY, L.J. & SUMMERLEE, A.J. (1994). Brain angiotensin-II partially mediates the effects of relaxin on vasopressin and oxytocin release in anesthetized rats. *Endocrinology*, 134, 1188-1192.
- GOLDSMITH, L.T., WEISS, G. & STEINETZ, B.G. (1995). Relaxin and its role in pregnancy. Endocrinology & Metabolism Clinics of North America, 24, 171-186.
- GORLOV, I.P., KAMAT, A., BOGATCHEVA, N.V., JONES, E., LAMB, D.J., TRUONG, A., BISHOP, C.E., MCELREAVEY, K. & AGOULNIK, A.I. (2002). Mutations of the GREAT gene cause cryptorchidism. Human Molecular Genetics, 11, 2309-2318.
- GOTO, M., SWANSON, L.W. & CANTERAS, N.S. (2001). Connections of the nucleus incertus. *Journal* of Comparative Neurology, **438**, 86-122.
- GUNNERSEN, J.M., CRAWFORD, R.J., TREGEAR, G.W. (1995). Expression of the relaxin gene in rat tissues. *Molecular & Cellular Endocrinology*, 110, 55-63.
- HALL, K. (1947). The effects of pregnancy and relaxin on the histology of the pubic symphysis in the mouse. *Journal of Endocrinology*, 5, 174-185.

HALLS, M.L. (2003). Characterisation of the signalling pathways of the LGR7 (relaxin) and LGR8 (Insl3) receptors. *BBioMedSc(Hons) Thesis*. pp. 1-78. Monash University, Victoria, Australia

- HALLS, M.L., BOND, C.P., BATHGATE, R.A. & SUMMERS, R.J. (2003). Characterisation of sites on the LGR7 (relaxin) and LGR8 (insulin3) receptors required for binding and signal transduction.
   In Proceedings of the Molecular Pharmacology of G Protein-Coupled Receptors. Poster 6. Victor Chang Cardiac Research Institute, NSW, Aust.
- HANSELL, D.J., BRYANT-GREENWOOD, G.D. & GREENWOOD, F.C. (1991). Expression of the human relaxin H1 gene in the decidua, trophoblast, and prostate. *Journal of Clinical Endocrinology* & Metabolism, 72, 899-904.
- HARLOW, E. & LANE, D. (1988). Antibodies: A laboratory manual. Cold Spring Habour, NY, USA: Cold Spring Harbour Laboratory.
- HISAW, F.L. (1926). Experimental relaxation of the pubic ligament of the guinea pig. Proceedings of the Society for Experimental Biology & Medicine, 23, 661-663.

÷.

÷

ŗ.

i.

Ì

**Chapter 9: References** 

- HORNSBY, D.J., POTERSKI, R.S. & SUMMERLEE, A.J.S. (2001). Relaxin binding and expression in the rat prostate. In *Relaxin 2000: Proceedings of the Third International Conference on Relaxin & Related Peptides*. eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. pp. 225-227. Netherlands: Kluwer Academic Publishers.
- HOUCK, K.A., FERRARA, N., WINER, J., CACHIANES, G., LI, B. & LEUNG, D.W. (1991). The vascular endothelial growth factor family: identification of a fourth molecular species and characterization of alternative splicing of RNA. *Molecular Endocrinology*, **5**, 1806-1814.
- HSU, C.J., MCCORMACK, S.M. & SANBORN, B.M. (1985). The effect of relaxin on cyclic adenosine 3',5'-monophosphate concentrations in rat myometrial cells in culture. *Endocrinology*, **116**, 2029-2035.
- HSU, S.Y., KUDO, M., CHEN, T., NAKABAYASHI, K., BHALLA, A., VAN DER SPEK, P.J., VAN DUIN, M.
  & HSUEH, A.J. (2000). The three subfamilies of leucine-rich repeat-containing G proteincoupled receptors (LGR): identification of LGR6 and LGR7 and the signaling mechanism for LGR7. *Molecular Endocrinology*, 14, 1257-1271.
- HSU, S.Y., NAKABAYASHI, K., NISHI, S., KUMAGAI, J., KUDO, M., SHERWOOD, O.D. & HSUEH, A.J. (2002). Activation of orphan receptor by the hormone relaxin. *Science*, **25**, 637-638.
- HSU, S.Y. (2003). New insights into the evolution of the relaxin-LGR signaling system. TRENDS in Endocrinology and Metabolism, 14, 303-309.
- HUDSON, A.L. (1993). Autoradiography techniques. In *Receptor Autoradiography: Principles and Practice.* eds Wharton, J. & Polak, J.M. pp. 57-77. New York: Oxford University Press Inc.
- HUDSON, P., HALEY, M., CRONK, M., CRAWFORD, R., HARALAMBIDIS, J., TREGEAR, G., SHINE, J. & NIALL, H. (1983). Structure of a genomic clone encoding biologically active human relaxin. *Nature*, 301, 628-631.
- HUDSON, P., JOHN, M., CRAWFORD, R., HARALAMBIDIS, J., SCANLON, D., GORMAN, J., TREGEAR, G., SHINE, J. & NIALL, H. (1984). Relaxin gene expression in human ovaries and the predicted structure of a human preprorelaxin by analysis of cDNA clones. *EMBO Journal*, 3, 2333-2339.
- HUGHES, S.J. & HOLLINGSWORTH, M. (1995). Cellular localization of the inhibitory action of relaxin against uterine spasm. *British Journal of Pharmocology*, **116**, 3028-3034.
- HWANG, J.J., SHANKS, R.D. & SHERWOOD, O.D. (1989). Monoclonal antibodies specific for rat relaxin. IV. Passive immunization with monoclonal antibodies during the antepartum period reduces cervical growth and extensibility, disrupts birth, and reduces pup survival in intact rats. Endocrinology, 125, 260-266.
- HWANG, J.J., LEE, A.B., FIELDS, P.A., HAAB, L.M., MOJONNIER, L.E. & SHERWOOD, O.D. (1991). Monoclonal antibodies specific for rat relaxin. V. Passive immunization with monoclonal antibodies throughout the second half of pregnancy disrupts development of the mammary apparatus and, hence, lactational performance in rats. *Endocrinology*, **129**, 3034-3042.
- ISAACS, N., JAMES, R., NIALL, H., BRYANT-GREENWOOD, G., DODSON, G., EVANS, A. & NORTH, A.C.T. (1978). Relaxin and its structural relationship to insulin. *Nature*, 271, 278-281.

.....

ł

÷

- IVELL, R., HUNT, N., KHAN-DAWOOD, F. & DAWOOD, M.Y. (1989). Expression of the human relaxin gene in the corpus luteum of the menstrual cycle and in the prostate. *Molecular & Cellular Endocrinology*, 66, 251-255.
- IVELL, R. (2002). Endocrinology. This hormone has been relaxin' too long! Science, 295, 637-638.
- IVELL, R., BALVERS, M., POHNKE, Y., TELGMANN, R., BARTSCH, O., MILDE-LANGOSCH, K., BAMBERGER, A.M. & EINSPANIER, A. (2003). Immunoexpression of the relaxin receptor LGR7 in breast and uterine tissues of humans and primates. *Reproductive Biology & Endocrinology*, 1, 114.
- JOHNSON, M.R., ABDALLA, H., ALLMAN, A.C., WREN, M.E., KIRKLAND, A. & LIGHTMAN, S.L. (1991). Relaxin levels in ovum donation pregnancies. *Fertility & Sterility*, **56**, 59-61.
- KAJAVA, A.V, VASSART, G. & WODAK, S.J. (1995). Modeling of the three-dimensional structure of proteins with the typical leucine-rich repeats. *Structure*, **3**, 867-877.
- KAKOURIS, H., EDDIE, L.W. & SUMMERS, R.J. (1992). Cardiac effects of relaxin in rats. Lancet, 339, 1076-1078.
- KENAKIN, T. (1995). Agonist-receptor efficacy. I: Mechanisms of efficacy and receptor promiscuity. Trends in Pharmacological Sciences, 16, 188-192.
- KHAN-DAWOOD, F.S., GOLDSMITH, L.T., WEISS, G. & DAWOOD, M.Y. (1989). Human corpus luteum secretion of relaxin, oxytocin, and progesterone. *Journal of Clinical Endocrinology & Metabolism*, 68, 627-631.
- KIZAWA, H., NISHI, K., ISHIBASHI, Y., HARADA, M., ASANO, T., ITO, Y., SUZUKI, N., HINUMA, S., FUJISAWA, Y., ONDA, H., NISHIMURA, O. & FUJINO, M. (2003). Production of recombinant human relaxin 3 in AtT20 cells. *Regulatory Peptides*, 113, 79-84.
- KOAY, E.S.C., BRYANT-GREENWOOD, G.D., YAMAMOTO, S.Y. & GREENWOOD, F.C. (1986). The human fetal membranes: a target tissue for relaxin. Journal of Clinical Endocrinology & Metabolism, 62, 513-521.
- KOHSAKA, T., MIN, G., LUKAS, G., TRUPIN, S., CAMPBELL, E.T. & SHERWOOD, O.D. (1998). Identification of specific relaxin-binding cells in the human female. *Biology of Reproduction*, **59**, 991-999.
- KOMPA, A.R., SAMUEL, C.S. & SUMMERS, R.J. (2002). Inotropic responses to human gene 2 (B29) relaxin in a rat model of myocardial infarction (MI): effect of pertussis toxin. *British Journal* of Pharmacology, 137, 710-718.
- KRAMER, S.M., GIBSON, U.E., FENDLY, B.M., MOHLER, M.A., DROLET, D.W. & JOHNSTON, P.P. (1990). Increase in cyclic AMP levels by relaxin in newborn rhesus monkey uterus cell culture. In Vitro Cellular & Developmental Biology, 26, 647-656.
- KRANTZ, J.C., BRYANT, H.H. & CARR, C.J. (1950). The action of aqueous corpus luteum extract upon uterine activity. Surgery Gynecology & Obstetrics, 90, 372-375.
- KROC, R.L., STEINETZ, B.G. & BEACH, V.L. (1959). The effects of estrogen, proestrogens, and relaxin in pregnant and nonpregnant laboratory rodents. *Annals New York Academy of Sciences*, 75, 942-980.

-----

Chapter 9: References

- KUENZI, M.J. & SHERWOOD, O.D. (1995). Immunohistochemical localization of specific relaxinbinding cells in the cervix, mammary glands, and nipples of pregnant rats. *Endocrinology*, 136, 1367-1373.
- KUMAGAI, J., HSU, S.Y., MATSUMI, H., ROH, J.S., FU, P., WADE, J.D., BATHGATE, R.A. & HSUEH, A.J. (2002). INSL3/Leydig insulin-like peptide activates the LGR8 receptor important in testis descent. *Journal of Biological Chemistry*, 277, 31283-31286.
- KUZNETSOVA, L., PLESNEVA, S., DERJABINA, N., OMELJANIUK, E. & PERTSEVA, M. (1999). Or the mechanism of relaxin action: the involvement of adenylyl cyclase signalling system. *Regulatory Peptides*, 80, 33-39.
- LAYDEN, S.S. & TREGEAR, G.W. (1996). Purification and characterization of porcine prorelaxin. J. Biochemical & Biophysical Methods, 31, 69-80.
- LEACH, R.E., KHALIFA, R., RAMIREZ, N.D., DAS, S.K., WANG, J., DEY, S.K., ROMERO, R. & ARMANT, D.R. (1999). Multiple roles for heparin-binding epidermal growth factor-like growth factor are suggested by its cell-specific expression during the human endometrial cycle and early placentation. *Journal of Clinical Endocrinology & Metabolism*, 84, 3355-3363.
- LESSEY, B.A., GUI, Y., APPARAO, K.B., YOUNG, S.L. & MULHOLLAND, J. (2002). Regulated expression of heparin-binding EGF-like growth factor (HB-EGF) in the human endometrium: A potential paracrine role during implantation. *Molecular Reproduction & Development*, 62, 446-455.
- LESSING, J.B., BRENNER, S.H., COLON, J.M., GINSBURG, F.W., SCHOENFELD, C., GOLDSMITH, L.T., SAROSI, P., AMELAR, R.D., DUBIN, L. & WEISS, G. (1986). Effect of relaxin on human spermatozoa. *Journal of Reproductive Medicine*, 31, 304-309.
- LINDHARD, A., BENTIN-LEY, U., RAVN, V., ISLIN, H., HVIID, T., REX, S., BANGSBOLL, S. & SORENSEN, S. (2002). Biochemical evaluation of endometrial function at the time of implantation. *Fertility & Sterility*, **78**, 221-233.
- LIU, C., CHEN, J., SUTTON, S., ROLAND, B., KUEI, C., FARMER, N., SILLARD, R. & LOVENBERG, T.W. (2003a). Identification of relaxin-3/INSL7 as a ligand for GPCR142. Journal of Biological Chemistry, 278, 50765-50770.
- LIU, C., ERISTE, E., SUTTON, S., CHEN, J., ROLAND, B., KUEI, C., FARMER, N., JORNVALL, H., SILLARD, R. & LOVENBERG, T.W. (2003b). Identification of relaxin-3/INSL7 as an endogenous ligand for the orphan G-protein-coupled receptor GPCR135. Journal of Biological Chemistry, 278, 50754-50764.
- LONGO, M., JAIN, V., VEDERNIKOV, Y.P., GARFIELD, R.E. & SAADE, G.R. (2003). Effects of recombinant human relaxin on pregnant rat uterine artery and myometrium in vitro. *American Journal Obstetric & Gynecology*, 188, 1468-1476.
- LOUMAYE, E., DECOOMAN, S. & THOMAS, K. (1980). Immunoreactive relaxin-like substance in human seminal plasma. Journal of Clinical Endocrinology & Metabolism, 50, 1142-1143.

- LOWRY, O.H., ROSEBROUGH, N.J., FARR, A.L. & RANDALL, R.J. (1951). Protein measurement with the folin phenol reagent. *Journal of Biological Chemistry*, 193, 265-275.
- LUNA, J.J., RJESEWIJK, A., HORCAJADAS, J.A., DE VAN OS, R., DOMINGUEZ, F., MOSSELMAN, S., PELLICER, A. & SIMON, C. (2004). Gene expression pattern and immunoreactive protein localisation of LGR7 receptor in human endometrium throughout the menstrual cycle. *Molecular Human Reproduction*, 10, 85-90.
- MACKIN, R.B. (1999). Streamlined procedure for the production of normal and altered versions of recombinant human proinsulin. *Protein Expression and Purification*, **15**, 308-313.
- MACLENNAN, A.H., GRANT, P., NESS, D. & DOWN, A. (1986a). Effect of porcine relaxin and progesterone on rat, pig and human myometrial activity in vitro. *Journal of Reproductive Medicine*, **31**, 43-49.
- MACLENNAN, A.H., GREEN, R.C., GRANT, P. & NICOLSON, R. (1986b). Ripening of the human cervix and induction of labor with intracervical purified porcine relaxin. Obstetrics & Gynecology, 68, 598-601.
- MACLENNAN, A.H. & GRANT, P. (1991). Human relaxin. In vitro response of human and pig myometrium. Journal of Reproductive Medicine, 36, 630-634.
- MACLENNAN, A.H., GRANT, P. & BRYANT-GREENWOOD, G. (1995). hRLX-1. In vitro response of human and pig myometrium. *Journal of Reproductive Medicine*, **40**, 703-706.
- MARRIOTT, D., GILLECE-CASTRO, B. & GORMAN, C.M. (1992). Prohormone convertase-1 will process prorelaxin, a member of the insulin family of hormones. *Molecular Endocrinology*, 6, 1441-1450.
- MASINI, E., BANI, D., BELLO, M.G., BIGAZZI, M., MANNAIONI, P.F. & BANI-SACCHI, T. (1997). Relaxin counteracts myocardial damage induced by ischemia-reperfusion in isolated guinea pig hearts: evidence for an involvement of nitric oxide. *Endocrinology*, 138, 4713-4720.
- MCDONALD, G.A., SARKAR, P., RENNKE, H., UNEMORI, E.N., KALLURI, R. & SUKHATME, V.P. (2003). Relaxin increases ubiquitin-dependent degradation of fibronectin in vitro and ameliorates renal fibrosis in vivo. American Journal of Physiology - Renal Physiology, 285, F59-F67.
- MCKINLEY, M.J., BURNS, P., COLVILL, L.M., OLDFIELD, B.J., WADE, J.D., WEIEINGER, R.S. & TREGEAR, G.W. (1997). Distribution of Fos immunoreactivity in the lamina terminalis and hypothalamus induced by centrally administered relaxin in conscious rats. *Journal of Neuroendocrinology*, 9, 431-437.
- MCKINLEY, M.J. & OLDFIELD, B.J. (1998). The brain as an endocrine target for peptide hormones. TRENDS in Endocrinology and Metabolism, 9, 349-354.
- MERCADO-SIMMEN, R.C., BRYANT-GREENWOOD, G.D. & GREENWOOD, F.C. (1982). Relaxin receptor in the rat myometrium: regulation by estrogen and relaxin. *Endocrinology*, **110**, 220-226.
- MIN, G., HARTZOG, M.G., JENNINGS, R.L., WINN, R.J. & SHERWOOD, O.D. (1997). Evidence that endogenous relaxin promotes growth of the vagina and uterus during pregnancy in gilts. *Endocrinology*, 138, 560-565.

-----

- MUKHIJA, R., RUPA, P., PILLAI, D. & GARG, L.C. (1995). High-level production and one-step purification of biologically active human growth hormone in *Escherichia coli*. Gene, 165, 303-306.
- MUMFORD, A.D., PARRY, L.J. & SUMMERLEE, A.J. (1939). Lesion of the subformical organ affects the haemotensive response to centrally administered relaxin in anaesthetized rats. *Journal of Endocrinology*, 122, 747-755.
- NEF, S. & PARADA, L.F. (1999). Cryptorchidism in mice mutant for Insl3. Nature Genetics, 22, 295-299.
- NGUYEN, B.T., YANG, L., SANBORN, B.M. & DESSAUER, C.W. (2003). Phosphoinositide 3-kinase activity is required for biphasic stimulation of cyclic adenosine 3',5'-monophosphate by relaxin. *Molecular Endocrinology*, 17, 1075-1084.
- NISHIKORI, K., WEISBRODT, N.W., SHERWOOD, O.D. & SANBORN, B.M. (1983). Effects of relaxin on rat uterine myosin light chain kinase activity and myosin light chain phosphorylation. *Journal* of Biological Chemistry, 258, 2468-2474.
- NOALL, M.W. & FRIEDEN, E.H. (1956). Variations in sensitivity of ovariectomized guinea pigs to relaxin. *Endocrinology*, 58, 659-664.
- NOYES, R., HERTIG, A. & ROCK, J. (1950). Dating the enodmetrial biopsy. Fertility & Sterility, 1, 3-25.
- O'BYRNE, E.M., FLITCRAFT, J.F., SAWYER, W.K., HOCHMAN, J., WEISS, G. & STEINETZ, B.G. (1978). Relaxin bioactivity and immunoactivity in human corpora lutea. *Endocrinology*, 107, 1641-1644.
- O'BYRNE, E.M., BRINDLE, S., QUINTAVALLA, J., STRAWINSKI, C., TABACHNICK, M. & STEINETZ, B.G. (1982). Tissue distribution of injected <sup>125</sup>I-labeled porcine relaxin: organ uptake, whole-body autoradiography, and renal concentration of radiometabolites. *Annals of the New York Academy of Sciencees*, **380**, 187-197.
- OLIVER, R.M., FIELDS, P.A. & LARKIN, L.H. (1978). Separation of relaxin activities in extracts of ovaries of pregnant sows by polyacrylamide gel electrophoresis. *Journal of Endocrinology*, 76, 517-525.
- OSHEROFF, F.L., LING, V.T., VANDLEN, R.L., CRONIN, M.J. & LOFGREN, J.A. (1990). Preparation of biologically active 32P-labeled human relaxin. Displaceable binding to rat uterus, cervix, and brain. *Journal of Biological Chemistry*, **265**, 9396-9401.
- OSHEROFF, P.L. & PHILLIPS, H.S. (1991). Autoradiographic localization of relaxin binding sites in rat brain. Proceedings for the National Academy of Science USA, 88, 6413-6417.
- OSHEROFF, P.L., CRONIN, M.J. & LOFGREN, J.A. (1992). Relaxin binding in the rat heart atrium. Proceedings for the National Academy of Science U S A, 89, 2384-2388.
- OSHEROFF, P.L. & HO, W.H. (1993). Expression of relaxin mRNA and relaxin receptors in postnatal and adult rat brains and hearts. Localization and developmental patterns. *Journal of Biological Chemistry*, 268, 15193-15199.

213

. . . . . . . . . .

1
OSHEROFF, P.L. & KING, K.L. (1995). Binding and cross-linking of <sup>32</sup>P-labeled human relaxin to human uterine cells and primary rat atrial cardiomyocytes. *Endocrinology*, **136**, 4377-4381.

- OVERBEEK, P.A., GORLOV, I.P., SUTHERLAND, R.W., HOUSTON, J.B., HARRISON, W.R., BOETTGER-TONG, H.L., BISHOP, C.E. & AGOULNIK, A.J. (2001). A transgenic insertion causing cryptorchidism in mice. *Genesis*, **30**, 26-35.
- PALEJWALA, S., STEIN, D., WOJTCZUK, A., WEISS, G. & GOLDSMITH, L.T. (1998). Demonstration of a relaxin receptor and relaxin-stimulated tyrosine phosphorylation in human lower uterine segment fibroblasts. *Endocrinology*, 139, 1208-1212.
- PALEJWALA, S., TSENG, L., WOJTCZUK, A., WEISS, G. & GOLDSMITH, L.T. (2002). Relaxin gene and protein expression and its regulation of procollagenase and vascular endothelial growth factor in human endometrial cells. *Biology of Reproduction*, 66, 1743-1748.
- PALMA, F., LONGH!, S., AGOSTINI, D. & STOCCHI, V. (2001). One-step purification of a fully active hexahistidine-tagged human hexokinase type I overexpressed in *Escherichia coli*. Protein Expression and Purification, 22, 38-44.
- PARRY, L.J., POTERSKI, R.S., SUMMERLEE, A.J.S. & JONES, S.A. (1990). Mechanism of the haemotensive action of porcine relaxin in anaesthetized rats. *Journal of Neuroendocrinology*, 2, 53-58.
- PARRY, L.J. & SUMMERLEE, A.J. (1991). Central angiotensin partially mediates the pressor action of relaxin in anesthetized rats. *Endocrinology*, **129**, 47-52.
- PARSELL, D.A., MAK, J.Y., AMENTO, E.P. & UNEMORI, E.N. (1996). Relaxin binds to and elicits a response from cells of the human monocytic cell line, THP-1. Journal of Biological Chemistry, 271, 27936-27941.
- PEAVY, D.E., ABRAM, J.D., FRANK, B.H. & DUCKWORTH, W.C. (1984). In vitro activity of biosynthetic human proinsulin. Receptor binding and biologic potency of proinsulin and insulin in isolated rat adipocytes. Diabetes, 33, 1062-1067.
- PETERSEN, L.K., SVANE, D., ULDBJERG, N. & FORMAN, A. (1991). Effects of human relaxin on isolated rat and human myometrium and uteroplacental arteries. Obstetrics & Gynecology, 78, 757-762.
- PIATIBRATOV, M., HOU, S., BROOUN, A., YANG, J., CHEN, H. & ALAM, M. (2000). Expression and fast-flow purification of a polyhistidine-tagged myoglobin-like aerotaxis transducer. Biochimica et Biophysica Acta, 1524, 149-154.
- PORTER, D.G., LYE, S.J., BRADSHAW, J.M. & KENDALL, J.Z. (1981). Relaxin inhibits myometrial activity in the ovariectomized non-pregnant ewe. Journal of Reproduction & Fertility, 61, 409-414.
- QUAGLIARELLO, J., GOLDSMITH, L., STEINETZ, B., LUSTIG, D.S. & WEISS, G. (1980). Induction of relaxin secretics in nonpregnant women by human chorionic gonadotropin. Journal of Clinical Endocrinology & Metabolism, 51, 74-77.

214

. ......

٠,

- RAAB, G., KOVER, K., PARIA, B.C., DEY, S.K., EZZELL, R.M. & KLAGSBRUN, M. (1996). Mouse preimplantation blastocysts adhere to cells expressing the transmembrane form of heparinbinding EGF-like growth factor. *Development*, 122, 637-645.
- REDDY, G.K., GUNWAR, S., GREEN, C.B., FEI, D.T., CHEN, A.B. & KWOK, S.C. (1992). Purification and characterization of recombinant porcine prorelaxin expressed in *Escherichia coli*. *Archives of Biochemistry & Biophysics*, 294, 579-585.
- REID, J.J., TRAN, N.H., TREGEAR, G.W. & ROCHE, P.J. (2001). The effect of relaxin on vascular function in rat aorta. In *Relaxin 2000: Proceedings of the Third International Conference on Relaxin and Related Peptides.* eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. pp. 183-184. Netherlands: Kluwer Academic Publishers.
- REINIG, J.W., DANIEL, L.N., SCHWABE, C., GOWAN, L.K., STEINETZ, B.G. & O'BYRNE, E.M. (1981). Isolation and characterization of relaxin from the sand tiger shark (Odontaspis taurus). *Endocrinology*, 109, 537-543.
- REMBIESA, B., BRACEY, R., BULLESBACH, E.E. & SCHWABE, C. (1993). The conserved basic residues in relaxin A chains are not required for biological activity - Synthesis and properties of a tetra-citrulline human relaxin 11. *Endocrine Journal*, 1, 263-268.
- ROCHE, P.J., CRAWFORD, R.J. & TREGEAR, G.W. (1993). A single-copy relaxin-like gene sequence is present in sheep. *Molecular & Cellular Endocrinology*, **91**, 21-28.
- ROH, J., VIRTANEN, H., KUMAGAI, J., SUDO, S., KALEVA, M., TOPPARI, J. & HSUEH, A.J. (2003). Lack of LGR8 gene mutation in Finnish patients with a family history of cryptorchidism. *Reproductive Biomedicine Online*, 7, 400-406.
- SAKBUN, V., ALI, S.M., GREENWOOD, F.C. & BRYANT-GREENWOOD, G.D. (1990). Human relaxin in the amnion, chorion, decidua parietalis, basal plate, and placental trophoblast by immunocytochemistry and northern analysis. *Journal of Clinical Endocrinology and Metabolism*, 70, 508-514.
- SAMBROOK, J. & RUSSELL, D.W. (2001). *Molecular Cloning: A laboratory manual*. Cold Spring Hoabour, New York: Cold Spring Habour Laboratory Press.
- SAMUEL, C.S., PARRY, L.J. & SUMMERS, R.J. (2003a). Physiological or pathological a role for relaxin in the cardiovascular system? *Current Opinion in Pharmacology*, **3**, 152-158.
- SAMUEL, C.S., TIAN, H., ZHAO, L. & AMENTO, E.P. (2003b). Relaxin is a key mediator of prostate growth and male reproductive tract development. *Laboratory Investigation*, 83, 1055-1067.
- SAMUEL, C.S., ZHAO, C., BATHGATE, R.A., BOND, C.P., BURTON, M.D., PARRY, L.J., SUMMERS, R.J., TANG, M.L., AMENTO, E.P. & TREGEAR, G.W. (2003c). Relaxin deficiency in mice is associated with an age-related progression of pulmonary fibrosis. *FASEB Journal*, 17, 121-123.
- SAMUEL, C.S., ZHAO, C., BOND, C.P., HEWITSON, T.D., AMENTO, E.P. & SUMMERS, R.J. (2004). Relaxin-1 deficient mice develop an age-related progression of renal fibrosis. *Kidney International*, Accepted for publication.

215

- SCHMIDT, M., BABU, K.R., KHANNA, N., MARTEN, S. & RINAS, U. (1999). Temperature-induced production of recombinant human insulin in high-cell density cultures of recombinant *Escherichia coli. Journal of Biotechnology*, **68**, 71-83.
- SCHWABE, C. & BRADDON, S.A. (1976). Evidence for one essential tryptophan residue at the active site of relaxin. *Biochemical & Biophysical Research Communication*, 68, 1126-1132.
- SCHWABE, C. & MCDONALD, J.K. (1977a). Primary structure of the B-chain of porcine relaxin. Biochemical & Biophysical Research Communication, 75, 503-510.
- SCHWABE, C. & MCDONALD, J.K. (1977b). Relaxin: a disulfide homolog of insulin. Science, 197, 914-915.
- SCHWABE, C. & BULLESBACH, E.E. (1994). Relaxin: structures, functions, promises, and nonevolution. FASEB Journal, 8, 1152-1160.
- SEIBOLD, J.R., CLEMENTS, P.J., FURST, D.E., MAYES, M.D., MCCLOSKEY, D.A., MORELAND, L.W., WHITE, B., WIGLEY, F.M., ROCCO, S., ERIKSON, M., HANNIGAN, J.F., SANDERS, M.E. & AMENTO, E.P. (1998). Safety and pharmacokinetics of recombinant human relaxin in systemic sclerosis. *Journal of Rheumatology*, 25, 302-307.
- SEIBOLD, J.R., KORN, J.H., SIMMS, R., CLEMENTS, P.J., MORELAND, L.W., MAYES, M.D., FURST, D.E., ROTHFIELD, N., STEEN, V., WEISMAN, M., COLLIER, D., WIGLEY, F.M., MERKEL, P.A., CSUKA, M.E., HSU, V., ROCCO, S., ERIKSON, M., HANNIGAN, J., HARKONEN, W.S. & SANDERS, M.E. (2000). Recombinant human relaxin in the treatment of scleroderma. A randomized, double-blind, placebo-controlled trial. *Annals of Internal Medicine*, 132, 871-879.
- SHARMA, R.C. & SCHIMKE, R.T. (1996). Preparation of electrocompetent *E. coli* using salt-free growth medium. *Biotechniques*, 20, 42-44.
- SHERWOOD, C.D. & O'BYRNE, E.M. (1974). Purification and characterization of porcine relaxin. Archives of Biochemistry & Biophysics, 160, 185-196.
- SHERWOOD, O.D., CHANG, C.C., BEVIER, G.W. & DZIUK, P.J. (1975a). Radioimmunoassay of plasma relaxin levels throughout pregnancy and at parturition in the pig. *Endocrinology*, 97, 834-837.
- SHERWOOD, O.D., ROSENTRETER, K.R. & BIRKHIMER, M.L. (1975b). Development of a radioimmunoassay for porcine relaxin using <sup>125</sup>I-labeled polytyrosyl-relaxin. *Endocrinology*, 96, 1106-1113.
- SHERWOOD, O.D. (1979). Purification and characterisation of rat relaxin. Endocrinology, 104, 886-892.
- SHERWOOD, O.D., CRNEKOVIC, V.E., GORDON, W.L. & RUTHERFORD, J.E. (1980). Radioimmunoassay of relaxin throughout pregnancy and during parturition in the rat. Endocrinology, 107, 691-698.
- SHERWOOD, O.D. (1994). Relaxin. In The physiology of reproduction. ed Knobil, E.a.N., J.D. pp. 861-1009. New York: Raven Press, Ltd.

- SHIRE, S.J., HOLLADAY, L.A. & RINDERKNECHT, E. (1991). Self-association of human and porcine relaxin as assessed by analytical ultracentrifugation and circular dichroism. *Biochemistry*, 30, 7703-7711.
- SINNAYAH, P., BURNS, P., WADE, J.D., WEISINGER, R.S. & MCKINLEY, M.J. (1999). Water drinking in rats resulting from intravenous relaxin and its modification by other dipsogenic factors. *Endocrinology*, 140, 5082-5086.
- SMITH, K.J., WADE, J.D., CLAASZ, A.A., OTVOS, L., JR., TEMELCOS, C., KUBOTA, Y., HUTSON, J.M., TREGEAR, G.W. & BATHGATE, R.A. (2001). Chemical synthesis and biological activity of rat INSL3. Journal of Peptide Science, 7, 495-501.
- SNELL, C.R. & SMYTH, D.G. (1975). Proinsulin: a proposed three-dimensional structure. Journal of Biological Chemistry, 250, 6291-6295.
- SOKOL, R.Z., WANG, X.S., LECHAGO, J., JOHNSTON, P.D. & SWERDLOFF, R.S. (1989). Immunohistochemical localization of relaxin in human prostate. *Journal of Histochemistry & Cytochemistry*, 37, 1253-1255.
- STEINETZ, B.G., BEACH, V.L., KROC, R.L., STASILLI, N.R., NUSSBAUM, R.E., NEMITH, P.J. & DUN, R.K. (1960). Bioassay of relaxin using a reference standard: a simple and reliable method utilizing direct measurement of interpubic ligament formation in mice. *Endocrinology*, 67, 102-115.
- STEINETZ, B.G., BEACH, V.L., TRIPP, L.V. & DEFALCO, R.J. (1964). Reactions of antisera to porcine relaxin with relaxin-containing tissues of other species in vivo and in vitro. Acta Endocrinologica, 47, 371-384.
- STEINETZ, B.G., BEACH, V.L. & KROC, R.L. (1969). Bioassay of relaxin. In Methods in Hormone Research. ed Dorfman, R.I. pp. 482-513. New York: Academic Press.
- STEINETZ, B.G., O'BYRNE, E.M., WEISS, G. & SCHWABE, C. (1982). Bioassay methods for relaxin: uses and pitfalls. Advances in Experimental Medicine & Biology, 143, 79-113.
- STEWART, A.G., RICHARDS, H., ROBERTS, S., WARWICK, J., EDWARDS, K., BELL, L., SMITH, J. & DERBYSHIRE, R. (1983). Cloning and expression of a porcine prorelaxin gene in *E.coli*. *Nucleic Acids Research*, 11, 6597-6609.
- STEWART, D.R. & PAPKOFF, H. (1986). Purification and characterisation of equine relaxin. Endocrinology, 119, 1093-1099.
- STEWART, D.R., CELNIKER, A.C., TAYLOR, C.A., JR., CRAGUN, J.R., OVERSTREET, J.W. & LASLEY, B.L. (1990). Relaxin in the peri-implantation period. Journal of Clinical Endocrinology & Metabolism, 70, 1771-1773.
- STEWART, D.R., ADDIEGO, L.A., PASCOE, D.R., HALUSKA, G.J. & PASHEN, R. (1992). Breed differences in circulating equine relaxin. Biology of Reproduction, 46, 648-652.
- STEWART, D.R., ERIKSON, M.S., ERIKSON, M.E., NAKAJIMA, S.T., OVERSTREET, J.W., LASLEY, B.L., AMENTO, E.P. & SEPPALA, M. (1997). The role of relaxin in glycodelin secretion. Journal of Clinical Endocrinology & Metabolism, 82, 839-846.

- STEWART, D.R. & VANDEVOORT, C.A. (1999). Relaxin secretion by human granulosa cell culture is predictive of *in-vitro* fertilization-embryo transfer success. *Human Reproduction*, 14, 338-344.
- ST-LOUIS, J. (1982). Pharmacological studies on the action of relaxin upon KCl-contracted rat uterus. Pharmacology, 25, 327-337.
- SUDO, S., KUMAGAI, J., NISHI, S., LAYFIELD, S., FERRARO, T., BATHGATE, R.A. & HSUEH, A.J. (2003). H3 relaxin is a specific ligand for LGR7 and activates receptor by interacting with both the ectodomain and the exoloop 2. Journal of Biologico mistry, 278, 7855-7862.
- SUGINO, N., KASHIDA, S., KARUBE-HARADA, A., TAKIGUCHI, S. & D. D., H. (2002). Expression of vascular endothelial growth factor (VEGF) and its receptors in human endometrium throughout the menstrual cycle and in early pregnancy. *Reproduction*, **123**, 379-387.
- SUMMERLEE, A.J., O'BYRNE, K.T., PAISLEY, A.C., BREEZE, M.F. & PORTER, D.G. (1984). Relaxin affects the central control of oxytocin release. *Nature*, **309**, 372-374.
- SUMMERLEE, A.J., O'BYRNE, K.T., JONES, S.A. & ELTRINGHAM, L. (1987). The subformical organ and relaxin-induced inhibition of reflex milk ejection in lactating rats. *Journal of Endocrinology*, 115, 347-353.
- SUMMERLEE, A.J., RAMSEY, D.G. & POTERSKI, R.S. (1998). Neutralization of relaxin within the brain affects the timing of birth in rats. *Endocrinology*, **139**, 479-484.
- SZLACHTER, N., O'BYRNE, E., GOLDSMITH, L., STEINETZ, B.G. & WEISS, G. (1980). Myometrial inhibiting activity of relaxin-containing extracts of human corpora lutea of prognancy. *American Journal of Obstetrics & Gynecology*, 136, 584-586.
- TAN, Y.Y., WADE, J.D., TREGEAR, G.W. & SUMMERS, R.J. (1998). Comparison of relaxin receptors in rat isolated atria and uterus by use of synthetic and native relaxin analogues. British Journal of Pharmacology, 123, 762-770.
- TAN, Y.Y. (1999). Characterisation and localisation of relaxin receptors. PhD Thesis. pp. 1-221. Universiti Sains Malaysia, Penang, Malaysia.
- TAN, Y.Y., WADE, J.D., TREGEAR, G.W. & SUMMERS, R.J. (1999). Quantitative autoradiographic studies of relaxin binding in rat atria, uterus and cerebral cortex: characterization and effects of oestrogen treatment. British Journal of Pharmacology, 127, 91-98.
- TAN, Y.Y., BOND, C.P., WADE, J.D., TREGEAR, G.W. & SUMMERS, R.J. (2001a). Bioassays for relaxin and relaxin-related peptides. In *Relaxin 2000: Proceeding of the Third International Conference on Relaxin and Related Peptides*. eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. pp. 271-272. Netherlands: Kluwer Academic Publishers.
- TAN, Y.Y., DAWSON, N.F., KOMPA, A.R., BOND, C., CLAASZ, A., WADE, J.D., TREGEAR, G.W. & SUMMERS, R.J. (2001b). Structure-activity relationships of relaxin and relaxin-like factor/insulin 3. In *Relaxin 2000; Proceedings of the Third International Conference on Relaxin and Related Peptides*. eds Tregear, G.W., Ivell, R., Bathgate, R.A.D. & Wade, J.D. pp. 255-260. Netherlands: Kluwer Academic Publishers.

1

: : .

٠ļ

- TAN, Y.Y., DAWSON, N.F., KOMPA, A.R., BOND, C.P., CLAASZ, A., WADE, J.D., TREGEAR, G.W. & SUMMERS, R.J. (2002). Structural requirements for the interaction of sheep insulin-like factor
   3 with relaxin receptors in rat atria. *European Journal of Pharmacology*, 457, 153-160.
- TANG, J.G., WANG, Z.H., TREGEAR, G.W. & WADE, J.D. (2003). Human gene 2 relaxin chain combination and folding. *Biochemistry*, 42, 2731-2739.
- TANG, X.-M. & CHEGINI, N. (1995). Human fallopian tube as an extraovarian source of relaxin: messenger ribonucleic acid expression and cellular localization of immunoreactive protein and 125I-relaxin binding sites. *Biology of Reproduction*, 52, 1343-1349.
- TELGMANN, R. & GELLERSEN, B. (1998). Marker genes of decidualization: activation of the decidual prolactin gene. Human Reproduction Update, 4, 472-479.
- THEIS, S., DORING, F. & DANIEL, H. (2001). Expression of the myc/His-tagged human peptide transporter hPEPT1 in yeast for protein purification and functional analysis. *Protein Expression and Purification*, 22, 436-442.
- TOTH, M., TASKINEN, P. & RUSKOAHO, H. (1996). Relaxin stimulates atrial natriuretic peptide secretion in perfused rat heart. Journal of Endocrinology, 150, 487-495.
- TREGEAR, G.W., FAGAN, C., REYNOLDS, H., SCANLON, D., JONES, P., KEMP, B., NIALL, H.D. & DU, Y.C. (1982). Porcine relaxin: synthesis and structure activity relationships. In Peptides, Synthesis, Structure, Function: Proceedings of the Seventh American Peptide Symposium. eds Rich, D.H. & Gross, E. pp. 249-252. Rockford, IL: Pierce Chemical Co.
- TREGEAR, G.W., DU, Y.C., WANG, K.Z., SOUTHWELL, C., JONES, P., JOHN, M., GORMAN, J., KEMP, B. & NIALL, H.D. (1983). The chemical synthesis of relaxin. In *Biology of Relaxin and its Role* ing the Human. eds Bigazzi, M., Greenwood, F.C. & Gasparri, F. pp. 42-55. Amsterdam: Excerpta Medica.
- TREGEAR, G.W. & WADE, J.D. (1995). Chemistry, synthesis and processing of relaxin. In Progress in Relaxin Research: The proceedings of the second international congress on the homone relaxin. eds MacLennan, A.H., Tregear, G.W. & Bryant-Greenwood, G.D. pp. 31-44. Singpore: World Scientific Publishing.
- TREGEAR, G.W., MATHIEU, M.N., DAWSON, N.F., SMITH, K.J., CLAASZ, A.A., CLIPPINGDALE, A.B., FU, P. & WADE, J.D. (2001). Chemical synthesis of relaxin analogues: Current status and future developments. In *Relaxin 2000: Proceedings of the Third International Conference on Relaxin & Related Peptides*. eds Tregear, G.W., Ivell, R., Bathgate, R.A. & Wade, J.D. pp. 231-236. Netherlands: Kluwer Academic Publishers.
- TSENG, L. & LANE, B. (1995). Role of relaxin in the decidualization of human endometrial cells. In Progress in Relaxin Research: The proceedings of the second international congress on the homone relaxin. eds MacLennan, A.H., Tregear, G.W. & Bryant-Greenwood, G.D. pp. 325-338. Singpore: World Scientific Publishing.
- TSENG, L., ZHU, H.H., MAZELLA, J., KOISTINEN, H. & SEPPALA, M. (1999). Relaxin stimulates glycodelin mRNA and protein concentrations in human endometrial glandular epithelial cells. *Molecular Human Reproduction*, 5, 372-375.

- UNEMORI, E.N. & AMENTO, E.P. (1990). Relaxin modulates synthesis and secretion of procollagenase and collagen by human dermal fibroblasts. *Journal of Biological Chemistry*, 265, 10681-10685.
- UNEMORI, E.N., BAUER, E.A. & AMENTO, E.P. (1992). Relaxin alone and in conjunction with interferon-gamma decreases collagen synthesis by cultured human scleroderma fibroblasts. *Journal of Investigative Dermetology*, **99.** 337-342.
- UNEMORI, E.N., PICKFORD, L.B., SALLES, A.L., PIERCY, C.E., GROVE, B.H., ERIKSON, M.E. & AMENTO, E.P. (1996). Relaxin induces an extracellular matrix-degrading phenotype in human lung fibroblasts in vitro and inhibits lung fibrosis in a murine model in vivo. Journal of Clinical Investigation, 98, 2739-2745.
- UNEMORI, E.N., ERIKSON, M.E., ROCCO, S.E., SUTHERLAND, K.M., PARSELL, D.A., MAK, J. & GROVE, B.H. (1999). Relaxin stimulates expression of vascular endothelial growth factor in normal human endometrial cells *in vitro* and is associated with menometrorrhagia in women. *Human Reproduction*, 14, 800-806.
- VANDLEN, R., WINSLOW, J., MOFFAT, B. & RINDERKNECHT, E. (1995). Human relaxin: purification, characterization and production of recombinant relaxins for structure function studies. In Progress in Relaxin Research: The proceedings of the second international congress on the homone relaxin. eds MacLennan, A.H., Tregear, G.W. & Bryant-Greenwood, G.D. pp. 59-72. Singpore: World Scientific Publishing
- VU, A.L., GREEN, C.B., ROBY, K.F., SOARES, M.J., FEI, D.T.W., CHEN, A.B. & KWOK, S.C.M. (1993). Recombinant porcine prorelaxin produced in chinese hamster ovary cells is biologically active. *Life Sciences*, 52, 1055-1061.
- WADE, J.D., LIN, F., SALVATORE, D., OTVOS, J.L. & TREGEAR, G.W. (1996). Synthesis and charcterization of human gene 1 relaxin peptides. *Biomedical Peptides, Proteins & Nucleic* Acids, 2, 27-32.
- WALSH, J.R. & NIALL, H.D. (1980). Use of an octadecylsilica purification method minimizes proteolysis during isolation of porcine and rat relaxins. *Endocrinology*, **107**, 1258-1260.
- WARD, D.G., THOMAS, G.R. & CRONIN, M.J. (1992). Relaxin increases rat heart rate by a direct action on the cardiac atrium. *Biochemical & Biophysical Research Communication*, 136, 999-1005.
- WEISS, G., O'BYRNE, E.M. & STEINETZ, B.G. (1976). Relaxin: A product of the human corpus luteum of pregnancy. *Science*, **194**, 948-949.
- WEISS, T.J. & BRYANT-GREENWOOD, G.D. (1982). Localization of relaxin binding sites in the rat uterus and cervix by autoradiography. *Biology of Reproduction*, 27, 673-679.
- WESTON, G.C., HAVIV, I. & ROGERS, P.A. (2002). Microarray analysis of VEGF-responsive genes in myometrial endothelial cells. *Molecular Human Reproduction*, **8**, 855-863.
- WETZEL, R., KLEID, D.G., CREA, R., HEYNEKER, H.L., YANSURA, D.G., HIROSE, T., KRAŠZEWSKI, A., RIGGS, A.D., ITAKURA, K. & GOEDDEL, D.V. (1981). Expression of *Escherichia coli* of a chemically synthesised gene for a "mini-C" analog of human proinsulin. *Gene*, 16, 63-71.

- WILLIAMS, D.C., VAN FRANK, R.M., MUTH, W.L. & BURNETT, J.P. (1982). Cytoplasmic inclusion bodies in *Escherichia coli* producing biosynthetic human insulin proteins. *Science*, 215, 687-689.
- WILLIAMS, E.J., BENYON, R.C., TRIM, N., HADWIN, R., GROVE, B.H., ARTHUR, M.J., UNEMORI, E.N.
  & IREDALE, J.P. (2001). Relaxin inhibits effective collagen deposition by cultured hepatic stellate cells and decreases rat liver fibrosis in vivo. Gut, 49, 577-583.
- WINN, R.J., BAKER, M.D., MERLE, C.A. & SHERWOOD, O.D. (1994). Individual and combined effects of relaxin, estrogen, and progesterone in ovariectomized gilts. II. Effects on mammary development. *Endocrinology*, 135, 1250-1255.
- WINSLOW, J.W., SHIH, A., BOURELL, J.H., WEISS, G., REED, B., STULTS, J.T. & GOLDSMITH, L.T. (1992). Human seminal relaxin is a product of the same gene as human luteal relaxin. Endocrinology, 130, 2660-2668.
- WINTER, J., LILIE, H. & RUDOLPH, R. (2002a). Renaturation of human proinsulin a study on refolding and conversion to insulin. *Analytical Biochemistry*, **310**, 148-155.
- WINTER, J., LILIE, H. & RUDOLPH, R. (2002b). Recombinant expression and in vitro folding of proinsulin are stimulated by the synthetic dithiol Vectrase-P. FEMS Microbiology Letters, 213, 225-230.
- WU, X., ENGLUND, K., LINDBLOM, B. & BLANCK, A. (2003). mRNA expression of often used housekeeping genes and the relation between RNA and DNA are sex-steroid-dependent parameters in human myometrium. *Gynecologic & Obstetric Investigation*, 55, 225-230.
- YANG, S., REMBIESA, B., BULLESBACH, E.E. & SCHWABE, C. (1992). Relaxin receptors in mice: demonstration of ligand binding in symphyseal tissues and uterine memorane fragments. *Endocrinology*, 130, 179-185.
- YANG, S., HEYN, H., ZHANG, Y.Z., BULLESEACH, E.E. & SCHWABE, C. (1993). The expression of human relaxin in yeast. Archives of Biochemistry & Biophysics, 300, 734-737.
- YKI-JARVINEN, H., WAHLSTROM, T. & SEPPALA, M. (1983). Immunohistochemical demonstration of relaxin in the genital tract of pregnant and non-pregnant women. Journal of Clinical Endocrinology and Metabolism, 57, 451-454.
- YKI-JARVINEN, H, WAHLSTROM, T. & SEPPALA, M. (1985). Human endometrium contains relaxin that is progesterone-dependent. Acta Obstetricia et Gynecologica Scandinavica, 64, 663-665.
- YU, S.S. & KITABCHI, A.E. (1973). Biological Activity of Proinsulin and Related Polypeptides in the Fat Tissue. Journal of Biological Chemistry, 248, 3753-3761.
- ZARREH-HOSHYARI-KHAH, R., BARTSCH, O., EINSPANIER, A., POHNKE, Y. & IVELL, R. (2001). Bioactivity of recombinant prorelaxin from the marmoset monkey. *Regulatory Peptides*, 97, 139-146.
- ZHANG, S. & YU, L. (1995). Identification of dynorphins as endogenous ligands for an opioid receptor-like orphan receptor. *Journal of Biological Chemistry*, **270**, 22772-22776.

- ZHAC, L., ROCHE, P.J., GUNNERSEN, J.M., HAMMOND, V.E., TREGEAR, G.W., WINTOUR, E.M. & BECK, F. (1999). Mice without a functional relaxin gene are unable to deliver milk to their pups. *Endocrinology*, 140, 445-453.
- ZHAO, L., SAMUEL, C.S., TREGEAR, G.W., BECK, F. & WINTOUR, E.M. (2000). Collagen studies in late pregnant relaxin null mice. *Biology of Reproduction*, **63**, 697-703.
- ZHAO, S., KUENZI, M.J. & SHERWOOD, O.D. (1996). Monoclonal antibodies specific for rat relaxin. IX. Evidence that endogenous relaxin promotes growth of the vagina during the second half of pregnancy in rats. *Endocrinology*, 137, 425-430.
- ZIMMERMANN, S., STEDING, G., EMMEN, J.M., BRINKMANN, A.O., NAYERNIA, K., HOLSTEIN, A.F., ENGEL, W. & ADHAM, I.M. (1999). Targeted disruption of the Insl3 gene causes bilateral cryptorchidism. *Molecular Endocrinology*, 13, 681-691.